

REQUEST FOR PROPOSALS FOR PRECONSTRUCTION AND LEASE-LEASEBACK SERVICES FOR PROPOSED CARMEL DEL MAR SCHOOL MODERNIZATION AND ASHLEY FALLS SCHOOL MODERNIZATION

Mandatory Project Meeting:

October 15, 2025, at 1:00:00 PM at

Carmel Del Mar School, 12345 Carmel Park Drive, San Diego, CA 92130

(Parties will subsequently travel to Ashley Falls School as well)

Proposal Deadline Date:

November 5, 2025, at 11:00:00 AM

Submit to:

Del Mar Union School District 11232 El Camino Real San Diego, CA 92130 Attn: Marley Nelms

TABLE OF CONTENTS

		<u>Page</u>
l.	PURPOSE OF THE RFP:	1
II.	BACKGROUND ON THE PROPOSED PROJECT:	2
III.	RFP TIMELINE:	3
IV.	QUESTIONS AND CLARIFICATION OF THE RFP	3
V.	PRELIMINARY SITE ANALYSIS AND PRECONSTRUCTION SERVICES	3
VI.	CONSTRUCTION SERVICES	5
VII.	DIR REGISTRATION AND PREVAILING WAGES	5
VIII.	SUBCONTRACTOR DESIGNATIONS	5
IX.	CONTENTS OF THE PROPOSAL	6
X.	PREPARATION AND SUBMITTAL OF THE PROPOSAL	13
XI.	PROPOSAL EVALUATION AND BEST VALUE SCORE	15
ΧII	GENERAL TERMS AND CONDITIONS	17

I. PURPOSE OF THE RFP:

By way of this Request for Proposals ("RFP"), the Del Mar Union School District ("District") seeks proposals from lease-leaseback contractors ("Contractor" or "Firm") to provide services in a multi-phase process related to the following proposed projects:

- (1) Carmel Del Mar School Modernization Project; and
- (2) Ashley Falls School Modernization Project.

Each of the foregoing may be referred to collectively as the "Proposed Projects." The first phase for each of the Proposed Projects will include preconstruction services. The second phase includes lease-leaseback construction services.

The purpose of this RFP is to obtain information that will enable the District to select one or more lease-leaseback Contractors using the "best value" competitive procurement process under Education Code section 17400 *et seq.*, that can assist the District with the above-listed Proposed Projects. The "best value" competitive procurement process is an evaluation process whereby a Firm is selected by the District on the basis of objective criteria for evaluating the qualifications of Firm(s), with the selected Firm(s) representing the best combination of price and qualifications. Each Contractor responding to this RFP should be prepared and qualified to provide the services described in this RFP to the District in an expeditious and timely manner so as to enable the District to meet time deadlines and schedules.

It is the District's strong preference and intention to award both Proposed Projects to a single Contractor. Accordingly, the District expects that the Contractor will be able to achieve economies of scale and will provide its best team (including, but not limited to, the best job superintendents) for the Proposed Projects. Notwithstanding the foregoing, the District may also decide to award the Proposed Projects to different Contractors, without limitation, and in any combination of awards. Notwithstanding anything to the contrary any Contractor awarded both Proposed Projects will be required to execute separate lease-leaseback agreements for each awarded Proposed Project.

For construction services, as opposed to preconstruction services, the Proposed Projects are anticipated to commence with the Carmel Del Mar School Modernization Project and then proceed to the Ashley Falls School Modernization Project. Any Contractor awarded both Proposed Projects will be required to agree to contractual language allowing the District to terminate any and/or both awarded Proposed Projects for any reason, at any time, and without penalty to the District. Submission of a proposal hereunder shall be considered acceptance of the foregoing requirement. So, for example, if the District is dissatisfied with the work on the first project, the District must be able to cancel any part or all of the other project in addition to the first project, without penalty. The foregoing contractual language will be in the form Lease-Leaseback Agreements that are being provided with this RFP. Such language will be removed, however, if a Contractor is awarded

just one Proposed Project.

II. BACKGROUND ON THE PROPOSED PROJECTS:

The Proposed Project, if approved, will be constructed using the lease-leaseback project delivery method authorized by Education Code section 17400 *et seq*. The District has contracted with the following architects for the Proposed Projects:

- (1) Carmel Del Mar School Modernization Project DLR Group, Inc.; and
- (2) Ashley Falls School Modernization Project HMC Architects.

The foregoing design firms will be the Architects of Record for the Proposed Projects.

The lease-leaseback Contractor will be expected to provide preconstruction services, and lease-leaseback construction services for the Proposed Projects as described below:

Carmel Del Mar School Modernization Project

The District owns real property located at 12345 Carmel Park Drive, San Diego, California 92130 ("Carmel Del Mar Property"). The Carmel Del Mar School Modernization Project will be constructed on the Carmel Del Mar Property, which consists of approximately 3 acres. Total combined building area of the Proposed Project is +/- 47,000 sq.ft.

The proposed project would include modernizations such as roofing/HVAC, painting, upgrading sewer, ADA accessibility; modernizing learning studios (whiteboards, tackboards, carpet, casework replacement, etc.); upgrading technology infrastructure; and interior upgrades to the Innovation Center (library), Multi-Use Room, exterior shade and landscaping improvements, and front office. The District will be pursuing State funds. The proposed project would not affect student enrollment, staffing, or circulation patterns at the site.

The estimated construction budget for the Carmel Del Mar School Modernization Project is \$10,000,00.00.

The estimated performance period for the construction services is 14 months to begin in June 2026.

Ashley Falls School Modernization Project

The District owns real property located at 13030 Ashley Falls Drive, San Diego, CA 92130 ("Ashley Falls Property"). The Ashley Falls School Modernization Project will be constructed on the Ashley Falls Property, which consists of approximately 4.1 acres. Total combined building area of the Ashley Falls School Modernization Project is +/- 65,000 sq.ft.

The proposed project would include modernizations such as roofing/HVAC, painting, upgrading sewer, ADA accessibility; modernizing learning studios (whiteboards,

tackboards, carpet, casework replacement, etc.); upgrading technology infrastructure; and interior upgrades to the Innovation Center (library), Multi-Purpose Room, exterior shade and landscaping improvements, and front office. The District will be pursuing State funds. Site demolition would be minimal and non-invasive. The proposed project would not impact student enrollment, staffing, or circulation patterns at the site.

The estimated construction budget for the Ashley Falls School Modernization Project is \$13,500,00.00

The estimated performance period for the construction services is 18 months to begin in June 2027 or earlier.

III. RFP TIMELINE:

Request for Proposals IssuedOct. 1, 2025
Mandatory Project MeetingOct. 15, 2025, by 1:00:00 pm
Deadline for Submittal of QuestionsOct. 17, 2025, by 11:00:00 am
Responses to the Questions SubmittedOct. 22, 2025
Due Date for Prequalification ApprovalOct. 31, 2025
(or 5 days before the Final Due Date for Submittal of Proposals)
Due Date for Submittal of ProposalsNov. 5, 2025, by 11:00:00 am
Interviews, if requiredNov. 24, 2025*
Anticipated Board Approval DateDec. 10, 2025*
Notification of Selected FirmDec. 11, 2025*
* Estimated deadlines subject to revision at the District's discretion.

IV. QUESTIONS AND CLARIFICATION OF THE RFP

All questions, requests for explanation or clarifications of any kind in regard to this RFP shall be made in written form, submitted via email to Marley Nelms, at mnelms@dmusd.org. All questions, requests for explanation or clarifications of any kind in regard to this RFP must be submitted by no later than 11:00:00 am, Oct. 17, 2025. A response will not be provided to any late questions, or requests for explanation or clarifications. All addenda and clarifications will be posted on the District's website, https://www.dmusd.org/Departments/Business-Services/Purchasing/Bid-Opportunities, and provided to those Firms that have attended the mandatory project meeting. Any interpretation, clarification, or correction of this RFP will only be made by addendum as noted above. No person or Firm is authorized to make any oral interpretation of any provision in this RFP, nor shall any oral interpretation be binding on the District.

V. PRECONSTRUCTION SERVICES

The District anticipates that the successful Contractor(s) will provide preconstruction services for one or both Proposed Projects including, but not limited to, reviewing the Proposed Project's plans and specifications during the design of a Proposed Project to identify and note all deficiencies, incongruities and inconsistencies that may affect constructability of a Proposed Project including, but not limited to, design and

specification omissions, incomplete and/or inconsistent plans, details and specifications, and any lack of coordination, together with all other appropriate, necessary and/or required services to facilitate and prepare for the successful development and construction of a Proposed Project.

The preconstruction services will also include, but not be limited to, the following tasks:

- 1. Design meetings with the architects and engineers ("A/E");
- 2. Review and validation of estimates prepared by the architect;
- 3. Preparation of a master critical path method schedule for a Proposed Project;
- 4. Preparation of cost estimates or updated cost estimates as follows:
 - (a) At the end of Architectural Schematic Design and Engineering Basis of Design reports. Cost estimating by Contractor will include Life Cycle Cost Analysis for building systems, which will allow District to decide direction for A/E team;
 - (b) At the end of Design Development. Contractor will participate in A/E Building Information Modeling Clash Detection process as constructability advisor;
 - (c) Construction Documents 50% same as Design Development, plus Contractor will provide Guaranteed Maximum Price ("GMP") for grading/utility construction at this point including allowances, contingencies, general conditions, costs and fees
 - (d) Construction Documents 100% Same as Design Development, plus Contractor will provide GMP for remainder of construction at this point including allowances, contingencies, general conditions, costs and fees:
- 5. Constructability reviews; value engineering; construction planning and phasing, and cost proposal strategies all with the goal that the DSA-approved plans and specifications for a Proposed Project will be complete such that a Proposed Project can be constructed by a competent licensed general building contractor in strict accordance with the DSA-approved plans and specifications without change orders, delays, or additional charges to District.

The successful Contractor shall not provide any preconstruction services or work that requires a contractor's license pursuant to Business and Professions Code section 7065 et seq. before receipt of the required Division of the State Architect approval.

Each Contractor shall submit with its proposal(s) line-item descriptions and costs for the preconstruction services it is proposing to perform. If any service is or has a reoccurring monthly cost, Contractor shall provide the total amount and the monthly amount.

VI. CONSTRUCTION SERVICES

The District anticipates that the successful Contractor will provide construction services for one or both Proposed Projects. Such cost proposals shall be based on a preliminary GMP using the information provided in this Request for Proposals. Once the final construction plans and specifications have been approved by the California Department of General Services, Division of the State Architect, the parties will establish a final GMP.

VII. <u>DIR REGISTRATION AND PREVAILING WAGES</u>

<u>DIR Registration.</u> Contractors and their subcontractors (of any tier) shall not be qualified to submit or be listed on a proposal, or engage in the performance of any contract for public work, as defined in the Labor Code, unless currently registered and qualified to perform public work pursuant to Section 1725.5 of the Labor Code. It is not a violation of this section for an unregistered contractor to submit a proposal that is authorized by Section 7029.1 of the Business and Professions Code or by Section 20103.5 of the Public Contract Code, provided the contractor is registered to perform public work pursuant to Section 1725.5 of the Labor Code at the time the contract is awarded.

<u>Prevailing Wages.</u> The Contractor and all subcontractors shall comply with the requirements set forth in Division 2, Part 7, Chapter 1 of the Labor Code. Pursuant to Labor Code section 1770 et seq., the District has obtained from the Director of the Department of Industrial Relations the general prevailing rate of per diem wages and the general prevailing rate for holiday and overtime work in the locality in which this work is to be performed for each craft, classification or type of worker needed to execute the contract. Copies are available from the District to any interested party on request and are also available from the Director of the Department of Industrial Relations.

VIII. SUBCONTRACTOR DESIGNATIONS

Pursuant to Education Code section 17406(a)(4), the District is <u>NOT</u> requiring Firms to identify and designate the subcontractors in their proposals. However, in the event the District approves a Proposed Project, and after award of the lease-leaseback contract, in accordance with Education Code section 17406(a)(4)(B), any subcontractor whose subcontract value exceeds one-half of one percent of the price allocable to the total construction work on a Proposed Project must be awarded a subcontract in accordance with the following process:

- A. Provide public notice of availability of work to be subcontracted in accordance with publication requirements applicable to the competitive bidding process of the District, including a fixed date and time on which qualifications statements, bids, or proposals will be due.
- B. Establish reasonable qualification criteria and standards.

C. Award the subcontract either on a best value basis or to the lowest responsible bidder.

The process above may include prequalification or short-listing except for those subcontractors requiring prequalification, which is described in this Section VIII below. Subcontractors awarded subcontracts as set forth above shall be afforded all the protections of the Subletting and Subcontracting Fair Practices Act. (Public Contract Code section 4100 et seq.)

All subcontractors (of any tier) performing any portion of the work must comply with Labor Code sections 1725.5 and 1771.1 and must be properly and currently registered with the California Department of Industrial Relations and qualified to perform public works pursuant to Labor Code section 1725.5 throughout the duration of the Proposed Project.

Additionally, all mechanical, electrical, and plumbing subcontractors, defined in Section IX(D)(1)(a) below, must be prequalified per Public Contract Code section 20111.6 prior to award of the applicable subcontract.

IX. CONTENTS OF THE PROPOSAL

Firms <u>must</u> submit one original, four (4) hard copies, and a digital copy (on a thumb drive) of each or any proposal. All proposals should address the requested information for each of the evaluation categories below. The proposal shall demonstrate the qualifications, competence, and capacity of the Firm:

- A. Cover Letter/Letter of Interest Include a cover letter, addressed to Marley Nelms, stating the eligibility of the Firm to respond to this RFP, a brief description and history of the Firm, and a statement of interest for one or both Proposed Projects.
- **B.** Table of Contents The table of contents shall reflect the order stated herein and shall include section titles and page numbers.
- C. Exceptions to the Lease-Leaseback Agreements The form Lease-Leaseback Agreements (Site Lease, Leaseback Agreement, and Construction Services Agreement) are attached to this RFP as Attachment 3. Please review each agreement and provide any proposed exceptions to those agreements on Attachment 1, Firm Questionnaire, Section E.

D. Evaluation Categories

1. **Mandatory Requirements** – The following requirements are mandatory and must be satisfied. The mandatory requirements will be scored on a pass/fail basis. Failure to meet any one of the mandatory requirements specified in this Section X(D)(1) will disqualify your Firm from any further consideration for this RFP.

a. Lease-Leaseback Contractor and Subcontractor **Prequalification** – All Firms submitting a proposal to this RFP must be prequalified with the District pursuant to Public Contract Code section 20111.6 (b)-(m) without exception prior to submitting a proposal. Any Firm that submits a proposal and is not pregualified will be deemed non-responsive and that Firm's proposal will be rejected and returned to the Firm unopened. Pregualification documents are available from the Del Mar Union School District website at https://www.dmusd.org/, and at 11232 El Camino Real, Ste. 100, San Diego, CA 92130. Additionally, the District's prequalification program will be administered by a thirdparty contractor named PQBids, which can be reached at https://pqbids.com/dmusd/. Prequalification documents must be submitted and approved by Oct. 31, 2025. Contractors will be notified by telephone, email, or mail of their prequalification rating within a reasonable period of time after submission of their pregualification documents. Contractors currently pregualified with the District will not need to submit another prequalification package. Provide a copy of your current prequalification status letter from the District with your proposal.

All mechanical, electrical or plumbing ("MEP") subcontractors (defined as contractors that **hold** a C-4, C-7, C-10, C-16, C-20, C-34, C-36, C-38, C-42, C-43 or C-46 license) must also be prequalified. This prequalification requirement applies even if the subcontractor will perform, or is designated and identified to perform, work that does not require one of the licenses listed above, but the subcontractor **holds** one of the licenses listed above.

MEP subcontractors (as defined above) must be prequalified prior to the award of their respective subcontract. A list of already prequalified MEP subcontractors, if any, will be made available by the District upon request. However, it is the responsibility of the Contractor to ensure that all MEP subcontractors **holding** any of the licenses listed above are properly prequalified.

b. **Contractor Responsibility** – Identify if your Firm has ever had the following occur in the past seven (7) years. For the purposes of this paragraph, "Firm" shall include any present or past (over the last seven years), officers, owners, principals, partners, or any qualifying individuals including any Responsible Managing Employee ("RME") or Responsible Managing Officer ("RMO"). Any occurrence of the following in

the past seven (7) years shall render the Firm not qualified to submit a proposal:

- Found to be a non-responsible contractor by any public agency;
- Convicted for false claims;
- Firm's license revoked or suspended;
- Debarred or otherwise ineligible to bid on or be awarded a public works contract;
- Terminated for cause or defaulted on a construction contract; or
- Convicted of a crime involving the awarding of a construction contract, or the bidding or performance of a construction contract.
- c. License Requirements Pursuant to Business and Professions Code section 7028.15 and Public Contract Code section 3300, the Contractor must possess a California Contractor's Class "B" license at the time of submittal of its proposal, and for the duration of any contract, if awarded. Subcontractors must possess the appropriate license for the work to be performed on a Proposed Project.
- d. **Performance and Payment Bonds** All Firms submitting a proposal to this RFP must be able to provide separate faithful payment and performance bonds, each in an amount equal to 100% of the total contract amount for each Proposed Project awarded. All bonds must be issued by a California admitted surety as defined in California Code of Civil Procedure section 995.120. Firms must provide a letter from their surety indicating the Firm's current and overall bonding capacity, and the ability to meet the bond requirements in the Construction Services Agreement.
- e. **Insurance Requirements** All Firms submitting a proposal to this RFP must have the ability to meet all of the insurance requirements set forth in the Construction Services Agreement for each Proposed Project. Firms must include a copy of their current certificate of insurance in their proposals evidencing the following minimum insurance requirements:

Comprehensive general liability insurance with a combined single limit per occurrence of not less than \$2,000,000 or commercial general liability Insurance (including automobile insurance) which provides limits of not less than:

- Per occurrence (combined single limit): \$3,000,000
- Project Specific Aggregate (for each individual Proposed Project): \$6,000,000
- Products and Completed Operations: \$6,000,000
- Personal and Advertising Injury Limit: \$3,000,000

The following special hazards shall be covered by riders or riders to above mentioned public liability insurance or property damage insurance policy or policies of insurance, in amounts as follows:

- Automotive and truck where operated: \$3,000,000
- Material Hoist where used: \$1,000,000
- Explosion, Collapse and Underground (XCU coverage): \$1,000,000

Excess Liability Insurance coverage in the amount of \$5,000,000

Workers' Compensation Insurance shall be provided in accordance with the provisions of the California Labor Code adequate to protect any person, firm, or corporation employed directly or indirectly in connection with the work of the Contractor from claims under Workers' Compensation Acts which may arise for operations, whether such operations be by any person, firm, or corporation, employed directly or indirectly by the Contractor upon or in connection with the work. Notwithstanding anything to the contrary, the District is requiring workers' compensation insurance in the amount of at least \$3,000,000.

Additional insurance requirements are provided in the Construction Services Agreement attached hereto.

Provide a letter from your insurance broker stating that you will be able to provide and meet the insurance requirements in the Construction Services Agreement for each Proposed Project.

- 2. Firm Personnel, Capacity, and Methodology Attachment 1. Each Firm must completely answer all questions in Attachment 1 of the RFP. Note: Questions may be answered in other sections of the proposal if clearly and conspicuously identified and referenced in the proposal. The following shall be stated:
 - a. **Description of Firm** Include a description of the Firm's qualifications for providing preconstruction and lease-leaseback services on California school construction projects. Include information regarding the size of the Firm, location of the office from which the required services will be performed and/or headquartered, nature of all work performed, and the number of years in this particular business. The Firm shall provide an affirmative statement that it is independent of the District as defined by generally accepted standards.
 - b. Firm's Personnel and Staffing Resources - Submit resume(s) or profiles for each key staff who will be proposed provide the requested services. including qualifications and recent relevant experience providing similar services. Each resume shall include, without limitation, the following information; (a) education; (b) years of relevant experience; (c) professional registrations, certifications and affiliations (d) project-specific experience with focus on public works projects and emphasis on K-12 projects providing preconstruction and lease-leaseback services, including dates and durations of each project listed and the name of the firm where employed. Include a discussion on the Firm's philosophy and approach for providing outstanding customer service.
 - c. Capacity & Methodology Describe how the Firm will provide services and fulfill the requirements and expectations of the District and this RFP. Use this section to address the ability of your Firm to undertake and accomplish the required scope of services for any Proposed Project while meeting deadlines, the Firm's record of meeting schedules and deadlines of other clients, advantages over other firms in the same industry, strength and stability as a business, and supportive client references. Describe the Firm's ability to provide preconstruction and lease-leaseback services exclusively and in a timely manner for the District and the Firm's commitment to providing experienced personnel assigned to District's Proposed Projects.
 - d. **Litigation** Furnish and provide specific information on any termination for convenience, litigation settled or judgments

entered within the last five (5) years, as well as any civil judgments within the last five (5) years. Identify if the Firm or any employee of the Firm is a party to an existing dispute with an owner, or owner's consultants, related to any project for which the Firm provided construction services. If so, please describe the nature of the dispute and its anticipated outcome.

Identify if the Firm has ever filed a petition for bankruptcy. If so, please provide the date the petition was filed and identify the jurisdiction in which it was filed.

- 3. Relevant Experience and Past Performance Description of past performance and related experience. Each Firm is required to submit a list of its preconstruction and lease-leaseback services provided in California in the past five (5) years that are of the approximate size of a Proposed Project described in the RFP. The list shall include: (1) a description and size of the project, (2) the general scope of the work, (3) the dates services were performed for preconstruction services and for lease-leaseback services, (4) the total price for the project (please state amounts separately for preconstruction services and for lease-leaseback services and include the final guaranteed maximum price and all contingencies and allowances), and (5) the owner's name, address, email(s), and phone number.
- 4. **Preconstruction Services** Describe your methodology in providing preconstruction services for one or more Proposed Projects, specifically discussing value engineering, constructability review, estimating, and scheduling. Provide examples of constructability reviews that you performed that resulted in the identification of significant design conflicts or omissions, and of value engineering that resulted in significant savings of money or time. State whether your firm has building information modeling capability and use of BIM on prior lease-leaseback projects.
- 5. Labor Compliance/Skilled and Trained Workforce Describe your ability to comply with statutory requirements for the payment of prevailing wages, including the monitoring and enforcement of your subcontractor's payment of prevailing wages. Provide copies of any DIR Civil Wage and Penalty Assessment issued against your Firm, explain the circumstances for the Civil Wage and Penalty Assessment, and the final resolution.

Further describe your plan and methodology to comply with the requirements for the use of a "skilled and trained workforce" as defined in Education Code section 17407.5 and Public Contract Code section 2600 et seq., for each apprenticable occupation that will be used on a Proposed Project, including all subcontractors of

any tier. Include in your discussion your plan and methodology to comply with the percentage requirements for the use of "skilled journeypersons" for each apprenticable occupation and the required monthly report demonstrating compliance. Please include a copy of a sample monthly report prepared by your Firm for another owner if available. Finally, identify and discuss which apprenticable occupation(s) will be the most difficult to meet the percentage requirements for skilled journeypersons on the Proposed Project and state why.

6. Safety – Discuss your plan to maintain a safe worksite. In your discussion, include whether your Firm has an Injury and Illness Prevention Program that complies with 8 CCR § 1509, whether your Firm has a safety program that meets Cal/OSHA requirements, and whether your Firm will provide a full-time person dedicated to safety on a Proposed Project.

Please state whether you have had any accidents in the past five years that resulted in a construction fatality on any of your projects and provide any details for each incident.

Please state whether you have had any recordable injuries in the past five years and provide the average total recordable injuries for the past five years.

Please provide an EMR verification from the State of California or an insurance company for each of the past five years.

- 7. **Locality** Under this criteria the District will assign additional points for Contractors that are located, or have an office situated, within 30 road miles from the District's office located at 11232 El Camino Real, Ste. 100, San Diego, CA 92130.
- E. Fee Proposal Preconstruction Fee, Lease-Leaseback Fee, and General Conditions "Attachment 2"

For each Proposed Project for which a Contractor wants to be considered, provide a lump sum fee to provide the preconstruction services, the lease-leaseback fee, and a monthly general conditions fee as set forth on Attachment 2. The proposed fees should include all direct labor costs, fringe benefits, insurance, overhead, profit, and all other expenses the Contractor will incur in providing the preconstruction services and the lease-leaseback construction services. There fees, including general conditions, should be include separate line items for each fee/cost. On a separate sheet, please explain whether and how the Contractor can obtain economies of scale if they are awarded more than one Proposed Project.

X. PREPARATION AND SUBMITTAL OF THE PROPOSAL

A. Proposal Submittal and Deadline

One original, four (4) hard copies and a digital copy (on a thumb drive) of each proposal must be submitted under sealed cover by no later **than 11:00:00 am on Nov. 5, 2025.** Label the outside of the sealed proposal envelope or box with your company name, proposal title and RFP deadline.

Proposals shall be delivered to the attention of:

Marley Nelms, Business Services Department Del Mar Union School District, 11232 El Camino Real, Ste. 100 San Diego, CA 92130

It is the sole responsibility of the Firm submitting the proposal to ensure that its proposal is actually received in the office prior to the deadline time and due date. Unless this RFP is extended by a written amendment from the District, proposals received after the time on the due date will not be considered. Faxed or emailed proposals will not be accepted.

B. <u>Proposal Completeness</u>

Proposals shall be completed in all respects as required by the instructions herein. A proposal may be rejected if it is conditional or incomplete, or if it contains alterations of form or major irregularities of any kind as determined solely by the District. A proposal will be rejected if, in the opinion of the District, the information contained therein was intended to mislead the District in the evaluation of the proposal.

C. <u>District Not Responsible For Preparation Costs</u>

All costs incurred in the preparation, submission and/or presentation of Firms responding to the RFP including, but not limited to, the Firm's travel expenses or long distance charges to attend any pre-conferences, presentations, interviews, and negotiation sessions, shall be the sole responsibility of the Firm and will not be reimbursed by the District. The District shall not pay for any costs incurred for proposal or contract preparation as a result of termination of this RFP or termination of the contract resulting from this RFP.

D. Right to Use Ideas

All proposals and other materials submitted become the property of the District. District reserves the right to use any ideas presented in any response to the RFP. Selection or rejection of any proposal shall not affect this right.

E. <u>Modification or Withdrawal Of Proposal</u>

A Firm may modify or withdraw a proposal after submission by written request of withdrawal and re-submission, provided that the proposal withdrawal is prior to the proposal deadline specified.

F. Amendments

Firms are advised that the District reserves the right to amend this RFP at any time. Amendments will be done formally by providing written amendments to all potential Firms known to have received a copy of the RFP and/or by publishing the amendment on the District's website.

G. Equal Opportunity

The Firm shall certify that it is an Equal Opportunity Employer and has made a good faith effort to improve minority employment and agrees to meet federal and state guidelines. Legal residents of the United States of America shall be used in providing all services under this RFP.

Firm shall not discriminate nor permit discrimination against any person because of race, color, religion, age, national origin, ancestry, creed, handicap, sexual orientation, or union membership in the performance of the work, including but not limited to preparation, manufacturing, fabrication, installation, erection and delivery of all supplies and equipment. In the event of receipt of such evidence of such discrimination by the Firm or its agents, employees or representatives, District shall have the right to rescind and terminate any and/or all the contract(s).

The successful Firm agrees to include the paragraph above with appropriate adjustments in all subcontracts, which are entered into for work to be performed pursuant to the contract.

H. Waiver or Breach Thereof

No term or provision of this RFP shall be deemed waived and no breach excused, unless such waiver or consent shall be in writing and signed by an individual authorized to so waive or consent for each party. Any consent by either party to, or waiver of, a breach by the other, whether express or implied, shall not constitute a consent to, waiver of, or excuse for, any other breach or subsequent breach, except as may be expressly provided in the written waiver or consent.

I. Covenant Against Gratuities

The Firm warrants that no gratuities (in the form of entertainment, gifts, or otherwise) were offered or given by the Firm or any agent or representative of the Firm, to any officer or employee or agent/consultant of the District with a view toward securing the resultant contract or securing favorable treatment with respect to any determinations concerning the award of any contract(s). For breach or violation of this provision, the District shall

have the right to terminate any negotiation or any resultant contract(s), either in whole or in part, and any loss or damage sustained by the District in procuring on the open market any items which the Firm agreed to supply shall be borne and paid for by the Firm. The rights and remedies of the District provided in this clause shall not be exclusive and are in addition to any other rights and remedies provided in the Lease-Leaseback Agreements or by law.

J. Conflict of Interest

The Firm represents that it presently has no interest and will not acquire any interest, directly or indirectly, which would conflict in any manner or degree with the performance of the services hereunder. The Firm further agrees that no person having any such known interest or conveyed an interest shall be employed, directly or indirectly, in the delivery of services under this RFP.

K. <u>Independent Contractor</u>

The Firm represents itself as an independent contractor offering such services to the general public and shall not represent him/herself or his/her employees to be an employee of the District. Therefore, the Firm shall assume all legal and financial responsibility for taxes, FICA, employee fringe benefits, workers compensation, employee insurance, minimum wage requirements, overtime, and other expenses.

L. <u>Precedence of Documents</u>

The contract(s) between the District and the successful Firm(s) shall consist of (1) this Request for Proposals (RFP) and any amendments thereto, (2) the Agreements to be executed with the successful Firm(s); and (3) the proposal submitted by the Firm to the District in response to the RFP. In the event of a conflict in language between the documents referenced above, the provisions and requirements set forth and/or referenced in the Agreements shall govern.

M. Compliance with Laws

Notwithstanding anything to the contrary, in connection with the furnishing of services or performance of work under this RFP, the Firm agrees to comply with the Fair Labor Standards Act, Equal Opportunity Employment Act, and all other applicable federal and state laws, regulations and executive orders to the extent that the same may be applicable.

XI. PROPOSAL EVALUATION AND BEST VALUE SCORE

A. Proposal Evaluation Committee

The District's Proposal Evaluation Committee will consist of at least three (3) members and will score each proposal based on the evaluation categories and points set forth in the RFP (See, Section XI(B)). The District shall have sole discretion and control over the number and selection of individuals who will serve on the Proposal Evaluation Committee.

Each Firm's proposal will be evaluated and scored only on the information that is included in the Firm's proposal. If any information is missing or incomplete in any proposal, no Firm will be provided the opportunity to supply the missing or incomplete information, nor will the District seek clarification of any information included in the proposals. Each proposal must be capable of being evaluated independently based solely on the information contained in the proposal.

B. <u>Evaluation Categories, Points, and Scoring</u>

Each member on the Proposal Evaluation Committee will independently score each proposal and each Firm's score will be equal to the average score from the Evaluation Committee (i.e., the total number of points from the Proposal Evaluation Committee divided by the number of Proposal Evaluation Committee members: base score = sum total of points from all members of the Proposal Evaluation Committee divided by the number of committee members). The base score will be calculated to two decimal places. Although the Proposal Evaluation Committee will independently score each proposal, the members reserve the right to discuss the RFP process and information in any proposal with other members and/or District legal counsel. Finally, scores will be given for each Proposed Project for which a Contractor seeks consideration. For example, if a Contractor submits price sheets (Attachment 2) for each Proposed Project, that Contractor will be scored separately on each project.

The RFP contains nine (9) Evaluation Categories, as discussed herein, and the maximum number of points for each category is shown in the table below. The initial base scoring will include the first eight (8) Evaluation Categories. The top three (3) Firms may be asked to interview with the District. Only these top top three (3) Firms may be further scored as demonstrated below.

In the event of any tie, the Proposal Evaluation Committee shall have the sole discretion to select one of the proposals in any manner it deems reasonable and fair.

EVALUATION CATEGORY	POINTS
Mandatory Requirements	Pass/Fail
Firm Personnel, Capacity, and Methodology	25
Relevant Experience and Past Performance	30
Preconstruction Services	30
Labor Compliance/Skilled and Trained Workforce	20
Safety	10
Locality	5
Price (Attachment 2)	30
MAXIMUM BASE SCORE	150
Interview (top 3 Firms may be considered)	50
MAXIMUM TOTAL SCORE	200

C. Short List Interviews

After each Firm's score is calculated using the first eight (8) Evaluation Categories, the Proposal Evaluation Committee may, in its sole discretion, determine a short list of Firms to interview with the Proposal Evaluation Committee. If interviews are conducted, the interview will consist of a presentation followed by a question-and-answer period. After the interview, the Proposal Evaluation Committee will add the points for the "Interview" Evaluation Category (again calculated as an average as described above) based on the information presented at the interviews and the Firm's performance to determine the Firm's total final score. If the Firm provides information during the Interview that differs from, or otherwise clarifies any information in the proposal submitted, the Proposal Evaluation Committee reserves the right to adjust any previous points given to the Firm in the appropriate Evaluation Category. The total final scores will be used to determine the Best Value Scores (as defined in Education Code section 17400). Best Value Score = sum total of points from all members of the Proposal Evaluation Committee divided by the number of committee members. If interviews are required by the District, a Best Value Score will not be calculated for those Firms not invited to interview with the Proposal Evaluation Committee.

XII. GENERAL TERMS AND CONDITIONS

District Obligation

Receipt of proposals and responses to this RFP does not obligate the District to any Firm, in any way. The District reserves the right to accept or reject any or all proposals, and to waive any minor irregularities or informalities in any proposal or in the RFP process.

Award of Contract(s)

This RFP implies no obligation to award contracts to any Firm. If it is in the best interest of the District, the District retains the sole and absolute right to select the Firm(s) that best meet(s) the District requirements. The award is subject to acceptance and approval by the District's Board of Trustees ("Board"). The District also reserves the right to reject any or all proposals.

Approval to Start Work

The successful Firm may perform the preconstruction services once Lease-Leaseback Agreements (Site Lease, Leaseback Agreement, and Construction Services Agreement) have been fully executed and approved by both parties and all appropriate documentation has been received and approved by the District. The foregoing includes the issuance of a duly executed Notice to Proceed. The District shall not be responsible for work done, even in good faith, prior to approval of the agreement by the District's Board and the receipt of a duly executed Notice to Proceed. Thereafter, no work shall be performed on the construction services until the District has issued a separate Notice to Proceed for the construction services. The District shall not be responsible for work done, even in good faith, prior to the District's issuance of the appropriate Notice to Proceed.

Ownership of Documents

All proposals and materials submitted in response to this RFP shall become the property of the District and shall be considered a part of public records and subject to disclosure under the California Public Records Act, unless exempted by law. In addition, all designs, drawings, specifications, notes and other work developed in the performance of any services resulting from this RFP shall be the sole property of District and may be used by District for any purpose without additional compensation to any Firm. Each Firm agrees that, by submitting any proposal in response to this RPF, such Firm will not assert any rights or establish any claim under the design patent or copyright laws.

Joint Ventures

Where two or more Firms desire to submit a single response to this RFP, they should do so on a prime-subcontractor basis rather than as a joint venture or informal team. The District intends to contract with a single Firm and not with multiple Firms doing business as a joint venture.

Fingerprinting

Education Code section 45125.1, subdivision (a), requires that entities under contract with the District shall ensure that each employee who interacts with a student outside the immediate supervision and control of the student's parent/guardian or of a District employee, shall have a valid criminal records summary as described in Education Code section 44237. Education Code section 45125.1, subdivision (c), authorizes the District to require compliance with the foregoing requirements with respect to employees in addition to those within the scope of Education Code section 45125.1, subdivision (a). Therefore, the successful Contractor and its subcontractor(s) shall comply in all respects with the requirements of Education Code section 45125.1. To the greatest extent permitted by law, the District may impose other requirements designed to protect students in connection with the performance of the Proposed Projects. Any successful Contractor that is a sole proprietor shall have an immediate and affirmative duty to inform the District of such status so as to allow the District to comply with Education Code section 45125.1. Notwithstanding the foregoing, compliance with Education Code Section 45125.1 shall be subject to the District's sole discretion to grant alternative compliance requirements per Education Code section 45125.2, subdivision (a).

(Remainder of page intentionally left blank.)

ATTACHMENT 1 - FIRM QUESTIONNAIRE

The Firm ("Respondent") shall furnish all the following information accurately and completely for the Respondent and each of the proposed staff and submit this with the proposal. Failure to comply with this requirement may cause rejection of the Respondent's qualifications. Attach additional sheets if necessary. "You" or "your" or "Respondent" as used herein collectively refers to the Respondent and/or any of its owners, officers, directors, shareholders, parties, principals, or any qualifying individuals including any RME or RMO.

If the same information is provided elsewhere in your qualification and qualification materials, then please clearly identify such in the following questions.

Please be advised that the District may request verbal or written clarifications, additional information, an interview or presentation at any time regarding this questionnaire.

SECTION A - GENERAL INFORMATION

Res	Respondent's name, address and contact information:				
	ephone:		Facsi	mile:	
Wek	osite:				
Тур	Type of Respondent: (check one)				
Indi	vidual	Partnership _		Corporation _	
Oth	er (please exp	lain)			
Nan	Names and titles of all principals/officers of the Respondent:				
Nan	ne		Title		Phone Number & email

5)	Pleas	se list any applicable certifications and licenses and their associated numbers:			
5)	Have you or any of your principals ever conducted similar services under a different name or certification or different license number?				
	(a)	If yes, give other name, address and certification or license number.			
		Name			
		Address			
		License No. (if any)			
')	How name	many years has Respondent been in business under its present business e?			
3)	How	many years of experience does Respondent have providing similar services?			
9)	For h	low many public agencies has Respondent provided similar services?			
,		many public agencies has respondent provided similar services:			

(11) Please attach a short history of the Respondent including whether it is local, regional, national, or international as well as approximate number of employees.

Also provide the number of offices and locations.

- (12) Identify preconstruction and lease-leaseback construction services performed for other school districts in accordance with parameters described above.
- (13) Describe how Respondent has successfully provided preconstruction and lease-leaseback construction services such as those described herein.
- (14) Describe the unique or innovative preconstruction and lease-leaseback construction services utilized on previous projects.

SECTION B - LEGAL

<u>3⊑∪</u>	IION B	- LEGAL			
(15)	Have you or any of your principals been in any claim, litigation or arbitration of any kind on a question or questions relating to similar services involving a public entity during the prior five (5) years?				
	(a)	If yes, provide the name of the public agency and briefly detail the dispute:			
(16)		you ever had a service agreement terminated for convenience or cause in			
	the p	rior five (5) years?			
	(a)	If yes, provide details including the name of the other party:			
(17)	Resp	espondent, owners, and/or any principal or manager involved in or is ondent aware of any pending litigation regarding professional misconduct, aith, discrimination, or sexual harassment?			
	(a)	If yes, provide details:			

(18)	any pe	spondent, owners, and/or any principals or manager involved in or aware of ending disciplinary action and/or investigation conducted by any local, state eral agency?
	(a)	If yes, provide details:
(19)		Respondent comply with all District, local, state and federal legal ements, regulations and laws?
SEC ¹	TION C	- ADDITIONAL INFORMATION
(20)		e provide any other information that may assist the District in ascertaining qualifications, capability and customer service under any resultant ment.
SEC ¹	TION D	- CONFLICT OF INTEREST
(21)	conne	you ever had any direct or indirect business, financial or other ction/relationship with any official, employee or consultant of the District? by any conflict of interest in (a):
	(a)	Please elaborate and discuss any potential, apparent or actual conflict of interest:

SECTION E. OBJECTIONS TO LE The Firm is required to list any	ASE-LEASEBACK AGREEMENTS of and all objections to the terms in the Lease-
Leaseback Agreement in Atta include proposed replacemen District will not consider any p	chment 3 below. Moreover, the Firm must also t language (use separate pages if necessary). The roposed changes to the Lease-Leaseback ained herein or attached hereto.
	penalty of perjury under the laws of the State led in the foregoing Firm Questionnaire is true ar
Executed this day of _	, 20, at
State of	
Company Name	Signature

<u>ATTACHMENT 2 – FEE PROPOSAL</u>

Prepare this document for each Proposed Project to be considered. If you intend to submit a proposal for both Proposed Projects, you must submit this document for each Proposed Project, but you must also include backup materials <u>clearly</u> demonstrating what, if any, cost savings can be achieved by awarding you both Proposed Projects.

This fee proposal, "Attachment 2", must be submitted with the proposal. Use additional sheets, if necessary.

Project: <u>Carmel Del Mar School Modernization</u>
The Firm proposes the following fees:
1. The preconstruction fee should be expressed as a lump sum firm fixed price based on the construction budget, schedule, and descriptions in this RFP.
Preconstruction Fee: [LUMP SUM] \$
2. The lease-leaseback fee shall include the Firm's overhead, profit, and all other costs (excluding general conditions), and should be expressed as a percentage.
NOTE : The lease-leaseback fee is not the GMP. The GMP shall not exceed the final construction budget plus the below-listed lease-leaseback fee.
NOTE : Should the Firm try to revise the Contractor's fee so that it exceeds the percentage below, the Firm agrees and acknowledges that the District has the right to deem the Firm's proposal non-responsive, cancel the lease-leaseback contract(s) without owing any fees or costs to the Firm, and award a contract to another contractor/Firm.
Lease-Leaseback Fee:% of construction budget (\$10,000,000).
3. The general conditions should be expressed as a monthly rate based on the construction budget, schedule, and description in this RFP.
General Conditions: \$/month*
*Please provide as backup for the general conditions a line-item listing of the proposed monthly general conditions and the corresponding monthly cost that you propose using for any/each Proposed Project. Failure to include this backup line-

4. Total Fee Proposal:

item listing will impact your score for this category.

For purposes of scoring the fee proposals, the monthly fees will be multiplied by the number of estimated construction months to calculate the total fee for evaluation purposes and the total fee will be used to calculate the Best Value Score. The actual

be based on the final construction budget	and term of the lease-leaseback contract.
a) Preconstruction Fee: = \$	
b) Lease-leaseback Fee: % x \$1	0,000,000 =
c) General Conditions: \$/m	onth x 14 months =
TOTAL PROPOSED FEE: [1+2+3] State numbers: \$	• •
Executed this day of	, 20
Company Name	Signature
Title	Print Name
Project: <u>Ashley Falls School Moderni</u>	ization_
The Firm proposes the following fees:	
1. The preconstruction fee should be expithe construction budget, schedule, and de	ressed as a lump sum firm fixed price based on escriptions in this RFP.
Preconstruction Fee: [LUMP SUM] \$	
2. The lease-leaseback fee shall include (excluding general conditions), and should	the Firm's overhead, profit, and all other costs d be expressed as a percentage.

Lease-Leaseback Fee and General Conditions price for the Firm awarded a contract will

NOTE: Should the Firm try to revise the Contractor's fee so that it exceeds the percentage below, the Firm agrees and acknowledges that the District has the right to deem the Firm's proposal non-responsive, cancel the lease-leaseback contract(s) without owing any fees or costs to the Firm, and award a contract to another contractor/Firm.

NOTE: The lease-leaseback fee is not the GMP. The GMP shall not exceed the final

construction budget plus the below-listed lease-leaseback fee.

Lease-Leaseback Fee:%	of construction budget (\$13,500,000).
3. The general conditions shou construction budget, schedule, an	ld be expressed as a monthly rate based on the d description in this RFP.
General Conditions: \$	/month*
proposed monthly general cond	the general conditions a line-item listing of the litions and the corresponding monthly cost that you posed Project. Failure to include this backup line ore for this category.
4. Total Fee Proposal:	
number of estimated construction purposes and the total fee will be Lease-Leaseback Fee and Gener	proposals, the monthly fees will be multiplied by the on months to calculate the total fee for evaluation used to calculate the Best Value Score. The actua al Conditions price for the Firm awarded a contract will budget and term of the lease-leaseback contract.
d) Preconstru	uction Fee: = \$
e) Lease-leaseback Fee:	% x \$13,500,000 =
f) General Conditions: \$	/month x 14 months =
TOTAL PROPOSED FEE: [1+2+3 numbers: \$	3] State the total proposed fee in both words and
Executed this day of _	, 20
Company Name	Signature
Title	Print Name

ATTACHMENT 3 – AGREEMENT FORMS

(Agreement Forms are also on file with the District Office)

1. Site Lease	
2. Leaseback Agreement	
3. Construction Services Agreement	

ATTACHMENT 3- AGREEMENT FORMS



_____ SCHOOL MODERNIZATION PROJECT SITE LEASE

By and Between

DEL MAR UNION SCHOOL DISTRICT

Dated as of _____

and

______ School Modernization Project
Site Lease

SCHOOL MODERNIZATION PROJECT SITE LEASE

				School Mod								
("District")		blic sc		istrict organ						e of Cal		nd the
					desig	nated a	as entity	numb	oer	•	• •	y the
California S	Secret	ary of	State	and the les	ssee herein	. The Di	strict and	d the C	ontract	or may	be refer	red to
herein indi	vidua	lly as a	"Part	y" and colle	ctively as th	he "Part	ies."					
					REC	CITALS						
A.		Califor	nia Ed	ducation Cod	de Section 1	17406 pe	ermits the	e gover	ning bo	ard of a	school d	listrict
instrument	any t by w	perso hich s	on, firi uch pr	m, or corpo roperty is lea sed premise	oration any ased requir	real pres the le	operty o	wned constru	by the ct or pr	school ovide fo	district, or constri	if the uction
and provid	es tha	at title	to the	e improvemeng school fa	ents shall ve	est in th	e school d	district	at the e	xpiratio	n of the	lease.
B. Moderniza	tion	Projec	t (the	desires to e "Project" ated at 130	') using th	ne leas	e-leaseba	ick cor	nstruction	on deli	very me	
_	-			of California ched to this			of the Pr	oject ('	'Project	Site") i	s depicte	ed and
exception of including a unless the Quality Act Services Agusing the I	reement" ("L of the ony co Project ("CE greement to the	ents t easeb Schoo Preco nstruct is ap QA"). ient (c leaseb e Cons	hat areack Age on the proveduction of the policetic oack construction oach constru	y with enter re entitled: greement" ernization P ction Work to of the Project d and all app ourpose of to vely, the "Ll onstruction on Services	(i) "or "LBA"); Project" ("Coto be provided, shall not provals received this Site Lead LB Agreemed delivery management	and (ii) onstruct ded per occur, leived foll ase, the ents") is ethod a	School "Construction Service Exhibit A be authorous contraction owing contraction provide the a general school of the a general "Construction of the a general school of the general school of the a g	ol Mode uction ces Agre of the rized or mpletic ck Agre de for the de by Ed	ernization Service reemen CSA, the ran oblement, ne constitution cription	on Projos Agree t" or "Come remain igation in the contraction in the c	ect Leasement for SA"). Wining action of the Die Construction of the Posection 1 Project.	seback or the th the ivities, istrict, nental uction Project 17406.
interests of authorized Site Lease; establishing the work a will meet t	dition f the by Eq (ii) le g add nd se the Di	s preconstruction of the second secon	edent t to co on Coo the Pr terms requir s expe	s Governing set forth in construct the de Section 1 roject and the sand condition to be pectations. The and interrelations is set of the sand interrelations in the sand interrelations is set of the sand interrelations in the sand interrelations is set of the sand interrelations in the sand interrelations is set of the sand interrelations in the sand interrelations is set of the sand interrelations in the sand interrelations is set of the sand interrelations in the sand interrelations is set of the sand interrelations in the sand interrelations is set of the sand interrelations in the sand interrelations is set of the sand interrelations in the sand interrelations is set of the sand interrelations in the sand interrelations is set of the sand interrelations in the sand interrelations is set of the sand interrelations in the sand interrelations is set of the sand interrelations in the sand interrelations is set of the sand interrelations in the sand interrelations is set of the sand interrelations in the sand interrelations is set of the sand interrelations in the sand interrelations is set of the sand interrelations in the sand interrelations is set of the sand interrelations in the sand interrelations in the sand interrelations in the sand inte	n this Site Le Project using 17406 by: (i) the Project Stations for corperformed by the Project, the Project, the Project, the Project Stations for corperformed by the Project Stations for the Pro	ease an ing the I leasing leasing leasing lite back instruction the Co e Partie	d the oth ease-leas the Proj t from the on of the entractor s intend t	ner LLB seback ect Site e Contr Project under t that: (i)	Agreen constru to the factor u under the the LLB	nents, it ction de Contrac nder the he CSA Agreem 3 Agreen	t is in the elivery m ctor unde e LBA; an to ensur nents ("V ments sh	e best ethod er this nd (iii) re that Vork") nall be
					_	1_						

take effect; and (iii) any capitalized terms used, but not defined, in this Site Lease shall have the meanings ascribed to such terms in the other LLB Agreements.

E. The Contractor is duly licensed in the State as a general building contractor, is qualified and experienced in construction of the type of educational facilities included in the Project and is willing to perform the Work.

Now, in consideration of the foregoing and of their respective rights and obligations under the LLB Agreements, the Parties hereby agree as follows:

AGREEMENT

Section 1. Lease to Contractor. Subject to the provisions of this Site Lease, the District hereby leases the Project Site to the Contractor, and the Contractor hereby leases the Project Site from the District. During the Site Lease Term (defined in Section 3 herein), the Contractor shall have a leasehold interest in the Project Site, including, without limitation, any and all improvements thereto made in accordance with the LLB Agreements.

Section 2. Effective Date. With the exception of the "Preconstruction Work" (defined in Exhibit A to the CSA), but otherwise notwithstanding anything to the contrary, the LLB Agreements shall have no force or effect until the date all of the following events have occurred ("Effective Date"): (i) each Party has approved, signed, and delivered all of the LLB Agreements; and (ii) the District has completed CEQA review and obtained any and all approvals required to commence the Work. Such required approvals may include, but are not limited to, any approvals by: (i) the California Department of Education ("CDE"); (ii) the Department of General Services, Division of State Architect ("DSA"); (iii) the State Allocation Board ("SAB"); (iv) the Office of Public School Construction ("OPSC"); and/or (v) the Department of Toxic Substances Control ("DTSC").

Section 3. Construction Period and Site Lease Term. Upon satisfaction of the conditions precedent set forth in Section 2 herein, and if any and all other applicable conditions set forth in the LLB Agreements have been satisfied, the District, in accordance with Section 4 of the CSA, shall issue a notice to the Contractor to proceed with the Work ("Notice to Proceed"). Subject to contrary dates in the Notice to Proceed, the Work must commence on ________ ("Required Commencement Date") and must be fully and satisfactorily completed no later than _______ ("Required Completion Date"). The Contractor must complete each portion of the Work that is subject to a milestone completion date not later than the applicable milestone completion date. The term of this Site Lease ("Site Lease Term") shall commence on the Required Commencement Date and (unless the Site Lease Term is extended, or this Site Lease is earlier terminated as provided in any of the LLB Agreements) shall expire concurrently with the Required Completion Date or any other earlier termination under the LLB Agreements.

Section 4. Possession and Use by Contractor. The Contractor may use the Project Site solely for: (i) purposes related to performance of the Work in accordance with the LLB Agreements; and (ii) leasing the Project Site and the Project back to the District. The Contractor and, to the extent reasonably necessary for such purposes, its officers, employees, subcontractors, consultants, agents and other representatives (collectively, "Contractor Representatives") shall have the right at all reasonable times during the Site Lease Term to enter in and upon the Project Site for purposes of performing the Work, examining and inspecting the Work, and undertaking any and all actions necessary and/or convenient regarding the Work. Except for such authorized uses and any uses by the District in accordance with the

LLB Agreements, the Contractor shall not allow any other use of the Project Site or facilities on the Project Site. Except as provided in the LLB Agreements, the Contractor, at its own expense, must obtain or otherwise have in effect all permits, licenses and similar authorizations for its activities on and in the Project Site and the facilities on the Project Site. The Contractor must not use, operate or maintain the Project Site or facilities on the Project Site improperly or carelessly, and, during the Site Lease Term, the Contractor must comply with all applicable laws and any applicable orders or rules of any governmental entity with competent jurisdiction over the Project Site and/or the Project. However, nothing shall be deemed to prohibit the Contractor from non-compliance with any such law, order or rule during any action or other proceeding undertaken in good faith by the Contractor to contest the validity or application of the law, order or rule, but the Contractor shall indemnify, defend, and hold-harmless the District Indemnitees as provided in the CSA with respect to any such non-compliance.

Section 5. Performance of Work. The Contractor shall: (i) perform the Work in strict accordance with the LLB Agreements and other documents incorporated therein or approved in accordance therewith; and (ii) complete the Work within the Site Lease Term, as it may be modified from time to time in accordance with the LLB Agreements. The compensation payable to the Contractor for performance of the Work shall be determined as provided in the LBA and CSA.

Section 6. Quiet Enjoyment. Except as permitted in the LLB Agreements, the District shall not take any action during the Site Lease Term to prevent the Contractor's quiet enjoyment of the Project Site for the purposes authorized under this Site Lease. In the event any challenge to the District's fee title to the Project Site results in or is likely to result in interference with the Contractor's right to occupy or use the Project Site for such purposes, the District shall use such of its governmental powers at its disposal, including the power of eminent domain, as necessary to obtain unencumbered fee title to the Project Site. The District shall defend the Contractor's right to so occupy and use the Project Site for the purposes set forth in the LLB Agreements. No use of the Project Site or the facilities on the Project Site by the District that is contemplated in or by the LLB Agreements shall be deemed or construed to constitute interference with the Contractor's right to occupy and use the Project Site.

Section 7. No Waste or Illegal Acts. At all times that it occupies or is otherwise in possession of the Project Site or any portion thereof, the Contractor shall not commit, suffer, or permit any waste of the Project Site or the facilities on the Project Site, and the Contractor shall not negligently, willfully, or knowingly use or permit the use of the Project Site or the facilities on the Project Site for any illegal act or purpose.

Section 8. Payment of Rent. The Contractor shall have no obligation to pay rent unless and until
the conditions precedent set forth in Section 2 herein have been satisfied. Within sixty days following the
Effective Date, the Contractor shall pay to the District the total amount of \$ [TYPICALLY
\$1 PER YEAR – SO IF PROJECT WILL TAKE 2 YEARS, CHARGE \$2], which shall constitute rent for the lease
of the Project Site by the Contractor for up to months from the Effective Date. If the Site Lease Term
for any reason exceeds months, then, within thirty days following the end of the initialmonth
period, the Contractor must pay to the District the amount of \$1.00 for each year (or portion thereof) that
the Site Lease remains in effect following the initialmonth period. If, for any reason, this Site Lease
is terminated after the Effective Date, the Contractor shall not be entitled to a refund of any rent paid to
the District under this Section.

- 3 -

Section 9. Responsibility for Taxes. The Contractor, at its sole expense, shall be responsible for payment of all taxes, assessments, fees, and other charges as are attributable to actions by the Contractor or any of the Contractor Representatives in connection with the performance of the Work. In addition, the Contractor, at its own expense, shall be solely responsible for payment of any and all taxes on or measured by: (i) the Contractor's income in connection with the transaction(s) contemplated in the LLB Agreements; and (ii) the operation of the Contractor's business. If the terms of this Site Lease result in the creation of a taxable possessory interest, then, to the extent such possessory interest tax is levied on the Contractor, and notwithstanding the foregoing provisions of this Section, but subject to receipt of sufficient documentary evidence of the obligation and amount, the District shall pay or otherwise satisfy such possessory interest tax.

Section 10. Title to Property. Notwithstanding anything to the contrary, upon expiration of the Site Lease Term or other termination of this Site Lease, title to the Project Site and all facilities on the Project Site, including, without limitation, any and all improvements constructed by the Contractor under the LLB Agreements, shall fully vest in the District. Title to improvements constructed on the Project Site under the LLB Agreements shall progressively and proportionally vest in the District as the Contractor is paid under the LLB Agreements, but title shall not fully vest in the District until the Required Completion Date, as such date may be extended.

Section 11. District Conveyance of Interests In Project Site. During the Site Lease Term, the District shall not mortgage, sell, assign, transfer or otherwise convey the Project Site or any interest therein to any person or entity without first obtaining the written consent of the Contractor, if such action by the District would unreasonably impair any of the Contractor's rights under the LLB Agreements. The foregoing shall not be deemed or construed to preclude the District from granting utility easements, rights-of-way, or similar interests in the Project Site to facilitate the use and operation of the Project or as required by any governmental or quasi-governmental agency.

Section 12. District Right of Entry. At any and all times during the Site Lease Term, the District and/or its officers, employees, and other agents or representatives may enter in and upon the Project Site and the Project for inspection or other purposes related to the Work, or arising in connection with the operation and management of the Project Site and/or the Project. The District shall not, however, unreasonably interfere with the Contractor's right to use and occupy the Project Site and the Project. The foregoing shall be deemed and construed to include the right of the District and its students to occupy and use completed phases of the Work for educational and related purposes. The Contractor shall cooperate and coordinate with the District as necessary to facilitate such uses by the District.

Section 13. District Representations and Warranties. The District represents and warrants to the Contractor that, to the best of the District's knowledge:

- (i) The District has good and merchantable fee title to the Project Site and has authority to enter into, and perform its obligations under the LLB Agreements;
- (ii) The District is unaware of any tax, assessment or imposition of any kind applicable to the Project Site that has not been paid in full, excluding any current and future taxes, assessments or impositions that may be assessed in regard to the Project Site;

- 4 -

- (iii) The District is unaware of any pending or threatened litigation involving the Project Site or the facilities on the Project Site that might affect the ability to use them for the intended purposes of the LLB Agreements;
- (iv) Except for asbestos, lead, or other hazardous materials that the District anticipates may be present in or incorporated into existing improvements on and to the Project Site and the facilities on the Project Site, if any such existing improvements/facilities exist, the District is not aware of any dangerous, toxic or hazardous pollutants, contaminants, chemicals, waste, materials or substances, regulated pursuant to applicable local, State or federal environmental laws, rules, or regulations, that are located in, on or about the Project Site and that would subject the Contractor to any damages, penalties or liabilities under any such applicable local, State or federal environmental laws, rules, or regulations ("Environmental Laws"); and
- (v) The District has received no notice, from any governmental entity with competent jurisdiction, of any currently existing or alleged violation of any of the Environmental Laws with respect to the Project Site or the facilities on the Project Site.

Section 14. Contractor Representations and Warranties. The Contractor represents and warrants to the District that, to the best of the Contractor's knowledge:

- (i) The Contractor is duly organized, validly existing, and in good standing in accordance with the laws of the State, and has full power, authority, and legal right to lease and otherwise acquire and convey interests in real and personal property;
- (ii) The Contractor is aware of the requirements of Education Code section 17406, Education Code section 17407.5, Public Contract Code section 2600 *et seq.*, and Public Contract Code section 20111.6. Public Contract Code section 20111.6 requires that a person, firm, or corporation that constructs the building(s), including, but not limited to, the prime contractor and electrical, mechanical, and plumbing subcontractors be prequalified. Such prequalification is applicable to each public project projected to cost \$1,000,000 or more, regardless of the source of funds to pay for the project (this subdivision (ii) collectively referred to as the "Prequalification Requirement");
- (iii) The Contractor, any person, firm, or corporation that constructs the building, including, but not limited to, the prime contractor, and each MEP subcontractor with one of the specialty licenses specified in Public Contract Code Section 20111.6 has been prequalified under procedures adopted by the District Board;
- (iv) The Contractor has full power, authority and legal right to enter into and perform its obligations under the LLB Agreements, and the execution, delivery and performance thereof has been duly authorized by all necessary actions on the part of the Contractor and does not require any further approvals;
- (v) Execution, delivery and performance of the LLB Agreements does not and/or will not result in any breach of or constitute a default under any indenture, mortgage, contract, agreement or instrument to which the Contractor is a party or by which it or its property is bound; and

- 5 -

(vi) There are no pending or threatened actions or proceedings before any court or administrative agency that may materially and adversely affect the ability of the Contractor to perform its obligations under the LLB Agreements.

Section 15. Termination. Notwithstanding anything to the contrary, the District may unilaterally and immediately terminate the LLB Agreements by giving written notice to the Contractor, if any challenge to the validity of any of the LLB Agreements is filed in any court of competent jurisdiction, in which event neither Party shall thereafter have any obligation to the other Party under the LLB Agreements, except that, upon written request of the Contractor in such event, the District shall pay to the Contractor such amounts as would be payable under Public Contract Code Section 5110 if it were applicable. Upon expiration of the Site Lease Term or other termination of this Site Lease: (i) the Contractor shall quit and surrender the Project Site and the facilities on the Project Site, which must be delivered to the District in the condition as of the date of expiration or other term; (ii) to the extent provided in the LLB Agreements, the District shall remain obligated to compensate the Contractor for the performance of the Work completed as of the date of expiration or other termination; and (iii) in accordance with the LLB Agreements, the Contractor shall release, remove or cure any stop notices, liens and/or other encumbrances arising from the Work.

Section 16. Events of Default. Notwithstanding anything to the contrary, any Claim within the scope of Part 22 of the General Provisions: (i) shall be subject to the requirements of such Part 22; and (ii) shall not be subject to this Section or Section 17 of this Site Lease. For all purposes of the LLB Agreements other than any such Claims, each of the following events with respect to a Party shall be deemed a default by such Party of its obligations under the LLB Agreements (each an "Event of Default"):

- (i) The Party fails, within the time required, to pay any undisputed or unexcused payment due or asserted to be due to the other Party, or any undisputed or unexcused portion of a payment due or asserted to be due to the other Party under any of the LLB Agreements; or
- (ii) The Party fails to perform or observe any covenant, condition or agreement to be performed or observed by such Party under any of the LLB Agreements, and such failure materially and adversely affects the other Party's rights; or
- (iii) Any statement, representation or warranty by the Party and set forth in any of the LLB Agreements is determined to be intentionally misleading or erroneous in any material respect and materially and adversely affects the other Party's rights; or
- (iv) The Party makes any general assignment for the benefit of creditors, any voluntary or involuntary petition for bankruptcy or for reorganization pursuant to federal bankruptcy law is filed and not withdrawn or dismissed within sixty days of filing, a court of competent jurisdiction appoints a trustee or receiver to manage or control all or substantially all of the Party's assets, all or substantially all of the Party's assets are subject to attachment, execution or other judicial seizure, or a court of competent jurisdiction determines that such Party has become insolvent or unable to pay its debts when due.

Section 17. Notice and Opportunity to Cure. If a Party is alleged to be responsible for an Event of Default ("Defaulting Party"), the other Party ("Non-Defaulting Party") may provide written notice thereof to the Defaulting Party, specifying in reasonable detail the nature and extent of the alleged default ("Notice of Default"). If the Defaulting Party has not cured the Event of Default within twenty days after

- 6 -

receipt of the Notice of Default in the case of a monetary default (i.e., failure to pay money or secure the payment of money), or within thirty days after receipt of the Notice of Default in the case of a non-monetary default, the Non-Defaulting Party in its discretion may initiate the dispute resolution provisions set forth in Section 18 of this Site Lease. The giving of a Notice of Default and allowing the period for cure of the Event of Default in accordance with this Section shall be a condition precedent to the Non-Defaulting Party exercising any available remedy in response to the Event of Default. Nothing shall be construed to prohibit the Defaulting Party from disputing that an Event of Default has occurred. Neither the giving of any Notice of Default, nor the initiation by the Non-Defaulting Party of any dispute resolution, legal or equitable action, or other proceeding in connection with an Event of Default, shall by itself operate to terminate this Site Lease.

Section 18. Informal Attempts at Dispute Resolution. Except for any Claims subject to Part 22 of the General Provisions, if a dispute arises out of or directly or indirectly relates to any of the LLB Agreements ("Dispute"), the Parties shall attempt as provided in this Section to resolve the Dispute as quickly and as amicably as possible, including, without limitation, any Disputes as to the meaning of any provision of the LLB Agreements, the validity of any determination or calculation required under the LLB Agreements, or the rights or obligations of the Parties under the LLB Agreements. If the Dispute does not relate to an Event of Default or is not of such nature that a Party may give a Notice of Default, then the Party alleging the Dispute shall give to the other Party a written notice of the Dispute ("Notice of Dispute"). Within a reasonable time, not in excess of seven calendar days, after receipt of either a Notice of Default or a Notice of Dispute, the Parties shall commence attempts to informally resolve the Dispute as required under this Section. Such attempts shall include good-faith, reasonable, and diligent efforts by both Parties to communicate and, if possible, to reconcile or compromise their respective positions. The participation by a Party in such attempts to informally resolve a Dispute shall be a condition precedent to such Party exercising any available remedy in response to the Dispute. If, after diligently making the attempts required under this Section for at least thirty calendar days, the Parties cannot resolve a Dispute, either Party may give written notice to the other Party that the attempts have been unavailing and, therefore, such informal attempts have been terminated effective upon receipt of the notice.

Section 19. Exercise of Available Remedies. If attempts at informal resolution of a Dispute under Section 18 are terminated without the Dispute having been resolved to the satisfaction of a Party, the Party may initiate any legal or equitable action or other proceeding in response to the Dispute that is available under applicable law and the LLB Agreements. In addition, however, if a Party fails to respond to, or participate in good faith in, any requests or requirements for attempts at informal resolution of the Dispute, the other Party, in its discretion, without needing to further comply with Section 18 of this Site Lease, may initiate any legal or equitable action or other proceeding in response to the Dispute that is available under applicable law and the LLB Agreements. However, in any case in which a Notice of Default has been provided under Section 17 of this Site Lease, no such legal or equitable action may be initiated until the applicable period for cure of the Event of Default under such Section 17 has expired without cure of the Event of Default.

Section 20. Scope and Limitations of Available Remedies. Except as provided in this Section or other applicable provisions of the LLB Agreements, each Party may exercise any or all available legal or equitable rights in response to an unresolved Dispute. However, at all times while any informal disputeresolution attempts, legal or equitable actions, or other authorized proceedings relating to a Dispute are pending, each Party, except to the extent provided in this Section, shall continue to perform its obligations under the LLB Agreements. Notwithstanding the foregoing, a Party shall not be responsible for continued performance of its obligations under the LLB Agreements to the extent an Event of Default by the other

- 7 -

Party makes such performance impossible, impractical, or unreasonable. Except as expressly provided in any of the LLB Agreements, a Party shall be entitled to termination of any of the LLB Agreements in response to an uncured Event of Default only if a court of competent jurisdiction determines that damages and specific performance are not sufficient, appropriate, or available remedies. Any remedies available to a Party shall not be deemed exclusive, and the Party may exercise any remedy individually or in combination with any other remedy it has available.

Section 21. Indemnification by Contractor. With respect to any and all claims, demands, causes of action, other proceedings, losses, damages, costs, expenses, attorneys' fees, and other liabilities of any nature that arise in connection with, or are directly or indirectly related to, the Contractor's use and occupancy of the Project Site under this Site Lease, the Contractor shall indemnify, defend, and hold-harmless the District to the extent provided in the CSA. For purposes of the LLB Agreements, such liabilities include, without limitation, any and all matters arising out of any use, disposal, discharge or other release on or about the Project Site of any hazardous materials, substances, or waste. For purposes of this Section, the reference to the District shall be deemed and construed to be a reference also to the District Board and each member thereof, the District's other officers, employees, and agents, and each of them. Notwithstanding the District's above-referenced right to be indemnified for attorney's fees, no person, entity, or party, including but not limited to the District and the Contractor, shall be allowed to recover attorney's fees that are incurred to enforce or defend this Site Lease.

- **Section 22. Giving of Notice.** Any notice given by a Party under this Site Lease must be given in accordance with Section 23.10 of the General Provisions in the CSA.
- **Section 23. Governing Law and Venue.** The terms and provisions of this Site Lease shall be construed in accordance with the laws of the State. Any action, arbitration, mediation or other proceeding arising from this Site Lease shall be initiated and conducted only in the County.
- **Section 24. Modifications.** This Site Lease may not be amended or otherwise modified except by written agreement duly-approved by the District Board, signed, and delivered by the Parties.
- **Section 25. Recitals and Exhibits.** The Recitals set forth herein, and all Exhibits referenced herein and attached hereto, are hereby incorporated as operative and effective provisions of this Site Lease.
- **Section 26. Counterparts.** This Site Lease may be signed in one or more counterparts, which, taken together, shall constitute one and the same original instrument. Signature pages may be detached from counterpart originals and combined to physically form one or more original copies of this Site Lease bearing original signatures of both Parties.
- **Section 27. Due Authority of Signatories.** Each person signing this Site Lease represents and warrants that he or she has been duly authorized by appropriate action of the Party he or she represents to execute, and thereby bind such Party to, this Site Lease.

(Remainder of page left blank.)

- 8
School Modernization Project
Site Lease

In Witness Whereof, the Parties have executed this Site Lease as evidenced by the signatures, below, of their respective duly-authorized representatives.

Del Mar Union School District		
Pv:		
By:	Ву:	
Name:	Name:	
Title:	Title:	

EXHIBIT A Description and Depiction of Project Site

The Project Site includes the areas outlined, highlighted or otherwise designated in the description and depiction below.

[Insert Site Description & Depiction]



_____ SCHOOL MODERNIZATION PROJECT LEASEBACK AGREEMENT

By and Between

DEL MAR UNION SCHOOL DISTRICT

and

Dated as of ______

_____ SCHOOL MODERNIZATION PROJECT LEASEBACK AGREEMENT

This School Modernization Project Leaseback Agreement ("Leaseback Agreement" or "LBA"), dated for reference purposes as of, is made by and between the Del Mar Union School District ("District"), a public school district organized and existing under the laws of the State of California and the lessee herein, and ("Contractor"), a designated as entity number by the California Secretary of State and the lessor herein. The District and the Contractor may be referred to herein individually as a "Party" and collectively as the "Parties."
RECITALS
A. Concurrently with entering into this Leaseback Agreement, the Parties also entered into those certain agreements entitled: (i) " School Modernization Project Site Lease" ("Site Lease"); and (ii) "Construction Services Agreement for the School Modernization Project" ("Construction Services Agreement" or "CSA"). The Site Lease, this LBA and the CSA are defined in the Site Lease, collectively, as the "LLB Agreements."
B. The purpose of the LLB Agreements is to facilitate completion of the Project using the lease-leaseback construction delivery method authorized by Education Code Section 17406 by: (i) leasing the Project Site to the Contractor under the Site Lease; (ii) leasing the Project and the Project Site back to the District under this LBA; and (iii) setting forth additional terms and conditions for construction of the Project in the CSA, to ensure that the Work will meet the District's expectations. Therefore, the Parties intend that: (i) the LLB Agreements shall be interpreted as an integrated and interrelated set of agreements; (ii) the LLB Agreements shall concurrently take effect; and (iii) any capitalized terms used, but not defined, in this LBA shall have the meanings ascribed to such terms in the other LLB Agreements.
Now, in consideration of the foregoing and of their respective rights and obligations under the LLB Agreements, the Parties hereby agree as follows:
AGREEMENT
Section 1. Leaseback to the District. Subject to all provisions herein, the Contractor hereby leases to the District the Project Site and the Project, together with all improvements thereto, whether existing or made during the Site Lease Term, and the District hereby leases the Project Site and the Project from the Contractor. Notwithstanding anything to the contrary, such leaseback to the District shall not cause or result in a merger of the District's leasehold estate under this LBA and its fee estate as lessor under the Site Lease, and the Contractor, subject to the provisions of the LLB Agreements, shall continue to have and hold a leasehold estate in the Project Site under the Site Lease.
Section 2. Effective Date and Leaseback Term. This Leaseback Agreement shall have the same Effective Date as the Site Lease. Subject to contrary dates in the Notice to Proceed, the term of this Leaseback Agreement ("Leaseback Term") shall commence on the Required Commencement Date and, unless this LBA is terminated earlier as provided in any of the LLB Agreements, the Leaseback Term shall expire on the Required Completion Date, unless such time is extended by the Parties in writing.
- 1 -

Section 3. Possession and Use by District. During the Leaseback Term and subject to any limitations set forth in the LLB Agreements, the District may possess and use the Project Site and the Project: (i) to examine and inspect the Work; (ii) to inspect, maintain, and operate the Project Site and Project; (iii) for purposes of its educational, recreational, and other programs; and (iv) for other reasonable purposes of the District. The District, its officers, employees, contractors, consultants and agents (collectively, the "District Agents"), and others as necessary or convenient for District purposes, shall have the right at all reasonable times during the Leaseback Term to enter in and upon the Project Site in connection with such District uses. However, during the Leaseback Term, the District shall not permit any such use of the Project Site and/or the Project in a manner that unreasonably interferes with the Contractor's rights under the Site Lease to possess and use the same. The Parties anticipate that the District may conduct educational, recreational or other activities on or in one or more portions of the Project Site at times when Work is not occurring, but no such activities shall be deemed or construed to constitute unreasonable interference with such rights of the Contractor. Except as may be provided in the LLB Agreements, the District, during the Leaseback Term and at its own expense, must obtain or otherwise have in effect all permits, licenses and similar authorizations for its activities on, in, and at the Project Site. During the Leaseback Term, the District shall hold fee title to the Project Site and will obtain from the Contractor the fee title to the improvements constructed under the LLB Agreements, which shall vest progressively as payments are made in accordance with the LLB Agreements, but shall not fully vest until the Required Completion Date, unless such time is extended by the Parties in writing. During the Leaseback Term, the Contractor shall have a leasehold interest in the Project Site as provided in the Site Lease.

Section 4. Leaseback Payments. The District shall pay to the Contractor such lease payments as are specified in the Schedule of Leaseback Payments attached as Exhibit "A" to this LBA (each a "Leaseback Payment"). Each Leaseback Payment shall be due and payable as specified in the Schedule of Leaseback Payments. The Contractor shall separately invoice the District for the Leaseback Payments and the Construction Progress Payments due under the General Provisions, and each Party shall maintain adequate records distinguishing between Leaseback Payments and Construction Progress Payments. The District shall pay the undisputed portions of each such Leaseback Payment invoice within thirty (30) days following receipt of the invoice. The District, in its sole discretion, and at any time and without penalty, may prepay any one or more of the Leaseback Payments.

Section 5. Adequate Consideration. The Parties hereby acknowledge that payment by the District of the Leaseback Payments shall be full and adequate consideration for the District's right to use of the Project Site and the Project during the Leaseback Term. The Parties have determined and agreed that the Leaseback Payments are not in excess of the fair rental value of the Project Site and Project, and, in making such determination, the Parties have considered, among other matters: (1) the fair market value attributable to use of the improvements to be completed under the LLB Agreements; and (2) the District's limited access and use of the Project Site during construction of the Project.

Section 6. Nature of Payment Obligation; No Acceleration. The obligation of the District to pay Leaseback Payments under this LBA shall constitute a current expense of the District. Nothing in this LBA shall be deemed or construed to constitute: (i) debt of the District in contravention of any applicable constitutional or statutory limitations; or (ii) a pledge of the general tax revenues, general funds or general monies of the District. Notwithstanding anything to the contrary, in no event shall the Contractor have the right to require accelerated payment of any Leaseback Payment that is not then due or in default, or to otherwise declare any such payment immediately due and payable.

Section 7. Abatement of Leaseback Payments. Notwithstanding any damage to or destruction of the Project Site or the Project, or any other substantial interference with the District's rights to possess and use the Project Site and the Project under this LBA, this LBA shall continue in full force and effect. However, in addition to any other conditions on the District's obligation to make Leaseback Payments, any Leaseback Payments due hereunder shall be subject to abatement at any time during the Leaseback Term in which, due to material damage to or destruction of the Project Site or the Project, or for any other reason not the fault of the District, there is substantial interference with the District's rights under this LBA to possess and use the Project Site and/or the Project, or any substantial portion thereof. For each occurrence of potentially-substantial interference with such rights, the District shall reasonably determine: (i) whether abatement shall apply; (ii) the date upon which abatement shall commence; (iii) the portion of the Leaseback Payments to be abated; and (iv) the date such abatement ends. The amount of abatement shall be such that the Leaseback Payments paid by the District during the abatement period do not exceed the fair rental value of the portions of the Project Site and/or the Project that the District reasonably may continue to use in accordance with this LBA.

Section 8. District Failure to Pay Leaseback Payments. Except as may be provided in the LLB Agreements, the District's obligation to timely pay the Leaseback Payments to the Contractor shall be absolute and unconditional, and not be subject to any set off. Subject to the foregoing, if the District fails within the time permitted by this LBA to pay the undisputed portion of any Leaseback Payment, then, notwithstanding any other right of the Contractor under this LBA, such unpaid amount shall accrue interest at two percent (2%) simple interest per annum until paid.

Section 9. District Responsibility for Encumbrances and Taxes. During the Leaseback Term, the District shall keep the Project Site and the Project free of all newly-imposed levies, liens, and encumbrances other than those that: (i) are normally or reasonably incident to the ownership of real property; (ii) are necessary in connection with the maintenance and operation of the Project Site and/or the Project; or (iii) arise from actions of the Contractor or any of the Contractor Representatives. During the Leaseback Term, the District shall timely pay all taxes, assessments, and similar charges that now or hereafter may duly be imposed upon the District's ownership, leasing, renting, possession, or use of the Project Site and/or the Project.

Section 10. Responsibility for Utility Services. Except as may be provided in the LLB Agreements, the District shall contract for and pay the expenses of all utility services required for its possession and use of the Project Site and the Project, including, but not limited to, all electricity, gas, water, and sewer.

Section 11. Maintenance and Security. The District, at its expense, shall be responsible for the maintenance and security of such portions of the Project Site and the Project of which it has sole possession and use during the Leaseback Term. Except for the foregoing, the Contractor, at its expense, shall have sole responsibility for the maintenance and security of the Project Site and the Project during the Leaseback Term. Notwithstanding the other provisions of this Section, after District Board acceptance of the Project in accordance with Section 16.8 of the General Provisions, and except for warranty and other obligations of the Contractor under the LLB Agreements, the District shall have sole responsibility for maintenance and security of the Project Site and the Project.

Section 12. Indemnification of Contractor. The District shall indemnify, defend and hold-harmless the Contractor with respect to any and all claims, demands, causes of action, other proceedings, losses, damages, costs and expenses (attorneys' fees), and other liabilities of whatever nature as arise from, or are directly or indirectly related to, the District's occupancy and use of the Project Site and Project

during the Leaseback Term. The scope of the District's obligations under this Section shall be restricted to liabilities caused by third parties subject to the direct supervision and control of the District, including, without limitation, the District's other contractors, subcontractors, students, and other invitees. Any and all indemnification obligations arising in connection with the performance of the Work shall be governed by Part 21 of the CSA. Notwithstanding anything to the contrary, the District shall not be obligated under this Section to the extent any claim, demand, cause of action, other proceeding, loss, damage, cost and expense (attorneys' fees), or liability within the scope of this Section is attributable to the negligence or willful misconduct of the Contractor or any of its subcontractors, employees, agents, assigns, or other representatives. Notwithstanding the above-referenced right to be indemnified for attorney's fees, no person, entity, or party, including but not limited to the District and the Contractor, shall be allowed to recover attorney's fees that are incurred to enforce or defend this Leaseback Agreement.

Section 13. District Insurance. At all times that the District is occupying any portion of the Project during the Leaseback Term, the District, at its expense, shall have in effect liability and property insurance providing coverage in such amounts, and subject to such terms, as the District determines will adequately protect the District's interests in the Project Site and the Project. The District may, in its sole discretion, obtain and maintain any such insurance through an Owner Controlled Insurance Program ("OCIP") and/or through a joint-powers self-insurance cooperative of which the District is a member. The District shall cause such insurance to be endorsed to name the Contractor as an additional insured during the period required by this Section, and the District shall provide to the Contractor, upon request in writing, one or more certificates of insurance and endorsements evidencing the insurance coverage required by this Section. With respect to matters for which the District is responsible under Section 12 herein, and during the period the insurance coverage is required under this Section to be in effect, the District's insurance shall be primary and any coverage provided by the Contractor shall be secondary and non-contributing.

Section 14. Events of Default. For purposes of Section 15 herein, the term "Event of Default" shall mean any of the following:

- (i) The District fails to pay to the Contractor the undisputed portion of any Leaseback Payment within thirty (30) days after notice from the Contractor that such amount is overdue;
- (ii) The District fails to perform or otherwise comply with any of its material obligations under this LBA within thirty (30) days following written notice from the Contractor specifying in detail such failure by the District; and
- (iii) The District becomes insolvent, is unable to pay its debts as they become due, makes an assignment to creditors, applies or consents to the appointment of a receiver, trustee, conservator or liquidator of the District or of all or a substantial part of its assets, or a petition for relief is filed by the District under federal bankruptcy, insolvency or similar laws, and the District fails within sixty (60) days thereafter to cause such appointment or filing to be lifted or to otherwise make financial arrangements for payment of Leaseback Payments as are reasonably satisfactory to the Contractor.

Section 15. Termination. This LBA shall terminate automatically upon: (i) receipt by the Contractor of all Leaseback Payments due under this LBA; and (ii) expiration of the Leaseback Term. This LBA shall also terminate automatically upon earlier termination of this LBA, at which point no further Leaseback Pay shall be due. The Contractor may terminate this LBA, by giving written notice to the District, in the event the District fails to cure an Event of Default specified in clause (i) or clause (ii) of Section 14

within twenty (20) days. Subject to applicable bankruptcy or other law, the Contractor may terminate this LBA, by giving written notice to the District, in the event the District fails, within the applicable period of time permitted under Section 14 herein, to cure an Event of Default specified in clause (iii) of that Section. Nothing in this LBA shall be deemed or construed to preclude the District from seeking damages from the Contractor or other remedies for wrongful termination.

Section 16. Additional Representations and Warranties. In addition to the representations and warranties of the Parties set forth in the Site Lease, the Contractor further represents and warrants that:

- (i) To the best of its knowledge, and except to the extent of any asbestos, lead, or other hazardous materials that may be present in, or incorporated into, the existing improvements on the Project Site, if any such existing improvements exist, the Contractor is not aware of any hazardous materials, substances or wastes that are located in, on or about the Project Site that would subject the District to any damages, penalties or liabilities under any Environmental Laws;
- (ii) The Contractor has received no notice, from any governmental entity with competent jurisdiction, of any currently existing or alleged violation of any Environmental Laws with respect to the Project Site or any facilities on the Project Site; and
- (iii) Except to the extent permissibly incorporated into the Work in accordance with the LLB Agreements, the Contractor shall keep the Project Site and the facilities on the Project Site free of hazardous materials, substances, and wastes.
- **Section 17. Giving of Notice.** Each notice from one Party to the other given under this LBA must be given in accordance with Section 23.10 of the General Provisions.
- **Section 18. Governing Law and Venue.** The terms and provisions of this LBA shall be construed in accordance with the laws of the State, notwithstanding any choice-of-law, conflict-of-law, or other provision in any federal, state or other law. Any action, arbitration, mediation, or other proceeding arising from this LBA shall be initiated and conducted only in the County.
- **Section 19. Modifications.** This LBA may not be amended or otherwise modified except by written agreement duly-approved by the District Board, signed, and delivered by the Parties.
- **Section 20. Recitals and Exhibit.** The Recitals set forth herein, and Exhibit A referenced herein and attached hereto, are hereby incorporated as operative and effective provisions of this LBA.
- **Section 21.** Counterparts. This LBA may be signed in one or more counterparts, which, taken together, shall constitute one and the same original instrument. Signature pages may be detached from counterpart originals and combined to physically form one or more original copies of this LBA bearing signatures of both Parties.
- **Section 22. Due Authority of Signatories.** Each person signing this LBA represents and warrants that he or she has been duly authorized by appropriate action of the Party he or she represents to execute, and thereby bind such Party to, this LBA.

- 5 -

Title:

In Witness Whereof, the Parties have executed this LBA as evidenced by the signatures, below, of

Title:

EXHIBIT "A" SCHEDULE OF LEASEBACK PAYMENTS

Project:	School Modernization	Project
	School Wiederinzacion	0,00

Payment Month *	Beginning Balance	Payment	Ending Balance
1	\$[Initial amount]		
2			
3			
4			
5			
6			
7			
8			
9			
10			
11			
12			\$ 0.00
Total			

* The District's obligation to make Leaseback Payments commences within 10 business days following the Required Commencement Date. The Leaseback Payment shall be thereafter paid on the same day in each subsequent month. For example, if the District pays the first Leaseback Payment on the 23rd of the month, each Leaseback Payment shall be paid on or before the 23rd of the next month. The final Leaseback Payment shall be paid as above or on the Required Completion Date, whichever is earlier.



CONSTRUCTION SERVICES AGREEMENT

FOR THE

By and Between

DEL MAR UNION SCHOOL DISTRICT

and

Dated as of _____

CONSTRUCTION SERVICES AGREEMENT FOR THE SCHOOL MODERNIZATION PROJECT

This Construction Services Agreement for the School Modernization Project ("Construction Services Agreement" or "CSA"), dated for reference purposes as of, is made by and between the Del Mar Union School District ("District"), a public school district organized and existing under the laws of the State, and ("Contractor"), a designated as entity number by the California Secretary of State. The District and Contractor may be referred to herein individually as a "Party" and collectively as the "Parties."
RECITALS
A. Concurrently with entering into this Construction Services Agreement, the Parties also entered into those certain agreements entitled: (i) "School Modernization Project Site Lease" ("Site Lease"); and (ii) " School Modernization Project Leaseback Agreement" ("Leaseback Agreement" or "LBA"). The Site Lease, the LBA and this CSA are defined in the Site Lease, collectively, as the "LLB Agreements."
B. The purpose of the LLB Agreements is to facilitate completion of the Project using the lease-leaseback construction delivery method authorized by Education Code Section 17406 by: (i) leasing the Project Site to the Contractor under the Site Lease; (ii) leasing the Project and the Project Site back to the District under the LBA; and (iii) setting forth additional terms and conditions for construction of the Project under this CSA to ensure that the Work will meet the District's expectations. The Parties intend, therefore, that: (i) the LLB Agreements shall be interpreted as an integrated and interrelated set of agreements; (ii) the LLB Agreements shall concurrently take effect; and (iii) any capitalized terms used, but not defined, in this CSA shall have the meanings ascribed to such terms in the other LLB Agreements.
Now, in consideration of the foregoing and of their respective rights and obligations under the LLB Agreements, the Parties hereby agree as follows:
AGREEMENT
Section 1. Effective Date. With the sole exception of Preconstruction Work, this Construction Services Agreement: (a) is subject to the approvals referred to in Recitals B and C and Section 2 of the Site Lease including, but not limited to the District's completion of its analysis under CEQA; and (b) shall have the same Effective Date as the Site Lease and the Leaseback Agreement.
Section 2. Scope of Work. The scope of the Preconstruction Work and the Work to be completed by the Contractor under the LLB Agreements is generally described in Exhibit "A" attached to this CSA. The Contractor must complete the Preconstruction Work and Work in strict accordance with the Contract Documents.
Section 3. Component Parts of Contract.
(a) The Contract. This Construction Services Agreement is but one of the agreements and other documents that collectively set forth the complete understanding and agreement of the Parties
- 1 - School Modernization Project

with respect to the performance of the Preconstruction Work and the Work (the "Contract"). Each of the Contract Documents, as those may be made or duly amended from time to time, is hereby incorporated as an operative and effective part of the Contract and LLB Agreements. In connection with the execution and delivery of this CSA, and within the times required by this CSA, the Contractor must complete, execute, and deliver to the District each of the various Contract-related forms set forth in Exhibit "B" attached to this CSA ("Required Contract Forms"). As and when required by this CSA during the course of the Work, the Contractor must complete, execute, and deliver to the District the various Project-related forms set forth in Exhibit "C" attached to this CSA ("Required Project Forms"). The Contractor may obtain copies of the Required Contract Forms and the Required Project Forms from the District.

- (b) The Contract Documents. The "Contract Documents" include, but are not limited to: (i) the Site Lease, the Leaseback Agreement, and this CSA; (ii) any and all drawings, plans, elevations, sections, details, schedules, and diagrams that illustrate any or all of the Work ("Drawings") and any and all written requirements for materials, equipment, construction systems, quality workmanship, services, and other things to be furnished in connection with the Work or the Project ("Specifications") approved for the Project, including, to the extent applicable, by the DSA in accordance with Education Code Sections 17280 through 17316, inclusive; (iii) any and all supplemental drawings, profiles, cross-sections, specifications and other documents approved and/or issued by the Architect for purposes of illustrating or describing the location, character, dimensions, or details of the Work; (iv) the general provisions of this CSA attached as Exhibit "D" to this CSA ("General Provisions"); (v) any and all performance and payment bonds, work schedules, schedules of value, guarantees, and other construction-related documents prepared, provided, approved and/or adopted in connection with the Project and/or Work; (vi) any and all duly-authorized agreements or duly-authorized orders providing for changes in the Work; (vii) any and all forms and other documents duly-submitted by the Contractor in connection with the Work; and (viii) the District's Request for Proposals and the proposal submitted by the Contractor for the Project.
- (c) **Complementary Nature.** The Contract Documents shall be deemed and construed to be complementary and an integrated whole. Any requirement or provision set forth in one Contract Document, although not set forth in any one or more of the other Contract Documents, shall be interpreted as if set forth in or applicable to all Contract Documents.

Section 4. Notice of Award and Notice to Proceed.

- (a) **Notice of Award.** The LLB Agreements require that, prior to commencing the Work, the Contractor must provide to the District certain bonds, schedules, forms, and other documentation. To assist in ensuring that the District receives such documentation, the District may issue to the Contractor a written notice that the District has awarded the Project/Contract to the Contractor and of the District's intent to subsequently issue a Notice to Proceed ("Notice of Award"). The Notice of Award, if any, will specify the anticipated date on which the District will issue a Notice to Proceed, and the District shall issue the Notice of Proceed in accordance with the provisions of Sections 2 and 3 of the Site Lease. HOWEVER, IF THE DISTRICT DOES NOT ISSUE A NOTICE OF AWARD, THE CONTRACTOR SHALL NONETHELESS BE OBLIGATED, AS REQUIRED BY THIS CSA, TO SUBMIT ALL SUCH DOCUMENTATION TO THE DISTRICT PRIOR TO COMMENCING THE WORK.
- (b) **Notice to Proceed.** The District shall issue to the Contractor a written Notice to Proceed as soon as practicable after all conditions precedent thereto are satisfied. Although the satisfaction of the conditions precedent to issuance of the Notice to Proceed may not be completely within the control of either Party, each Party must make reasonable efforts to ensure that such conditions

precedent are timely satisfied so that the District may issue the Notice to Proceed in time to permit construction to commence in accordance with Section 5 of this CSA. Notwithstanding anything to the contrary, if such conditions precedent are not satisfied concurrently for all portions of the Work (including, but not limited to, because the DSA will be issuing approvals for the Project in increments), the District may issue separate Notices to Proceed for distinct portions of the Work as such conditions precedent are satisfied for such portions of the Work. If, after issuance of any Notice to Proceed, the LLB Agreements are terminated following a challenge to the validity of the LLB Agreements as provided in Section 15 of the Site Lease, any Notice to Proceed shall be deemed to have automatically been rescinded upon such termination.

Section 5. Commencement and Completion of Work.

- (a) **Start and Finish Dates.** Subject to contrary dates in any one or more Notice to Proceed, the Contractor must: (i) commence the Work on the Required Commencement Date; and fully and satisfactorily have completed all of the Work not later than the Required Completion Date. The Contractor must complete each portion of the Work that is subject to milestone completion date(s) not later than the applicable milestone completion date(s). If, despite the reasonable efforts of the Parties, the District is for any reason unable to issue a Notice to Proceed in time to permit the Work to commence on the Required Commencement Date, the District shall issue the Notice to Proceed as soon thereafter as practicable.
- (b) **Extensions of Time.** The Contractor must commence and perform the Work with such continuous reasonable diligence as will ensure completion of the Work within the time permitted under the Contract ("Contract Time"). The Contract Time may be extended as provided in the Contract Documents or as the Parties otherwise may agree in writing.
- (c) **Scheduling Goals.** The Contractor acknowledges that the District has scheduled the Work for a specific period of time, in order to promote the best usage of school facilities, to promote an appropriate learning environment for students, and to avoid and/or minimize interference with the District's educational, recreational, and other programs. The Contractor further acknowledges that compliance with scheduling requirements for the Work is mandatory in order to accomplish such goals and, therefore, that time is of the essence with respect to the performance of the Work.
- (d) **Project on Hold during Preconstruction Work.** Notwithstanding anything to the contrary, if there is a delay while in the preconstruction phase based upon, but not limited to, a lack approvals, the District Board may make findings that it is in the District's best interest that the Project be paused. If the District Board makes such findings, the District shall give written notice to Contractor and Contractor shall perform no further services and/or shall not charge the District for services occurring after the date of such written notice. If the District Board does not restart the Project within six (6) calendar months following the notice pausing the Project, either the Contractor or the District may terminate the LLB Agreements in their entirety, and without penalty. No further compensation will be owed to Contractor other than for the Preconstruction Work already duly performed.

Section 6. Compensation to Contractor.

(a) **Guaranteed Maximum Price.** The total compensation payable to the Contractor under the LLB Agreements shall not exceed the cost for the Preconstruction Work and a specific maximum amount for the Work in exchange for which, as described in more detail in Subsection (f) of this Section,

the Contractor guarantees that it will perform all of its obligations under the LLB Agreements in strict accordance with the Contract Documents (the "Guaranteed Maximum Price" or "GMP"). The Preliminary GMP and/or Final GMP payable to the Contractor under the LLB Agreements is set forth and itemized in Exhibit "E" attached to this CSA. The GMP shall include District payments to the Contractor of Construction Progress Payments and Leaseback Payments.

- (b) **Preliminary Estimated GMP.** The preliminary estimate of the GMP payable to the Contractor pursuant to the LLB Agreements ("Preliminary GMP") is set forth and itemized in Exhibit "E" attached to this CSA. The Contractor represents and warrants that it has reviewed the Preliminary GMP and has confirmed that, subject to adjustment as described in Subsection (c) of this Section, the amounts comprising the Preliminary GMP are sufficient and appropriate compensation to the Contractor for performance of the Work. However, as noted, the Preliminary GMP is an estimate and it may vary from the GMP determined in accordance with Subsection (c) of this Section that, as adjusted in accordance with the Contract Documents, will be payable to the Contractor under the LLB Agreements.
- **Determining the Final GMP.** After completion of the Preconstruction Work, the (c) approval of the plans, Drawings, and Specifications by DSA, and the competitive process required pursuant to this CSA for selection of the Subcontractors that will perform portions of the Work, the Parties shall determine the total amount of the GMP payable to the Contractor pursuant to the LLB Agreements ("Final GMP"). The Final GMP shall be based on adjustment of the Preliminary GMP to reflect: (i) the difference between the estimates of the Subcontractors' costs included in the Preliminary GMP and the actual Subcontractors' costs determined through the applicable competitive process; and/or (ii) such other factors as the Parties may agree in writing. Except as the Parties so may agree, the other components of the Preliminary GMP (including, without limitation, the cost of any Work to be self-performed by the Contractor) shall be deemed sufficient compensation to the Contractor and, although, in determining the Final GMP, such components may be decreased as agreed by the Parties, such components shall not be increased. Education Code section 17406, subdivision (a)(3), requires District Board approval of the Final GMP at public meeting before the Contractor can commence the Work. If the District Board does not approve the amount of the Final GMP as determined pursuant to this Subsection for any reason including, but not limited to, if the District Board determines that it does not or will not have sufficient funding to pay the Final GMP, the District may provide written notice to the Contractor to thereby terminate the LLB Agreements, in which case neither Party shall have any obligations to the other Party under the LLB Agreements and Contractor shall not be entitled to any additional funds except for those previously due to Contractor for the Preconstruction Work. The Final GMP, as it may be adjusted pursuant to this CSA, shall be deemed sufficient compensation to the Contractor for full and satisfactory completion of the Project and performance of its other obligations pursuant to the LLB Agreements, including, without limitation, the Contractor's guarantee as described in Subsections (a) and (f) of this Section.

(d) **Contingencies and Allowances Included in GMP.**

(1) **Project Contingency.** The GMP shall include a contingency amount to be held by, and for the benefit of, the District ("Project Contingency") in the maximum amount specified in Exhibit "E" to this CSA. As directed by the District, the Project Contingency shall be used to pay costs for which the District is or might be responsible and that are attributable to: (i) errors, omissions, or other discrepancies in the Drawings and Specifications; (ii) unforeseen changes in any applicable federal, State or local laws, ordinances, codes, rules, regulations, standards, orders, and other requirements of any governmental and quasi-governmental entities with competent jurisdiction (each a "Legal Requirement"); and (iii) fires, windstorms, floods, earthquakes, or other acts of nature resulting in a natural disaster for which no person(s) can be held responsible. Any of the foregoing shall not qualify to be paid using the

Project Contingency unless it materially affects and increases the cost and performance of the Work. Any portion of the Project Contingency remaining after completion of all Work shall be returned to the District entirely.

- (2) **Contractor Contingency.** The GMP shall include a maximum contingency amount to cover reasonably unanticipated Contractor issues ("Contractor Contingency") in the amount specified in Exhibit E to this CSA. Upon prior approval by the District in writing, the Contractor Contingency may be used to pay unanticipated costs associated with performance of the Work or items reasonably necessary for the Contractor to complete the Project. Any portion of the Contractor Contingency remaining after completion of all Work shall be considered a savings to the District and shall be returned to the District.
- (3) **Allowances.** To the extent any line-item included in the GMP consists of an "allowance" for a particular cost component of the Project: (i) if the total cost for such component ultimately exceeds the "allowance" amount, the difference shall, as directed by the District in writing, be paid using available Project Contingency or otherwise shall be the District's responsibility; and (ii) if the total cost of such component ultimately is less than the "allowance" amount, the difference shall be a credit to the District or otherwise serve to reduce the GMP. Each proposed allocation, use, or expenditure of any portion of any such "allowance" shall be subject to the advance written approval of the District. An "allowance" shall only be used for items listed in Exhibit "E" to this CSA.
- (e) Adjustment and Payment of the GMP. The GMP shall be subject to increase and/or decrease as provided in the Contract Documents. The District shall pay the total amount of the adjusted GMP to the Contractor, in partial payments from time to time, as provided in the General Provisions. Under no circumstances shall the sum total of the Construction Progress Payments (including, without limitation, payments of Retention) and Leaseback Payments made to the Contractor ever exceed the GMP. Upon completion of all Work, the Contractor shall provide a "final" change order to the District that establishes the total cost of the Work, including, without limitation, amounts attributable to sharing of savings, and the difference, if any, between such total cost and the GMP as adjusted under the Contract Documents.
- (f) Limitation on Compensation. The District's sole and exclusive liability for compensating the Contractor for full and satisfactory performance of its obligations under the LLB Agreements and other Contract Documents shall be deemed and construed to be limited to an amount equal to the GMP as it may be adjusted in accordance with the Contract Documents. The Contractor hereby represents and guarantees that it can and shall satisfactorily and completely perform all of its obligations under the LLB Agreements, including, without limitation, performing all Work in strict accordance with the Contract Documents, without seeking funds from the District in excess of the GMP, as may be adjusted, or requesting a redesign or change in scope of the Work in order to reduce the Contractor's costs, and regardless of any anticipated or unanticipated increases in costs of labor, materials, equipment, or other services or things necessary in connection with the Work. The Contractor shall be solely responsible for any and all costs it incurs in performing its obligations under the LLB Agreements and other Contract Documents that are in excess of the GMP, as may be adjusted, without right to reimbursement from the District.
- (g) **Prepayment Option.** The District shall have the option at any and all times during the performance of the Work, in its sole discretion, to prepay some or all of the Construction Progress Payments. Notwithstanding anything to the contrary, no election by the District to exercise such option

and no such prepayment by the District shall be deemed or construed to release the Contractor from any of its obligations under the LLB Agreements.

- (h) **Obligation to Minimize Cost of Project.** The Parties have entered into the LLB Agreements with the understanding that, through use of the lease-leaseback construction delivery method, they will make reasonable efforts to minimize the cost of the Project and, with respect to such efforts, will cooperate with each other and their respective consultants and/or subcontractors. When planning and preparing to undertake construction of the Project, and during the course of construction of the Project, the Contractor shall make reasonable attempts to identify and implement measures, construction techniques, and administrative procedures as will assist in minimizing the cost of the Project. As applicable, such measures, techniques, and procedures shall include, but are not limited to:
- (i) Identify and suggest to the District any value-engineering or other design changes as may result in a lower overall Project cost;
- (ii) Maintain forward-looking perspective sufficient to identify and resolve early any design, materials availability, and other issues, thereby avoiding adverse impacts on the critical path of the Project;
- (iii) Efficiently schedule, track, and use rented and owned equipment;
- (iv) Purchase in advance any materials anticipated to be subject to price increases;
- (v) Appropriately secure, track and use materials to prevent losses and unnecessary waste, and ensure that materials are acquired only if needed for the Project;
- (vi) Allocate only such administrative assistance and other direct and indirect overhead costs as are not in excess of what is reasonably necessary for the Project;
- (vii) Efficiently administer and use "general conditions," falsework, and other temporary facilities;
- (viii) Award multiple Subcontracts for a particular component of the Work if additional competitiveness among Subcontractors would be likely; and
- (ix) Ensure Work is ready for inspection prior to scheduling inspection, thereby minimizing reinspections.

The Parties have entered into the Contract with the understanding that: (i) the Contract shall not be deemed or construed to be a "stipulated sum" or "fixed price" contract; (ii) the Contractor's records relating to costs of the Project shall be accessible by the District at all reasonable times on an "open book" basis; and (iii) the Contractor shall be entitled to compensation under the Contract up to the amount of the GMP, as may be adjusted, but only for such costs as reasonably are attributable to and necessary for the Work. Nothing in this Subsection (h) shall be deemed or construed to limit or otherwise condition the Contractor's obligations under Subsection (f) of this Section 6.

Section 7. Contractor Insurance.

(a) The Contractor must comply with the insurance-related requirements set forth in Exhibit "F" attached to this CSA. The Contractor must also ensure compliance by its Subcontractors with the applicable provisions of such insurance-related requirements.

- 6 -

(b) During the term of the Site Lease and Leaseback Agreement, and until the subsequent return of the Project and Project Site to the District under the terms of the LLB Agreements, and only for third party Bodily Injury and/or Property Damage loss occurrences, Contractor shall be an additional insured for exposure as an owner or lessee of the project site and not for exposure as a builder or in any other capacity.

Section 8. Contractor Registration.

- (a) **Labor Code Requirements.** No contractor or subcontractor may engage in the performance of a public work contract unless the contractor or subcontractor is currently registered with the California Department of Industrial Relations ("DIR") and qualified to perform public works under Labor Code Section 1725.5.
- (b) Applicability to Contract. The Contractor shall be responsible for ensuring that it and all Subcontractors are currently and properly registered with the DIR and qualified to perform public work under Labor Code Section 1725.5. Prior to commencing the Work, the Contractor must complete, execute, and submit to the District the "Certification Regarding Contractor Registration" form included in the Required Contract Forms. Notwithstanding anything to the contrary, if at any time during the performance of the Work, the Contractor or any of its Subcontractors is not duly registered under Labor Code Section 1725.5 (including, without limitation, if the registration expires or the DIR revokes the registration), the District in its sole discretion may terminate the Contract and/or replace the Contractor or Subcontractor with a contractor or subcontractor that is duly registered under Labor Code Section 1725.5, at no cost to the District.

Section 9. Compliance with Labor Law Requirements. The Project is a "public works" project as defined in Section 1720 of the California Labor Code ("Labor Code") and, therefore, is subject to Part 7, Chapter 1, of the Labor Code and Title 8 of the California Code of Regulations, Section 16000 et seq. (collectively, "Labor Laws"). The Contractor acknowledges that, as provided by Senate Bill 854 (Stats. 2014, Ch. 28), the Project is subject to compliance monitoring and enforcement by the DIR. The Contractor, at no additional cost to the District, must: (i) comply with any and all applicable Labor Law requirements, including, without limitation, requirements for payment of "prevailing wages," inspection and submittal (electronically, as required) of payroll records, interview(s) of workers, et cetera; (ii) ensure that its Subcontractors are aware of and comply with the Labor Laws requirements; (iii) in connection with Labor Law compliance matters, cooperate with the DIR, the District and other entities with competent jurisdiction; and (iv) post all job-site notices required by law in connection with the Work, including, without limitation, postings required by DIR regulations. The District will coordinate and conduct any mandatory pre-construction conference, and the Contractor and each of its Subcontractors must attend the conference in order to ensure they are aware of applicable labor-law requirements. The General Provisions describe additional Labor Law requirements applicable to the Project. The following Labor Code sections are by this reference incorporated into and are a fully operative part of the Contract, and Contractor shall be solely responsible for compliance therewith:

- (i) Section 1735: Anti-Discrimination Requirements;
- (ii) Section 1775: Penalty for Prevailing Wage Rate Violations;
- (iii) Section 1776: Payroll Records;
- (iv) Sections 1777.5, 1777.6, and 1777.7: Apprenticeship Requirements;
- (v) Sections 1810 through 1812: Working Hour Restrictions;
- (vi) Sections 1813 and 1814: Penalty for Failure to Pay Overtime; and

(vii) Section 1815: Overtime Pay.

Section 10. Disabled Veterans Business Enterprises. The District has adopted a goal for participation in the Project by disabled veterans business enterprises ("DVBE") of three percent of the overall amount expended by the District for certain new-construction and modernization projects each year. The Contractor must make reasonable efforts to obtain and use DVBE services for performance of portions of the Work, whenever the opportunity arises prior to completion of the Work. Upon completion of the Work, and as a condition to final payment to the Contractor, the Contractor must complete, execute, and submit to the District the "Certification Regarding DVBE Participation" form included in the Required Project Forms.

Employee Background Checks. The District has considered the totality of the Section 11. circumstances relating to the Work and the Project (including, without limitation, whether minor-aged students may be present at some or all of the Project Site during performance of the Work) and has determined that, as of the Effective Date, the requirements for criminal-history background checks set forth in Section 8.2 of the General Provisions shall not initially be in effect. Notwithstanding the above, the District and Contractor agree that the District may, at any time during the performance of the Work, and at the District's sole discretion, require Contractor and any Subcontractor to comply with the requirements for criminal-history background checks set forth in Section 8.2 of the General Provisions. If the District requires such compliance, however, and upon request of the Contractor, the District (in its sole discretion and, in each case, with respect to specific circumstances) may consent to the Contractor implementing one or more alternatives to such background-check requirements as provided in Subsection 8.2.4 of the General Provisions. If the District requires such compliance, and the Contractor or any Subcontractor is a sole proprietor, the District shall prepare and submit the Contractor's/Subcontractor's fingerprints per Education Code section 45125.1. Any Contractor/Subcontractor that is a sole proprietor shall have an immediate and affirmative duty to inform the District of such status in order to allow the District to comply with Education Code section 45125.1.

Section 12. Regular Working Hours. For purposes of Section 11.2 of the General Provisions, "Regular Working Hours" shall mean any day, Monday through Friday, commencing at or after such time, and ending by or prior to such time, as may be specified in either an applicable local ordinance or any "Mitigation Monitoring Plan" adopted by the District under the California Environmental Quality Act, whichever is more restrictive.

Section 13. Liquidated Damages. The agreed amount of the liquidated damages, applicable if the Contractor is responsible for any delay in completion as provided in Section 11.8 of the General Provisions, shall be two thousand dollars (\$2,000) for each day or portion thereof.

Section 14. Copies of Drawings and Specifications. The District will provide one electronic copy of the Drawings and Specifications (in portable document format (or "pdf") or such other format as agreed by the Parties) to the Contractor at no charge. The Contractor shall be responsible, at its cost, for making such electronic or paper copies of the Drawings and Specifications as necessary for it to perform its obligations in connection with the Contract.

Section 15. Representatives and Contacts.

(a) **Authorized District Representatives.** The employees or consultants of the District who are authorized to represent the District in connection with the LLB Agreements for day-to-day communication and formal Notices (each an "Authorized District Representative") are identified in Exhibit

-8-

"G" attached to this CSA, which also specifies their address and contact information. Contractor acknowledges that the Authorized District Representatives do not have authority to approve change orders, modifications to the Contract Documents, authorize extra or out of scope work to be performed, or take any action to commit the District contractually. Any and all such actions must be brought to the District Board for its approval. The authority of the Authorized District Representatives is also subject to the District Board's power and to any limitations described in the LLB Agreements. The District may at any time change any of the Authorized District Representatives, and the District shall provide notice of any such change to the Contractor. Notwithstanding that the Project is being completed using the LLB construction delivery method, the District, in its sole discretion, may at any time contract for program management, construction management, project management and/or other consultant services in connection with the Project, in which event some of the District's and/or Architect's duties in regard to the Project, as well as additional duties in regard to the Project manager, or other consultant.

- (b) **Authorized Contractor Representatives.** The officers and staff of the Contractor who are authorized to represent the Contractor for purposes of the LLB Agreements (each an "Authorized Contractor Representative") are identified in Exhibit G to this CSA, which also specifies their address and other contact information. The Contractor hereby agrees that the Authorized Contractor Representatives have such authority to represent the Contractor as required under the LLB Agreements. Subject to any applicable provisions of the Contract Documents, the Contractor may change any of the Authorized Contractor Representatives, and the Contractor shall provide notice of any such change to the District.
- (c) **Architect Contacts.** The architect of record for the Project ("Architect") is identified in Exhibit G to this CSA. The officers and/or staff of the Architect who are authorized to represent the Architect in connection with the Project are also identified in Exhibit G, together with their address and other contact information. Subject to any applicable provisions of its agreement with the District, the Architect may change any of its representatives, and the Architect or the District shall provide notice of any such change to the Contractor. Notwithstanding the foregoing, if no Architect or Architect representative is specified in Exhibit G to this CSA, the District shall delegate any duties specified in the LLB Agreements as being the responsibility of the Architect, to the extent such duties are applicable to the Project.
- **Section 16. Modifications.** Except as may be permitted by the Contract Documents, this CSA may be amended or otherwise modified only by means of a written agreement duly-approved by the District Board, signed, and delivered by the Parties.
- **Section 17. Recitals and Exhibits.** The Recitals set forth herein and the Exhibits referenced herein and attached hereto are hereby incorporated as operative and effective provisions of this CSA.
- **Section 18.** Counterparts. This CSA may be signed in one or more counterparts, which, taken together, shall constitute one and the same original instrument. Signature pages may be detached from counterpart originals and combined to physically form one or more original copies of this CSA bearing original signatures of both Parties.
- **Section 19. Contracts regarding the Russian Federation.** The Contractor may be subject to Executive Order N-6-22, dated March 4, 2022, which was issued by the Governor of California. Executive Order N-6-22 ("EO") involves sanctions in response to the military conflict between the Russian Federation and the Ukraine. The Contractor will be responsible for compliance with the EO and shall submit with its

proposal the "Certification regarding Executive Order N-6-22 by Governor of the State of California Regarding Contracts and Sanctions on the Russian Federation" form.

Section 20. Compliance with California Air Resources Board Regulations. The California Air Resources Board ("CARB") has implemented regulations for the use of certain vehicles on public works projects. For any project awarded after January 1, 2024, and under the regulations (see Cal. Code Regs., tit. 13, § 2449), the District must obtain valid Certificates of Reported Compliance ("CRC") from all contractors and listed subcontractors before awarding a project. The District will be required to retain CRCs for three years after the project is complete, and the CRCs are subject to review by the CARB upon five calendar days' notice. The Contractor and Subcontractors shall comply with the foregoing requirements and shall submit the "California Air Resources Board Compliance Form."

Section 21. Due Authority of Signatories. Each person signing this CSA represents and warrants that he or she has been duly authorized by appropriate action of the Party he or she represents to execute, and thereby bind such Party to, this CSA.

In Witness Whereof, the Parties have executed this CSA as evidenced by the signatures of their authorized representatives below.

Del Mar Union School District		
Ву:	Ву:	_
Name:	Name:	_
Title:	Title:	_
Date Signed:	Date Signed:	

EXHIBIT A DESCRIPTION OF WORK

The Project and the LLB Agreements contemplate both preconstruction services and construction services. While the construction services (defined below and referred to in the LLB Agreements as the "Work") will not be authorized until the requirements of Sections 2 and 3 of the Site Lease are met, the Parties intend that preconstruction services will begin upon authorization and full execution of the LLB Agreements.

Preconstruction Work

The preconstruction services (defined here and referred to in the LLB Agreements as the "Preconstruction Work") consists of reviewing the Proposed Project's Drawings and Specifications during the design of the Project to identify and note all deficiencies, incongruities and inconsistencies that may affect constructability of the Project including, but not limited to, design and specification omissions, incomplete and/or inconsistent plans, details and specifications, and any lack of coordination, together with all other appropriate, necessary and/or required services to facilitate and prepare for the successful development and construction of the Project.

The Preconstruction Work also includes, but is not limited to, the following tasks: (i) design meetings with the Architects and engineers ("A/E"); (ii) review and validation of estimates prepared by the Architect; (iii) preparation of a master critical path method schedule for the Project; (iv) value engineering; and (v) preparation of cost estimates or updated cost estimates as follows:

- (a) At the end of architectural schematic design and engineering basis of design reports. Cost estimating by Contractor will include life cycle cost analysis for building systems, which will allow District to decide direction for A/E team;
- (b) At the end of design development. Contractor will participate in A/E building information modeling clash detection process as constructibility advisor;
- (c) Construction documents 50% same as design development above, plus Contractor will estimate Final GMP for grading/utility construction at this point including allowances, contingencies, general conditions, costs and fees; and
- (d) Construction documents 100% same as design development above, plus Contractor will provide Final GMP for remainder of construction at this point including allowances, contingencies, general conditions, costs and fees;

Constructability reviews; value engineering; construction planning and phasing, and cost proposal strategies shall be performed and completed all with the goal that the DSA-approved plans and specifications for the Project will be complete such that the Project can be constructed by a competent licensed general building contractor in strict accordance with the DSA-approved plans and specifications without change orders, delays, or additional charges to District.

Contractor shall not provide any Preconstruction Work or Work that requires a contractor's license pursuant to Business and Professions Code section 7065 et seq. unless and until: (1) the District has received approval from DSA; and (2) the District has authorized Contractor to commence the Work as required by the LLB Agreements.

All Preconstruction Work must be individually priced out by line item and no Preconstruction Work may occur unless the District gives prior written consent.

The Work

The Work consists of all labor, materials, equipment, tools, utilities, temporary facilities, transportation, goods, and other services and things of any nature whatsoever as are expressly and impliedly necessary to timely and satisfactorily complete the Project in strict accordance with the LLB Agreements, the Drawings and Specifications prepared by the Architect and, to the extent applicable, approved by the DSA, and all other Contract Documents for the Project, including any Contract Documents providing for changes to the initial scope of the Work.

The Project consists, generally, of all of	the following: The complete modernization of the
School located at 13030	Drive, San Diego, CA 92130.

EXHIBIT B REQUIRED CONTRACT FORMS

- 1. Certification Regarding Workers Compensation
- 2. Payment Bond
- 3. Performance Bond
- 4. Certification Regarding Drug-Free Workplace
- 5. Certification Regarding Tobacco-Free Workplace
- 6. Certification Regarding Asbestos
- 7. Notice and Certification Regarding Lead-Free Materials
- 8. Certification Regarding Iran Contracting Act
- 9. Certification Regarding Contractor Registration

CERTIFICATION REGARDING WORKERS COMPENSATION

(Labor Code § 1861)

	ct: Del Mar Union Schoo ct:	
	ractor:	
	indersigned hereby certi ornia, that all of the follow	es to the District, under penalty of perjury under the laws of the State of ing is true and correct:
(i)	•	uly-authorized representative of the Contractor and, in that capacity, has ion on behalf of the Contractor; and
(iii)	employer to be insure in accordance with the	e of the provisions of Labor Code Section 3700 et seq., which require every against liability for Workers' Compensation or to undertake self-insurance provisions of the Labor Code, and the Contractor shall comply with such mencing and throughout the entirety of performance of the Work on the
Repre	esentative Name:	
Repre	esentative Title:	
Repre	esentative Signature:	
Date 9	Signed:	

PAYMENT BOND

District: Del Mar Union School District		
Project:		
Contractor:		
Surety:		
Bond #:	_	
Penal Sum:	(\$	_)

WHEREAS, the Project identified above is a public works project in accordance with applicable law, and the District identified above awarded to the Contractor identified above (herein, the "Principal") a contract providing for completion of the Project ("Contract"); and

WHEREAS, the Contract and/or California Civil Code Section 9550 *et seq.* require that the Principal furnish a labor and materials payment bond to the District that ensures the faithful performance of the Principal's obligations under the Contract to pay for materials, labor, and other things as required by law, which bond must have a penal sum equal to one hundred percent of the total amount payable by the District to the Principal under the Contract; and

WHEREAS, the Surety identified above hereby represents to the District that the Surety is an "admitted surety insurer" in accordance with Section 995.120 of the California Code of Civil Procedure;

NOW, THEREFORE, we, the Principal and Surety, are hereby held and firmly bound to the District in an amount of lawful money of the United States of America equal to the Penal Sum specified above, the payment of which well and truly to be made, we hereby bind, jointly and severally, ourselves and our heirs, executors, administrators, and successors.

THE CONDITION OF THIS OBLIGATION is that, if the Principal, or any of its heirs, executors, administrators, successors or assigns, or any of its subcontractors, fail in connection with the Project to pay as and when required (1) any of the persons authorized by California Civil Code Section 9100 to assert a claim against the Payment Bond, (2) amounts due under the Unemployment Insurance Code with respect to work or labor performed under the Contract, or (3) any amounts required to be deducted, withheld, and paid over to the Employment Development Department from wages of employees of the Principal or any of its subcontractors under Section 13020 of the Unemployment Insurance Code with respect to such work and labor, then the Surety shall pay, in full, any and all claims for such amounts, in a total amount not in excess of the penal sum set forth herein, and also, in case suit is brought upon this Payment Bond, such reasonable attorneys' fees as are fixed by the court.

This Payment Bond shall inure to the benefit of any and all of the persons named in California Civil Code Section 9100, so as to give a right of action to those persons or their assigns in any suit brought upon this Payment Bond.

The Surety hereby stipulates and agrees that this Payment Bond shall not be exonerated and the Surety shall not be released from its obligations under this Payment Bond on account of: (1) any extension of time, change, amendment or other modification of the Contract or other Project documents, or of the Principal's rights and/or obligations pursuant thereto, and Surety hereby waives notice of any and all such modifications; (2) any rescission or attempted rescission of the Contract or this Payment Bond; (3) any fraud of any person or entity other than fraud of a claimant in making a claim on this Payment Bond; or

(4) any breach of any contract by the District, the Principal, or any person or entity for whose benefit Surety has issued this Payment Bond. The Surety further stipulates and agrees that this Payment Bond shall be construed most strongly against the Surety and in favor of the persons and entities for whose benefit Surety has issued this Payment Bond.

The Surety shall not be deemed to have fully and appropriately executed this Payment Bond unless Surety has provided all of the following information:

SURETY'S SOUTHERN CALIFORNIA/LOCAL REPRESENTATIVE	SURETY'S REPRESENTATIVE FOR FILING CLAIMS
Company Name	Company Name
Street Address	Street Address
City, State, Zip Code	City, State, Zip Code
Representative Name	Representative Name
Representative Telephone Number	Representative Telephone Number

(The remainder of this page intentionally left blank.)

Each person that signs this Payment Bond on behalf of the Principal or the Surety thereby represents and warrants that the party he or she represents has duly authorized him or her to sign, and thereby bind such party to, this Payment Bond.

IN WITNESS WHEREOF, the Principal and Surety have executed this Payment Bond as evidenced by the signatures, below, of their respective, duly-authorized representatives.

PRINCIPAL	SURETY
Company Name	Company Name
Representative Signature	Representative Signature
Representative Name	Representative Name
Representative Title	Representative Title
Date Signed	Date Signed
<u>Principal</u> : Attach Notary acknowledgment to this Payment Bond and, if applicable, imprint corporate seal in the space below this line.	Surety: Attach Notary acknowledgment and power of attorney to this Payment Bond and imprint corporate seal in the space below this line.

PERFORMANCE BOND

District: Del Mar Union School District		
Project:		
Contractor:		
Surety:		
Bond #:		
Penal Sum:	(\$)

WHEREAS, the Project identified above is a public works project in accordance with applicable law, and the District identified above awarded to the Contractor identified above (herein, the "Principal") a contract providing for completion of the Project ("Contract"); and

WHEREAS, the Contract requires that the Principal furnish a bond to the District that ensures the faithful performance of the Principal's obligations under the Contract to fully and satisfactorily perform the work and services required by the Contract, which bond must have a penal sum equal to one hundred percent of the total amount payable by the District to the Principal under the Contract; and

WHEREAS, the Surety identified above hereby represents and warrants to the District that the Surety is an "admitted surety insurer" in accordance with Section 995.120 of the California Code of Civil Procedure;

NOW, THEREFORE, we, the Principal and Surety, are hereby held and firmly bound to the District in an amount of lawful money of the United States of America equal to the Penal Sum specified above, the payment of which well and truly to be made, we hereby bind, jointly and severally, ourselves and our heirs, executors, administrators, and successors.

THE CONDITION OF THE FOREGOING PAYMENT OBLIGATION is that, if the Principal (or, to the extent permitted by the Contract, its heirs, executors, administrators, successors or assigns) shall fully perform all of the undertakings, terms, covenants, conditions, agreements and other obligations required of the Principal under the Contract and any amendments or changes thereto, in full and strict conformance with the requirements thereof, and all within the time and in the manner designated therein, in all respects according to their true intent and meaning, then such obligation shall become null and void; otherwise, it shall remain in full force and effect.

A condition precedent to the satisfactory completion of the Contract is that, after the acceptance of the Project by the District, the payment obligation of this Performance Bond shall remain in full force and effect, in the penal sum set forth herein, during all periods in which the Principal has any obligations under the Contract (including, without limitation, the obligations to make full, complete, and satisfactory repair and replacements of any defective materials and/or faulty workmanship, to pay liquidated damages and/or to indemnify the District or any other party), and the obligation of Surety hereunder shall continue so long as any such obligation of Principal continues to exist. The obligations of Surety under this Performance Bond are exclusive of and distinct from any obligations Surety may have under any labor and materials payment bond applicable to the Project.

Whenever the District declares the Principal to be in default of its obligations under the Contract, the District having performed its obligations thereunder, the Surety, promptly within the time required by the Contract, shall remedy the default or, at the District's discretion, shall:

- (i) Complete the Project in strict accordance with the terms and conditions of the Contract, including, without limitation, provisions for the time(s) within which the Surety must act; or
- (ii) Obtain, or permit the District to obtain, one or more bids or proposals for any and all work required to complete the Project in strict accordance with the terms and conditions of the Contract, and upon determination of each lowest responsive and responsible bidder or proposer, arrange for such bidder or proposer and the District to enter into a contract, which may be in accordance with Education Code Section 17406, and make available as the work progresses sufficient funds, up to and including a total amount equal to the penal sum set forth herein less the balance of the contract amount, to pay the cost of completing the Project.

The term "balance of the contract amount," as used in the foregoing paragraph, shall mean the total amount payable by the District to Principal under the Contract and any amendments and changes thereto, less the amounts thereof previously paid by the District to the Principal.

In performing its obligations under this Performance Bond, the Surety expressly agrees that: (i) absent the express written consent of the District, the Surety shall neither use the Principal nor accept a bid or proposal from the Principal for purposes of completing the Project; and (ii) the District shall have the right, in its reasonable discretion, to reject any contractor or subcontractor that the Surety may propose to fulfill such obligations. In the event the District provides notice to the Surety that the Principal is in default of its obligations under the Contract and, therefore, the Surety is required, as provided herein, to complete the Project or to arrange for the District to contract for completion of the Project, and, through no fault of the District, the Surety has exceeded the time permitted under the Contract for doing so, the District may arrange to use replacement contractor(s) selected and contracted for by the District to complete the Project and, in such event, the Surety's payment and other obligations under this Performance Bond shall not be thereby voided, diminished, or otherwise limited.

Notwithstanding anything to the contrary, in the event the District determines that the Principal has not met, or likely will be unable to meet, any deadline required under the Contract, or that Principal's performance does not conform with the requirements of the Contract, the District may notify the Surety. In such event, the Surety must make reasonable attempts to assist the Principal to resolve or avoid the default by the Principal. The Surety and Principal expressly agree that neither the giving of such notice by the District nor the giving of such assistance by the Surety shall be deemed or construed to constitute interference by the District or the Surety with the Contract or the ability of the Principal to obtain any bond(s) in any amount(s) from any surety insurer(s).

For value received, the Surety hereby stipulates and agrees that this Performance Bond shall not be exonerated and the Surety shall not be released from its obligations under this Performance Bond by any change, amendment or other modification of the Contract or other Project documents, or of the Principal's rights and/or obligations pursuant thereto, and Surety hereby waives notice of any and all such modifications.

(The remainder of this page intentionally left blank.)

The Surety shall not be deemed to have fully and appropriately executed this Performance Bond unless Surety has provided all of the following information:

SURETY'S SOUTHERN CALIFORNIA/LOCAL REPRESENTATIVE	SURETY'S REPRESENTATIVE FOR FILING CLAIMS	
Company Name	Company Name	
Street Address	Street Address	
City, State, Zip Code	City, State, Zip Code	
Representative Name	Representative Name	
Representative Telephone Number	Representative Telephone Number	

(The remainder of this page intentionally left blank.)

Each person that signs this Performance Bond on behalf of the Principal or the Surety thereby represents and warrants that the party he or she represents has duly authorized him or her to sign, and thereby bind such party to, this Performance Bond.

IN WITNESS WHEREOF, the Principal and Surety have executed this Performance Bond as evidenced by the signatures, below, of their respective, duly-authorized representatives.

PRINCIPAL	SURETY	
Company Name	Company Name	
Representative Signature	Representative Signature	
Representative Name	Representative Name	
Representative Title	Representative Title	
Date Signed	Date Signed	
Principal: Attach Notary acknowledgment to this Performance Bond and, if applicable, imprint corporate seal in the space below this line.	Surety: Attach Notary acknowledgment and power of attorney to this Performance Bond and imprint corporate seal in the space below this line.	

CERTIFICATION REGARDING DRUG-FREE WORKPLACE

Project	: :	ar Union School District		
The un	dersigne	d hereby certifies to the District, under penalty of perjury under the laws of the State of all of the following is true and correct:		
(i)		The undersigned is a duly-authorized representative of the Contractor and, in that capacity, has executed this certification on behalf of the Contractor;		
(ii)		The Contractor, in accordance with Government Code Section 8350 <i>et seq.</i> , the Drug-Free Workplace Act of 1990, shall provide a drug-free workplace by doing all of the following:		
	(1)	Publishing a statement notifying employees that the unlawful manufacture, distribution, dispensation, possession, or use of a controlled substance is prohibited in the Contractor's workplace and specifying actions which will be taken against employees for violations of the prohibition.		
	(2)	Establishing a drug-free awareness program to inform employees about all of the following: (a) The dangers of drug abuse in the workplace; (b) The Contractor's policy of maintaining a drug-free workplace; (c) Availability of drug counseling, rehabilitation and employee-assistance programs; and (d) The penalties that may be imposed upon employees for drug abuse violations.		
	(3)	Requiring that each employee engaged in the performance of Work on the Project be given a copy of the statement required by subdivision (1), above, and that, as a condition of employment on the Project, the employee agrees to abide by the terms of the statement.		
(iii)	The Contractor hereby acknowledges and agrees that, if the District determines that either (a) the certification herein is false, or (b) the Contractor violated this certification by failing to carry out the requirements of Section 8355, then the Contract awarded to the Contractor shall be subject to termination, suspension of payments, or both, and the Contractor shall be subject to debarment in accordance with the requirements of Section 8350 <i>et seq</i> .			
Repres	entative	Name:		
Repres	entative	Title:		
Repres	entative	Signature:		
Date Si	gned:			

CERTIFICATION REGARDING TOBACCO-FREE WORKPLACE

Projec	t:	ar Union School District		
	_	ed hereby certifies to the District, under penalty of perjury under the laws of the State of tall of the following is true and correct:		
(i)	The undersigned is a duly-authorized representative of the Contractor and, in that capacity, has executed this certification on behalf of the Contractor; and			
(ii)	The Contractor shall ensure a tobacco-free workplace by doing both of the following:			
	(1)	Providing the following provision, in writing, to each person providing any labor or services on or at the Project Site, including, without limitation, any delivery personnel:		
		All properties and facilities operated by the Del Mar Union School District, including, without limitation, the Project Site, are tobacco-free work places. It is strictly forbidden while in, on, or at any District property or facility (whether owned or leased) to smoke, chew or otherwise use tobacco products, including e-cigarettes or so-called "vape" accessories. The Contractor shall require each person (including, without limitation, any employee of the Contractor or any subcontractor or supplier) found in violation of these requirements to permanently leave the Project Site, and the Contractor shall not thereafter permit such person to be present in, on or at the Project Site.		
	(2)	Enforcing the requirements specified in the foregoing provision.		
Repres	sentative	e Name:		
Repres	sentative	e Title:		
Repres	sentative	e Signature:		
Date S	igned:			

CERTIFICATION REGARDING ASBESTOS

Dist Proj		pel Mar Union School District							
Con	tract	or:							
The	unde	rsigned hereby certifies to the District, under penalty of perjury under California law, that:							
(i)	The undersigned is a duly-authorized representative of the Contractor and, in that capacity, has executed this certification on behalf of the Contractor.								
(ii)	The Contractor is aware and acknowledges that, for purposes of this certification: (a) asbestos is any of chrysotile, crocidolite, amosite, anthophyllite, tremolite, actinolite or other minerals generally known as asbestos; and (b) an asbestos-containing material is any material or thing, or any component thereof, that contains, consists of, or is made up of greater than one-tenth of one percent (0.1%) asbestos.								
(iii)) The Contractor shall not use on, or incorporate into, the Project any asbestos or asbestos-containing materials, including, without limitation, in any tools, devices, clothing, or equipment used in the construction of any portion of the Project.								
(iv)) The Contractor has instructed its employees and subcontractors in regard to such prohibition against asbestos and asbestos-containing materials, and in regard to the hazards, risks and liabilities involved in the use of asbestos and asbestos-containing materials.								
(v)	The	Contractor acknowledges and agrees that:							
	(a)	Each dispute as to whether any material, equipment or other thing used on, or incorporate into, the Work contains asbestos or is an asbestos-containing material shall be settled electron microscopy; and							
	(b)	The costs of any such tests shall be paid by the Contractor if the material is found to conta asbestos at a level greater than as specified herein; and							
	(c)	The District shall reject any and all materials or other things incorporated into the Work th are determined to contain asbestos or asbestos-containing materials, and the Contractor, at a cost to the District, must remove, replace and/or repair as necessary any and all affects portions of the Work.							
Rep	reser	tative Name:							
Rep	reser	tative Title:							
Rep	reser	tative Signature:							
Date	e Sigr	ed:							

NOTICE AND CERTIFICATION REGARDING LEAD-FREE MATERIALS

District: Del Mar Union School District
Project:
Contractor:

NOTICE:

If the Work is other than entirely new construction, the Work may disturb lead-containing or lead-based paint and other building materials that may be incorporated into existing buildings or other improvements located on the site for construction of the Project ("Project Site"). Until sampling and testing confirms otherwise, it shall be presumed that all school buildings and improvements built in 1992 or earlier contain lead-based or lead-containing paint. The Contractor must complete, sign and submit a copy of this Notice and Certification Regarding Lead-Free Materials to the District prior to commencing the Work.

The Contractor shall be responsible for ensuring that its employees and subcontractors fully and adequately comply with, and that the Work is performed in conformance with, all applicable laws, ordinances, rules and regulations governing lead-based or lead-containing paint and other materials, including, but not limited to: (i) Education Code Section 32240 *et seq.*; (ii) Title 8, California Code of Regulations, Section 35001, *et seq.*

If the Work involves renovation, modernization or other disturbance of any existing school buildings or improvements, the Contractor shall sample and test all materials as appropriate to determine whether lead-based paint or other materials are present and may present a hazard or threat during any such renovation or modernization. The Contractor shall provide the District with the results of any and all such testing, whether conducted prior to commencement of the Work, during the Work, or after completion of the Work.

Any and all Work that may result in the disturbance of lead-containing building materials must be coordinated through the District. Consistent with applicable law, when a lead-based hazard is identified, the Contractor shall utilize personnel certified by the California Department of Public Health to perform lead-related services. The District may at any time request that Contractor provide the training and certification records of each employee of the Contractor or subcontractor who provides lead-related services. The Contractor shall be solely responsible for proper disposal, in conformance with all applicable laws, of any and all lead-containing, lead-based or hazardous waste products including, but not limited to, paint chips, residue, and any other material that may be exposed or disturbed during the course of the Work.

California law prohibits, in the construction of any new school facility or in the modernization or renovation of any existing school facility, the use of lead-containing or lead-based paint, plumbing, solders, and other materials that may constitute a potential source of lead contamination.

In the event the Contractor or its employees or subcontractors fail to comply with all applicable laws, rules and regulations related to lead-containing or lead-based paints and other materials, or fail to comply with any other requirements set forth in this Notice and Certification Regarding Lead-Free Materials, the Contractor shall be held solely responsible for any and all costs associated with any investigative and/or corrective actions deemed necessary by the District, and shall indemnify, defend and hold harmless the

District, under the indemnification provisions of the Contract for the Work, with respect to any and all claims, demands, actions, damages, costs, expenses, and other liabilities arising therefrom.

CERTIFICATION:

The undersigned, under penalty of perjury, hereby certifies that: (i) the undersigned is a duly-authorized representative of the Contractor and, in that capacity, has executed this certification on behalf of the Contractor; (ii) the Contractor has received notice that lead-containing or lead-based paint or other materials may be located on the Project Site; (iii) the Contractor has received notice that it must comply with all applicable laws, rules and regulations governing work with, and disposal of, such materials; and (iv) the Contractor shall comply with all such applicable laws, rules and regulations, as well as the other requirements of this Notice and Certification Regarding Lead-Free Materials.

Representative Name:	
Representative Title:	
Representative Signature:	
Date Signed:	

CERTIFICATION REGARDING IRAN CONTRACTING ACT

(Public Contract Code Section 2200 et seq.)

Pro	ject: _		ar Union School District						
		_	ed hereby certifies to the District, under penalty of perjury under the laws of the State of the following is true and correct:						
(i)		am a duly-authorized representative of the Contractor and, in that capacity, I have executed this ertification on behalf of the Contractor.							
(ii)) The appropriate box is checked immediately below (check only one box), and the statement relating to the Contractor's status in regard to the Iran Contracting Act of 2010 (Public Contract Code Section 2200 et seq.) following such box is true and correct.								
		The	Contractor is not:						
		(i)	identified on the current list of persons and entities engaging in investment activities in Iran prepared by the California Department of General Services in accordance with subdivision (b) of Public Contract Code Section 2203; or						
		(ii)	a financial institution that extends, for 45 days or more, credit in the amount of \$20,000,000 or more to any other person or entity identified on the current list of persons and entities engaging in investment activities in Iran prepared by the California Department of General Services in accordance with subdivision (b) of Public Contract Code Section 2203, if that person or entity uses or will use the credit to provide goods or services in the energy sector in Iran.						
		2010	District has exempted the Contractor from the requirements of the Iran Contracting Act of D after making a public finding that, absent the exemption, the District will be unable to ain the goods and/or services to be provided under the Contract.						
			maximum total amount payable to the Contractor in connection with the Project, as of the of this certification, does not exceed one million dollars (\$1,000,000.00).						
Rep	resent	ative	Name:						
Rep	resent	ative	Title:						
Rep	resent	ative	Signature:						
Dat	e Signe	ed:							
in (civil pe	naltie	rdance with Public Contract Code Section 2205, false certification of this form may result es equal to the greater of \$250,000 or twice the Contract Amount, termination of the r ineligibility to bid on contracts for three years.						

CERTIFICATION REGARDING CONTRACTOR REGISTRATION

Project	t: Del Mar Union School District t: ctor:						
	dersigned hereby certifies to the District, under penalty of perjury under the laws of the State of nia, that the following is true and correct:						
(i)	I am a duly-authorized representative of the Contractor and, in that capacity, I have executed this certification on behalf of the Contractor.						
(ii)	The Contractor is aware and acknowledges that, except as authorized by Business and Professions Code Section 7029.1 and Public Contract Code Section 20103.5, no contractor may bid on a public works project unless the contractor is, and no subcontractor may be listed in any bid for a public works project unless the subcontractor is, currently registered with the DIR and qualified to perform public work under Labor Code Section 1725.5.						
(iii)	The Contractor is aware and acknowledges that no contractor or subcontractor may be awarded a contract for work on a public works project, or may perform any work on a public works project, unless the contractor or subcontractor is currently registered with the DIR and qualified to perform public work under Labor Code Section 1725.5.						
(iv)	The Contractor is aware and acknowledges that, notwithstanding anything to the contrary, if at any time during the performance of the Work, the Contractor or any of its Subcontractors is not duly registered under Labor Code Section 1725.5 (including, without limitation, if the registration expires or the DIR revokes the registration), the District may cancel the Contract and/or replace the Contractor or Subcontractor with a contractor or subcontractor that is duly registered under Labor Code Section 1725.5, and the Contractor and/or its surety shall be responsible for any and all associated costs incurred by the District.						
(v)	The Contractor and each Subcontractor who will perform any of the Work are duly registered with the DIR under Labor Code Section 1725.5.						
(vi)	Evidence (in the form described in the note below) that the Contractor and each Subcontractor are duly registered with the DIR under Labor Code Section 1725.5 is attached to this certification.						
Representative Name:							
Representative Title:							
Representative Signature:							
Date Si	gned:						
Note:	This certification must be accompanied by print-outs of the applicable screens on the DIR website evidencing that the Contractor and all Subcontractors are currently registered under Labor Code Section 1725.5.						
	B-16						

EXHIBIT C REQUIRED PROJECT FORMS

- 1. Certification Regarding Employee Background Checks
- 2. Daily Time and Materials Report
- 3. Progress Payment Request
- 4. Conditional Waiver and Release (Progress Payment)
- 5. Unconditional Waiver and Release (Progress Payment)
- 6. Conditional Waiver and Release (Final Payment)
- 7. Unconditional Waiver and Release (Final Payment)
- 8. Certification Regarding Readiness for Inspection
- 9. Contractor Guarantee
- 10. Certification Regarding DVBE Participation
- 11. Short Interval Schedule

CERTIFICATION REGARDING EMPLOYEE BACKGROUND CHECKS

Proje	t: Del Mar Union School District t: actor:						
	ndersigned hereby certifies to the District, under penalty of perjury under the laws of the State on the following is true and correct:						
(i)	The undersigned is a duly-authorized representative of the Contractor and, in that capacity, has executed this certification on behalf of the Contractor;						
(ii)	The Contractor has fully complied with the requirements of Section 8.2 of the General Provisions for employee background checks, including, without limitation, with respect to employees of any subcontractors that the Contractor intends to use on the Project;						
(iii)	All of the employees of the Contractor and its subcontractors who will enter in or upon the Project Site, or be in the vicinity of the Project Site, in connection with the Project are identified on the Attachment(s) to this certification;						
(iv)	None of the employees of the Contractor or its subcontractors who are identified on the Attachment(s) to this certification have been convicted of a violent or serious felony as defined in Subdivision (c) of Education Code Section 45125.2; and						
(v)	Except for the employees identified on the Attachment(s) to this certification (and except for the employees identified on attachment(s) to other certifications using this form that the Contractor has submitted to the District in connection with the Project), the Contractor shall not suffer of permit any employees of the Contractor or any of its subcontractors to enter in or upon the Project Site, or to be in the vicinity of the Project Site, in connection with the Project.						
Repre	sentative Name:						
Repre	sentative Title:						
Repre	sentative Signature:						
Date	igned:						

ATTACHMENT TO CERTIFICATION REGARDING EMPLOYEE BACKGROUND CHECKS

Instructions

- (1) For each employee, insert all required information (as specified in the table below) in one row of the table.
- (2) In the "Driver License/Identification" column in the table below: (i) specify the number of the employee's driver's license or, if the employee does not have a driver's license, the number of the employee's state-issued identification; and (ii) specify the state that issued the driver's license or identification.
- (3) If identifying more than 15 employees: (i) use copies of this Attachment to identify the additional employees; and (ii) on each such copy, specify the page number and total number of pages where indicated at the bottom of this Attachment.

	Employee Name and Position	Sex	Date of Birth	Height	Weight		Driver License/Identification
1							
2							
3							
4							
5							
6							
7							
8							
9							
10							
11							
12							
13							
14							
15							

	<u> </u>	 			
10					
10 11 12			 		
13					
14					
15					
Page of _					

DAILY TIME AND MATERIALS REPORT

rojec	t:				
epor	: No.:		Date Work Performed:	Date of Repo	rt:
ABOR	•				
No.		Tim	e Description		Cost
				TOTAL	
				1017/12	
QUIP	MENT:				
No.	Hourly Rate	Tim	e Description		Cost
	•	•		TOTAL	
/ATFI	RIALS:				
		nit Cost	Description		Cost
				TOTAL	

CONTRACTOR CERTIFICATION: The undersigned hereby ACKNOWLEDGEMENT BY INSPECTOR OF certifies to the District, under penalty of perjury under the RECORD: Based on personal observation, laws of the State of California, that: (i) the undersigned is a the undersigned believes that the duly-authorized representative of the Contractor and, in descriptions of the labor, equipment and that capacity, has executed this Report on behalf of the materials employed in completing the Contractor; (ii) the descriptions of the labor, equipment Work covered by this Report are true, and materials employed in completing the Work covered accurate, and complete. This by this Report are true, accurate, and complete. acknowledgement does not constitute approval on behalf of the District. Representative Name: Inspector Name: Representative Title: Inspector Signature: Representative Signature: Date Signed: Date Signed:

PROGRESS PAYMENT REQUEST

	trict: Del Mar Union School District ject:		
	ntractor:		
Pay	yment Request No: Request Date:	For period to:	
Dat	te Request Received (to be completed by Architec	t):	
	NTRACTOR'S REQUEST FOR PROGRESS PAYMENT: nnection with the Project, as shown below and on a		
1.	Original GMP		\$
2.	Net approved and executed Change Orders (from	n Co	ontinuation Sheet(s)) \$
3.	GMP as adjusted to date (line 1 + line 2)		\$
4.	Total Work completed to date (including this Pro	gres	ss Payment Request) \$
5.	Retention from total Work completed to date (p	ontract)\$	
6.	Total Work completed less total Retention (line	ine 5)\$	
7.	Total prior Progress Payment Requests paid	\$	
8.	Payment due for this Progress Payment Request	(line	e 6 – line 7) \$
9.	Balance to finish, including Retention (line 3 – lin	ne 4))\$
hee per the ree cases are portaged are particular parti	ONTRACTOR CERTIFICATION: The undersigned creby certifies to the District, under penalty of cerjury under the laws of the State of California, nat: (i) the undersigned is a duly-authorized expresentative of the Contractor and, in that expacity, has executed this Progress Payment equest on behalf of the Contractor; (ii) the cortion of the Work covered by this Progress express express that been completed in strict excordance with the Contract Documents; (iii) the mount specified on Line 8, above, is now due and expable; and (iv) the Contractor has paid all subcontractors and others due any payment extributable to Work covered by prior Progress express expre		ARCHITECT CERTIFICATION: The undersigned hereby certifies to the District, under penalty of perjury under the laws of the State of California, that: (i) the undersigned is a duly-authorized representative of the Architect and, in that capacity, has executed this Request for Progress Payment on behalf of the Architect; (ii) the Architect has determined based on site-observations and the data substantiating this Progress Payment Request that the Work has progressed as indicated herein, the quality of the Work is in accordance with the Contract Documents, and the Contractor is hereby entitled to payment of the amount specified on Line 8, above.
N	ame:		Name:

Title:	Title:
Signature:	Signature:
Date Signed:	Date Signed:
Inspector of Record Acknowledgement of Review:	<u>District Acknowledgement of Review</u> :
Name:	Name:
Title:	Title:
Signature:	Signature:
Date Signed:	Date Signed:

PROGRESS PAYMENT REQUEST – CONTINUATION SHEET

District: Del Mar Union School Distri	ct	
Project:		
Contractor:		
Payment Request No:	Request Date:	For period to:

Α	В	С	D	Е	F	G	Н	1
	Description of Work	Schedule	Work Cor	npleted	Total	%	Balance	Retenti
Ite		d Value	All prior	This	Complete	(F/C	to Finish	on
m			Application	Period	To Date)	(C-F)	
No.			S		(D+E)			
1.								
2.								
3.								
4.								
5.								
6.								
7.								
8.								
9.								
10.								
11.								
12.								
13.								
14.								
15.								
16.								
17.								
	Sub-Total							
	Character Carlos March							
	Change Order Work							
	Change Order No. 1							
	Change Order No. 2							
	Change Order No. 3							
	Change Order No. 4							
	Change Order No. 5							
	Change Order No. 3							
						 		
					1	<u> </u>		

Total To Date				

Change Order Summary	Additions	Deductions	Net Change to Date
Total of all prior Change Orders			
Change Orders approved this month			
Total Additive/Deductive Changes to Date			

Continuation	Choot	oŧ
Continuation	Sneet	OT

CONDITIONAL WAIVER AND RELEASE (PROGRESS PAYMENT)

(Civil Code Section 8132)

NOTICE: THIS DOCUMENT WAIVES THE CLAIMANT'S LIEN, STOP PAYMENT NOTICE, AND PAYMENT BOND RIGHTS EFFECTIVE ON RECEIPT OF PAYMENT. A PERSON SHOULD NOT RELY ON THIS DOCUMENT UNLESS SATISFIED THAT THE CLAIMANT HAS RECEIVED PAYMENT.

IDENTIFYING INFORMATION Name of Claimant:
Name of Customer:
Job Location:
Owner:
Through Date:
CONDITIONAL WAIVER AND RELEASE This document waives and releases lien, stop payment notice, and payment bond rights the claimant has for labor and service provided, and equipment and material delivered, to the customer on this job through the Through Date of this document. Rights based upon labor or service provided, or equipment or material delivered, under a written change order that has been fully executed by the parties prior to the date that this document is signed by the claimant, are waived and released by this document, unless listed as an Exception below. This document is effective only on the claimant's receipt of payment from the financial institution on which the following check is drawn:
Maker of Check:
Amount of Check: \$
Check Payable to:
EXCEPTIONS This document does not affect any of the following: (1) Retentions. (2) Extras for which the claimant has not received payment. (3) The following progress payments for which the claimant has previously given a conditional waiver and release but has not received payment: Date(s) of waiver and release:
Amount(s) of unpaid progress payment(s): \$
SIGNATURE Claimant's Signature:
Claimant's Title:
Date of Signature:

UNCONDITIONAL WAIVER AND RELEASE (PROGRESS PAYMENT)

(Civil Code Section 8134)

NOTICE: THIS DOCUMENT WAIVES AND RELEASES LIEN, STOP PAYMENT NOTICE, AND PAYMENT BOND RIGHTS UNCONDITIONALLY AND STATES THAT YOU HAVE BEEN PAID FOR GIVING UP THOSE RIGHTS. THIS DOCUMENT IS ENFORCEABLE AGAINST YOU IF YOU SIGN IT, EVEN IF YOU HAVE NOT BEEN PAID. IF YOU HAVE NOT BEEN PAID, USE A CONDITIONAL WAIVER AND RELEASE FORM.

IDENTIFYING INFORMATION
Name of Claimant:
Name of Customer:
Job Location:
Owner:
Through Date:
UNCONDITIONAL WAIVER AND RELEASE This document waives and releases lien, stop payment notice, and payment bond rights the claimant has for labor and service provided, and equipment and material delivered, to the customer on this job through the Through Date of this document. Rights based upon labor or service provided, or equipment or material delivered, under a written change order that has been fully executed by the parties prior to the date that this document is signed by the claimant, are waived and released by this document, unless listed as an Exception below. The claimant has received the following progress payment \$ EXCEPTIONS This document does not affect any of the following: (1) Retentions. (2) Extras for which the claimant has not received payment. (3) Contract rights, including (A) a right based on rescission, abandonment, or breach of contract, and (B) the right to recover compensation for work not compensated by the payment.
SIGNATURE Claim and A Signature
Claimant's Signature:
Claimant's Title:
Date of Signature:

CONDITIONAL WAIVER AND RELEASE (FINAL PAYMENT)

(Civil Code Section 8136)

NOTICE: THIS DOCUMENT WAIVES THE CLAIMANT'S LIEN, STOP PAYMENT NOTICE, AND PAYMENT BOND RIGHTS EFFECTIVE ON RECEIPT OF PAYMENT. A PERSON SHOULD NOT RELY ON THIS DOCUMENT UNLESS SATISFIED THAT THE CLAIMANT HAS RECEIVED PAYMENT.

IDENTIFYING INFORMATION
Name of Claimant:
Name of Customer:
Job Location:
Owner:
CONDITIONAL WAIVER AND RELEASE This document waives and releases lien, stop payment notice, and payment bond rights the claimant has for labor and service provided, and equipment and material delivered, to the customer on this job. Rights based upon labor or service provided, or equipment or material delivered, under a written change order that has been fully executed by the parties prior to the date that this document is signed by the claimant, are waived and released by this document, unless listed as an Exception below. This document is effective only on the claimant's receipt of payment from the financial institution on which the following check is drawn:
Maker of Check:
Amount of Check: \$
Check Payable to:
EXCEPTIONS This document does not affect any of the following: Disputed claims for extras in the amount of: \$
SIGNATURE Claimant's Signature:
Claimant's Title:
Date of Signature:

UNCONDITIONAL WAIVER AND RELEASE (FINAL PAYMENT)

(Civil Code Section 8138)

NOTICE: THIS DOCUMENT WAIVES AND RELEASES LIEN, STOP PAYMENT NOTICE, AND PAYMENT BOND RIGHTS UNCONDITIONALLY AND STATES THAT YOU HAVE BEEN PAID FOR GIVING UP THOSE RIGHTS. THIS DOCUMENT IS ENFORCEABLE AGAINST YOU IF YOU SIGN IT, EVEN IF YOU HAVE NOT BEEN PAID. IF YOU HAVE NOT BEEN PAID, USE A CONDITIONAL WAIVER AND RELEASE FORM.

IDENTIFYING INFORMATION Name of Claimant:
Name of Customer:
Job Location:
Owner:
UNCONDITIONAL WAIVER AND RELEASE This document waives and releases lien, stop payment notice, and payment bond rights the claimant has for all labor and service provided, and equipment and material delivered, to the customer on this job. Rights based upon labor or service provided, or equipment or material delivered, under a written change order that has been fully executed by the parties prior to the date that this document is signed by the claimant, are waived and released by this document, unless listed as an Exception below. This claimant has been paid in full.
EXCEPTIONS This document does not affect any of the following: Disputed claims for extras in the amount of: \$
SIGNATURE Claimant's Signature:
Claimant's Title:
Date of Signature:

CERTIFICATION REGARDING READINESS FOR INSPECTION

District: Del Mar Union School District	
Project:	
Contractor:	
Inspection requested for (check only one of the following):	
~ All of the Work ~ Portion of the Work, described as:	
(Nets District consend as wind in advance for inspection of only a postion of the West N	_
(Note: District approval required, in advance, for inspection of only a portion of the Work.)	

The undersigned hereby certifies to the District, under penalty of perjury under the laws of the State of California, that all of the following is true and correct:

- (i) The undersigned is a duly-authorized representative of the Contractor and, in that capacity, has executed this certification on behalf of the Contractor;
- (ii) All inspection and testing of the Work has been completed as required by the Contract Documents and applicable law, including, without limitation, in the presence of, as applicable, the District, the Architect, the Inspector of Record, and other consultants, inspectors and government agency representatives;
- (iii) The Contractor has determined, in accordance with Section 16.1 of the General Provisions, that the Work has been satisfactorily performed in accordance with the Contract Documents (including, without limitation, that all equipment and systems installed in connection with the Work are fully operational), and the Work is now ready for inspection for purposes of determining whether the Work has been substantially completed;
- (iv) If this certification is submitted in connection with a re-inspection of any Work, the previously incomplete and/or unsatisfactory portions of the Work have been completed and/or corrected in accordance with the Contract Documents; and
- (v) The Contractor acknowledges that, as provided in Section 16.7 of the General Provisions, the Contractor shall be solely responsible and liable for any and all costs incurred by the District if the Contractor requests an inspection of any portion of the Work in any of the following circumstances:
 - (1) It is reasonably apparent that such portion of the Work is not complete and ready for inspection;
 - (2) In the reasonable opinion of the District, Architect, construction manager or Inspector of Record, the Contractor is using the inspection as a means to define or determine the scope of such Work or the scope of the uncompleted portions of any Work;

C-14

- (3) In the reasonable opinion of the District, Architect, construction manager and/or Inspector of Record, the Contractor is using the inspection as a means to accelerate the Work of any Subcontractor;
- (4) The Work fails to pass the inspection due to any negligence or misconduct of the Contractor or any Subcontractor or other person or entity on, at or in the vicinity of the Project Site on account of the Work;
- (5) Work noted as incomplete and/or unsatisfactory during a prior inspection is not reasonably complete and/or satisfactory upon re-inspection; or
- (6) The Work is not ready for inspection and it is reasonably apparent that, for any other reason, the Contractor did not request the inspection in good faith.

Representative Name:	
Representative Title:	
Representative Signature:	
Date Signed:	

CONTRACTOR GUARANTEE

Project	t: Del Mar Union School District t: ctor:
The un	dersigned, under penalty of perjury under the laws of the State of California , hereby certifies that:
(i)	The undersigned is a duly-authorized representative of the Contractor and, in that capacity, has executed this certification on behalf of the Contractor;
(ii)	The Contractor hereby acknowledges its obligations under the Contractor Guarantee (defined in Section 18.2 of the General Provisions), including, without limitation, the obligation, at its cost, to repair, replace or otherwise correct any Defective Work (defined in Section 18.1 of the General Provisions) that is discovered prior to the expiration of any applicable Guarantee Period (defined in Section 18.4 of the General Provisions);
(iii)	The Contractor hereby acknowledges that, except as the Contract Documents may otherwise provide:
	(1) The Guarantee Period applicable to the Work generally shall commence on the Substantial Completion Date (defined in Section 17.1 of the General Provisions) and end on the date that is one year after the Project Acceptance Date or the date that is two years after the Substantial Completion Date, whichever is sooner;
	(2) The Guarantee Period applicable specifically to heating, ventilation and air conditioning equipment, controls, et cetera, shall commence when such portion of the Work first begins and shall end on the date that is two years after the Project Acceptance Date or the date that is three years after the Substantial Completion Date, whichever is sooner; and
	(3) The Guarantee Period applicable specifically to roofing materials, membranes, sheet-metal, et cetera, shall commence when such portion of the Work first begins and shall end on the date that is five years after the Substantial Completion Date;
(iv)	The Contractor hereby acknowledges that any provision of the Contract Documents that specifically requires a longer Guarantee Period shall govern over the foregoing clause (iii), and in no event shall any applicable Guarantee Period serve as a limitation with respect to latent defects in the Work, which remain subject to applicable statute(s) of limitation and/or repose.
(v)	The Contractor hereby acknowledges that, in no event shall the Contractor Guarantee be deemed or construed to limit, in any manner, any manufacturer or other third-party guarantee or warranty. At all times while the Contractor Guarantee is in effect during an applicable Guarantee Period, the Contractor must assist the District in processing any manufacturer and other third-party guarantee or warranty claims with respect to systems, equipment, materials and/or other things incorporated into the Project as part of the Work.
	C 16

- (vi) The Contractor, at no cost to the District, must perform Guarantee Work (defined in Section 18.7 of the General Provisions) within ten days after receipt of written notice from the District or, if the Guarantee Work reasonably cannot be completed within ten days of notice from the District, within such reasonable time as determined by the District. The Contractor must coordinate all Guarantee Work with the District in order to avoid interfering with District operations and/or endangering anyone. The Contractor must provide written notice to the District upon completing any Guarantee Work.
- (vii) The Contractor hereby acknowledges that the District, at the Contractor's cost, may cause any required Guarantee Work to be performed if: (1) the Contractor fails to undertake and/or complete the Guarantee Work within the time permitted under the foregoing clause (vi); or (2) if the District reasonably determines that an emergency situation exists. In no event shall the District so causing any Guarantee Work to be performed be deemed or construed to limit or otherwise condition the responsibilities and/or liabilities of the Contractor under the Contractor Guarantee.
- (viii) The Contractor hereby acknowledges that this Contractor Guarantee form is a summary, and not a complete restatement, of the provisions of Part 18 of the General Provisions. In the event of a conflict between any provision of Part 18 of the General Provisions and the provisions of this Contractor Guarantee form, the General Provisions shall govern.

CERTIFICATION REGARDING DVBE PARTICIPATION

	:: Del Mar Unic									
Contra	ctor:									
	The undersigned hereby certifies to the District, under penalty of perjury under the laws of the State of California, that all of the following is true and correct:									
(i)	The undersign executed this		•		•	tative of the Co tractor;	ntractor	and,	in that cap	acity, has
(ii)						on the attachm ibed on the atta				rtification
(iii)	The dollar	amount	of	the	DVBE	participation	in t	he	Contract	totaled:
							dollars	(\$),
	which repres									
	GMP, includir	ng any and a	ıll Chan	ige Ord	ders.					
Repres	entative Name	_							_ _	
·	entative Signat	.ure: _							_	
Date Si		_					-		_	
	nent sheet. M				-	d by the list of ment sheet, as				_

CERTIFICATION REGARDING DVBE PARTICIPATION -- ATTACHMENT SHEET

District: Del Mar Union School District				
	oject:			
Contractor:				
At	tachment Sheet of			
1.	DVBE	Entity		Name:
	State DVBE Certification No.:			 No.:
	Business			Address:
	Participation in Contract: \$			
	Type	of		Participation:
2.	DVBE	Entity		Name:
	State DVBE Certification No.:			 No.:
	Business			Address:
	Participation in Contract: \$			
	Type	of		Participation:
3.	DVBE	Entity		Name:
				State
	DVBE Certification No.:		Telephone No.:	
	Business			Address:
	Participation in Contract: \$			
		C-19		

	Туре	ot		Participation:
4.	DVBE	Entity		Name:
	State DVBE Certification No.:		Telephone	No.:
	Business			Address:
	Participation in Contract: \$			
	Туре	of		Participation:

Note: Types of DVBE participation may include, but are not limited to: (i) construction; (ii) architectural and/or engineering; (iii) suppliers of materials, equipment and/or supplies; and (iv) information technology.

	<u> </u>	age
PART 1	ACKNOWLEDGMENT, ORGANIZATION, AND INTERPRETATION OF GENERAL PROVISIONS	D-1
1.1	Contractor Acknowledgment of General Provisions	D-1
1.2	Organization of General Provisions	D-1
1.3	Guides for Interpreting General Provisions	D-1
1.4	Locations of Definitions of Capitalized Terms	D-1
PART 2	DISTRICT ADMINISTRATION OF THE CONTRACT	D E
2.1	Status of Authorized District Representatives	
2.1	Services of Architect	
2.2	Services of Inspector of Record	
	•	
2.4	Authority to Reject Non-Conforming Work	
2.5	Authority to Stop or Suspend Work	
2.6	Contractor Must Cooperate and Provide Information	
2.7	Contractor Responsibility for Additional Professional Services	
2.8	Damages Incurred by District Under Other Contracts	D-6
PART 3		
3.1	Public Works Project	D-7
3.2	Status of Contractor	D-7
	3.2.1 Independent Contractor	D-7
	3.2.2 Contractor Investments in Iran	D-7
3.3	Contractor Solely Responsible for Work	D-7
3.4	On-Site Job Superintendent	D-8
	3.4.1 No Work if Not Present	D-8
	3.4.2 Responsibilities and Authority	D-8
3.5	Selecting and Contracting with Subcontractors	D-8
	3.5.1 Written Subcontracts Required	D-8
	3.5.2 Selection of Subcontractors	D-9
	3.5.2.1 Informal Selection Process	D-9
	3.5.2.2 Balance of Cost and Other Factors	D-9
	3.5.2.3 Contracting with Selected Subcontractors	D-9
	3.5.2.4 Replacement of Subcontractors E)-10
3.6	Subcontractors Have No Contractual Privity with District	
3.7	Prohibition Against Unlawful Discrimination	
3.8	Contractor Responsible for Subcontractor Compliance)-11
3.9	Contractor Must Maintain Reference Materials at Project Site E)-11
3.10	As-Built Drawings and Specifications)-11
3.11	Updates to As-Built Drawings and Specifications)-12
3.12	Contractor Must Maintain Records of the Work	
3.13	Review and Audit of Records of the Work)-12
PART 4	THE CONTRACT AND CONTRACT DOCUMENTS)-13
4.1	Construing the Contract Documents	

		<u>Page</u>
4.2	Work to be Inferred from Contract Documents	D-13
4.3	Geotechnical and Soils Reports	D-13
4.4	Contractor Must Know and Understand Applicable Legal Requirements	D-13
4.5	Pre-Construction Review of Contract Documents	D-14
4.6	Requests for Information	D-14
	4.6.1 Submittal By Contractor	D-14
	4.6.2 Response By Architect	D-14
4.7	Costs of Erroneous or Non-Conforming Work	D-15
4.8	Ownership and Rights to Contract Documents	D-15
PART 5		
5.1	General Requirements for Submittals	
5.2	Contractor Must Identify Deviations from Contract Requirements	D-16
5.3	Review and Approval of Submittals	D-16
5.4	No Work or Deliveries Permitted Absent Approved Submittals	D-16
PART 6	PERFORMANCE AND PAYMENT BONDING REQUIREMENTS	D-17
6.1	Surety Bonds a Condition Precedent to Commencing the Work	
6.2	Penal Sums of Surety Bonds	D-17
6.3	Surety Qualifications	D-17
PART 7	PERFORMANCE OF THE WORK GENERALLY	D-18
7.1	Contractor Awareness of Existing Site Conditions	
7.2	Contractor Must Furnish Everything Required	
7.3	Contractor Responsible for Permits and Fees	
7.4	Contractor Must Comply with Applicable Legal Requirements	
7.5	Coordination of the Work	D-19
7.6	Impacts of Work by Others	
7.7	Contractor Must Preserve Survey Monuments and Markers	
7.8	NPDES Permit and SWPPP	
	7.8.1 District to Obtain Coverage	
	7.8.2 Implementation and Compliance	
	7.8.3 Consequences of Failure to Comply	
7.9	Contractor Must Comply with Run-Off Control Requirements	
7.10	Contractor Must Not Disturb Run-Off Control Measures	D-20
7.11	Contractor Must Protect Site and Existing Improvements	
7.12	Requirements for Cuts, Patches and Attachments	D-21
7.13	Testing and Inspection	
7.14	Special or Additional Testing or Inspection	
7.15	Contractor Must File Verified Reports	D-22
7.16	Contractor Must File Daily Reports	
7.17	Contractor Must Control Noise and Dust	
7.18	Contractor Must Keep Project Site Clean	D-22

		Page
PART 8	PROJECT SITE DECORUM	D-24
8.1	Performance of Work at Operating School Facilities	
8.2	Procedures to Prevent Contact with Students	
	8.2.1 Significance of Requirements	
	8.2.2 Criminal-History Background Checks	
	8.2.3 Responsibility for Subcontractor Compliance	
	8.2.4 Alternatives to Fingerprinting and Background Checks	
	8.2.5 Consequences of Non-Compliance with Requirements	
8.3	Courteous and Professional Conduct and Appearance	
8.4	Requirements for Use of Restrooms	D-25
8.5	Control of Break-Time Activities	D-25
8.6	Prohibition Against Drugs (including Alcohol) and Tobacco	D-26
8.7	Parking Restrictions	D-26
8.8	Work Vehicles and Equipment	D-26
PART 9	PROJECT SITE SAFETY	D-27
9.1	General Safety-Related Responsibilities	D-27
9.2	Compliance with Safety Requirements	
9.3	Trench Safety Plans	D-27
9.4	Notice and Correction of Non-Compliance	D-28
9.5	Contractor Response to Emergency Situations	D-28
9.6	Accident Reports	D-28
DART 4	O COMPLIANCE WITH LABOR LAWS	D 20
	0 COMPLIANCE WITH LABOR LAWS	
10.1	Contractor Must Comply with Prevailing Wage Laws Copies of Prevailing Wage Rates	
10.2		
10.3 10.4	Penalties for Violations of Prevailing Wage Laws	
10.4	Prohibition Against Debarred Subcontractors	
10.5	Employment of Apprentices	
10.0	10.6.1 Apprentice-Related Requirements	
	10.6.2 Skilled Workforce-Related Requirements	
	10.0.2 Skilled Workforce Related Requirements	D 30
10.7	Limitations on Daily Hours of Work	D-30
10.8	Payroll Records	D-30
10.9	Contractor Must Know and Comply with All Labor Laws	D-31
	1 SCHEDULING OF WORK AND DELAYS	
11.1	Early Work Schedule and Master Construction Schedule	
	11.1.1 Early Work Schedule	
	11.1.2 Initial Submission of Master Construction Schedule	
	11.1.3 Modifications	D-32

		<u>Page</u>
	11.1.4 Updates and Revisions	D-32
	11.1.5 Schedule Compliance and Recovery from Delays	D-32
11.2	Work Outside Regular Working Hours	D-33
	11.2.1 Requested by Contractor	D-33
	11.2.2 Required by Contract or District	D-33
11.3	Delays Resulting from Abnormal Weather	D-33
	11.3.1 Normal Weather Deemed Foreseeable	D-33
	11.3.2 Extensions of Time	
11.4	Delays Arising from Hazardous Materials and Existing Structures	D-34
11.5	Non-Compensable Delays	D-34
11.6	Compensable Delays	
11.7	,	
11.8	Liquidated Damages for Contractor-Caused Delays	D-28
11.9		
11.10	O Requests for Additional Time and/or Compensation for Delays	D-35
11.11	1 Contractor Claims Arising from Delays	D-36
PART 1	12 SUBSTITUTION OF SPECIFIED ITEMS	D-37
12.1	Requests for Substitution of Specified Items	D-37
12.2	Conditions for Requesting Substitution	D-37
	12.2.1 Items That Become Commercially Unavailable	D-37
	12.2.2 Items Still Commercially Available	D-37
12.3	Substantiation of Requests for Substitution	D-37
12.4	Certification of Requests for Substitution	D-38
12.5	Approval of Requests for Substitution	D-38
12.6	Disapproval of Requests for Substitution	D-38
PART 1	13 HAZARDOUS EQUIPMENT, MATERIALS AND SUBSTANCES	D-39
13.1	Use of Hazardous Materials	D-39
13.2	Hazard Communication Program	D-39
13.3	Asbestos and Asbestos-Containing Materials	D-39
	13.3.1 Contractor Awareness of Asbestos	D-39
	13.3.2 Use of Asbestos Prohibited	D-39
	13.3.3 Liability for Asbestos	D-40
13.4	Discovery of Hazardous Materials	D-40
13.5	Discovery of Hazardous Waste in Excavations Deeper Than Four Feet	D-40
13.6		
PART 1	14 EXCAVATIONS AND UTILITIES	D-42
14.1	Underground Utilities and Installations	
14.2	Regional Notification Center	
14.3	<u> </u>	
14.4	Coordination with Utility Companies	

		<u>Page</u>
14.5	Excavation Permits	
14.6	Safety Plans for Trenches Deeper Than Five Feet	
14.7	Differing Conditions in Excavations Deeper Than Four Feet	D-43
	5 CHANGES IN THE WORK	
15.1	Authorization Required for Changes in the Work	D-44
	15.1.1 District Approval of Changes Required	D-44
	15.1.2 DSA Approval of Certain Changes	
	15.1.3 Direction to Proceed With Changes	
	15.1.4 Responsibility for Unauthorized Changes	
15.2	Changes Required by Change Order	
15.3	Changes Required by Architect Field Directive	
15.4	Changes Required by Unilateral Change Order	
15.5	Notice of Disagreement Regarding GMP or Time	
15.6	Consequences of Failure to Provide Notice	
15.7	Changes Requested by the District	
15.8	Changes Requested by the Contractor	
15.9	Determining Effect of Change on GMP	
	Determining Change Order Cost Based on Time and Materials	
	Determining Change Order Cost Based on Lump Sum Proposal	
	Determining Change Order Cost Based on Unit Pricing	
	Cost Components to be Included in All Estimates	
	Deductive or Reduced Change Order Costs	
	Discounts and Refunds Deducted from Change Order Costs	
	Substantiation of Subcontractor Pricing Included in Estimates	
15.17	Substantiation of Time and Materials and Unit-Price Costs	
	15.17.1 Requirement For Notice	
	15.17.2 Requirements for Daily Time and Materials Reports	
	15.17.3 Requirements for Separate Accounting Records	
	Change Orders Include Full and Final Compensation	
15.19	Correction of Non-Conforming Work	D-51
PART 1	6 FINAL INSPECTION AND COMPLETION OF WORK	
16.1	Contractor Must Determine When Work is Complete	
16.2	Initial Request for Final Inspection	D-52
16.3	Re-Inspection of Work	
16.4	Determination that Work is Substantially Complete	
16.5	Contractor Must Timely Complete Remaining Work	D-53
16.6	Contractor Request for Final Walk-Through	D-53
16.7	Contractor Responsibility for Inspection Costs	D-53
16.8	Acceptance of the Completed Project	D-53
PART 1	7 CLOSE-OUT OF THE WORK	D-54

		<u>Page</u>
17.1	Close-Out Submittals are Prerequisite to Final Payment	
17.2	As-Built Drawings and Specifications	D-54
17.3	Equipment Operations and Maintenance Manuals	D-54
	17.3.1 Required Contents	
	17.3.2 Required Correction and Certification	D-54
17.4	Log of Identification Tags and Labels	D-54
17.5	Keys for Doors, Panels, Cabinets, Et Cetera	D-55
17.6	Tools, Spare Parts, Et Cetera	D-55
17.7	Close-Out Materials	D-55
17.8	Other Contract Document Close-Out Requirements	D-55
17.9	Contractor Guarantee	D-55
PART 1	8 CONTRACTOR GUARANTEE OF WORK	
18.1	No Waiver of District Rights	
18.2	Contractor's General Guarantee of Work	
18.3	Limitations on Contractor Guarantee	
18.4	Applicable Guarantee Periods	
18.5	Specific Guarantee Periods for HVAC and Roofing	
18.6	Manufacturer and Other Third-Party Guarantees	
18.7	Guarantee Work by Contractor	
18.8	District Performance of Guarantee Work	
18.9	Warranty of Title to Work	D-57
PART 1	9 PROGRESS PAYMENTS AND FINAL PAYMENT	D-58
19.1	Schedule of Values	D-58
19.2	Contractor Must Obtain Approval of Schedule of Values	
19.3	Contractor Must Monitor and Update Schedule of Values	D-59
19.4	Materials Not Incorporated into Work	
19.5	Contractor Must Arrange Progress Payment Review Meetings	D-59
19.6	Contractor Submittal of Progress Payment Requests	D-59
	19.6.1 Timing and Content of Progress Payment Request	
	19.6.2 Materials to be Submitted with Progress Payment Request	
	19.6.3 Final Progress Payment Request	
	19.6.4 Copies of Progress Payment Requests and Supporting Materials	D-60
19.7	Requirements for Progress Payment Waivers and Releases	
	19.7.1 Conditional Waivers and Releases for Contractor	
	19.7.2 Unconditional Waivers and Releases for Contractor	D-60
	19.7.3 Unconditional Waivers and Releases for Subcontractors and Others	D-61
	19.7.4 Waiver and Release for Final Payment to Contractor	D-61
	19.7.5 Waivers and Releases for Final Payment to Subcontractors and Others	
19.8	Summary of Public Contract Code Section 20104.50	
19.9	Architect Decisions Regarding Progress Payment Requests	D-62
	19.9.1 Notice of Approval or Disapproval	D-62

			Page
		19.9.2 Certification of Payment Upon Approval	D-62
		19.9.3 Rejection and Resubmittal of Progress Payment Requests	D-62
	19.10	Retention from Construction Progress Payments	D-62
	19.11	Payment of Construction Progress Payments	D-63
	19.12	Final Payment to Contractor of Retention	D-63
	19.13	District Issuance of Joint Checks	D-63
	19.14	District Does Not Waive Rights by Inspecting, Approving or Paying	D-63
	19.15	Stop Notices and Liens	D-64
P	ART 2	0 TERMINATION AND OTHER REMEDIES	D-65
	20.1	Termination of Right to Perform Work for Cause	D-65
		20.1.1 Cause for Termination	D-65
		20.1.2 Notice of Intent to Terminate for Cause	
	20.2	Surety and District Rights to Perform Work After Termination for Cause	D-65
	20.3	District Performance of Work After Termination for Cause	D-66
	20.4	Contractor Compensation After Termination for Cause	D-66
	20.5	Termination of Right to Perform Work for Convenience	D-66
	20.6	Contractor Must Cease Work Upon Termination for Convenience	D-66
	20.7	Documenting Costs After Termination for Convenience	D-67
	20.8	Compensation After Termination for Convenience	D-67
		Termination by the Contractor for Cause	
	20.10	Documenting Costs After Contractor Termination for Cause	D-68
	20.11	Remedies for Default Other Than Termination	D-68
	20.12	Effect on other Projects with District	D-68
P	ART 2	1 INDEMNIFICATION OF DISTRICT AND OTHERS	D-69
	21.1	Indemnification of District and its Representatives	D-69
	21.2	Contractor Defense of District and its Representatives	D-69
	21.3	Indemnification of District Officers, Employees, and Agents	D-69
	21.4	Limitation on Contractor Indemnification Obligations	D-69
	21.5	Contractor Must Ensure Subcontractors Indemnify District Indemnitees	D-70
	21.6	Insurance Not a Limitation	D-70
	21.7	Attorney's Fees	D-70
P	ART 2	2 CLAIMS AND LEGAL PROCEEDINGS	D-71
	22.1	Requirements and Procedures for Filing Claims Are Mandatory	D-71
	22.2	Mandatory Time Limits for Filing of Claims	
	22.3	Content of Claims and Substantiating Materials	
	22.4	Mandatory Certification of Claims Subject to Penalty of Perjury	
	22.5	Prerequisites for Filing Delay Claims	
	22.6	Method of Filing Claims	
	22.7	PCC Claims Procedures	
	22.8	District Requests for Additional Information	D-72

	<u> </u>	age
22.9	Procedures for Initial Review of Claim by Architect)-73
22.10	District Response to Claim Following Initial Architect Review)-73
22.11	Initiation of Mandatory Informal Claim-Resolution Efforts)-73
22.12	Documentation of Compromise)-73
22.13	Architect Ruling if Claim Remains Unresolved)-73
22.14	Conditions Precedent to Initiating Subsequent Actions)-74
22.15	Contractor Must Continue Work While Claims Pending)-74
22.16	Resolving Disputes through Binding Arbitration)-74
22.17	Resolving Disputes in Court of Competent Jurisdiction)-74
22.18	Neither Party's Remedies Are Limited)-75
22.19	Applicable Law and Venue)-75
	3 MISCELLANEOUS PROVISIONS	
23.1	Entire Understanding and Agreement)-76
23.2	All Legal Requirements Deemed Included)-76
23.3	Execution of Documents in Counterparts)-76
23.4	Captions and Headings	
23.5	No Third-Party Beneficiaries of Contract)-76
23.6	Circumscribed Right to Assign Contract)-76
23.7	Waiver of Contract Requirements	
23.8	Requirements of Contract are Severable)-77
	Assignment of Anti-Trust Claims	
23.10	Service of Demands and Other Notices	
	23.10.1 General Requirements)-77
	23.10.2 Methods of Delivery	
	23.10.3 Persons to Whom Notices Must be Sent)-78
	23.10.4 Giving or Service of Notice)-78
	23.10.5 Applicability of Section	
	Public Inquiries and Complaints	
23.12	District Notice of Third-Party Claims)-78

PART 1 ACKNOWLEDGMENT, ORGANIZATION, AND INTERPRETATION OF GENERAL PROVISIONS

- 1.1 Contractor Acknowledgment of General Provisions. These General Provisions are an integral component of the Contract Documents. The Contractor: (i) acknowledges that it had the opportunity to review and seek clarification of any and all provisions of these General Provisions from the District and/or Architect and/or from independent legal counsel, if any; and (ii) acknowledges and agrees that the Contractor shall be deemed and construed to have read and that it fully understands these General Provisions.
- **1.2** Organization of General Provisions. These General Provisions are organized according to general subject matter. However, requirements applicable to any particular item, material, service, scope, et cetera, may be included in multiple provisions of these General Provisions. Therefore, unless expressly stated otherwise, the requirements of any one provision in these General Provisions shall not be deemed or construed to govern or apply exclusively over other provisions in these General Provisions applicable to the same item, material, service, scope, et cetera.
- 1.3 Guides for Interpreting General Provisions. As used in these General Provisions, "shall" and "must" shall be deemed and construed as mandatory, and "may" shall be deemed and construed as permissive. Any pronoun, term or phrase in these General Provisions that is stated in either the masculine or feminine sense (e.g., "materialman"), unless otherwise required by context, shall be interpreted as including both or either of such genders. Any reference in these General Provisions to a number of "days," unless specified otherwise, shall mean consecutive calendar days. Any reference in these General Provisions to a number of hours, days, or other units of time, unless specified otherwise, shall mean, respectively, consecutive hours, days or other units of time. Language in these General Provisions to the effect that the Contractor or other person or entity is to provide or furnish something, or words to the same or similar effect, shall be deemed to mean that the thing must be "provided, complete in place," "furnished and installed," or otherwise incorporated into the Work and/or used as necessary to complete the Work and/or administer the Contract in accordance with the Contract Documents. For the convenience of the reader and in order to shorten the length of these General Provisions, capitalized terms are used in these General Provisions to designate defined terms. Each term or phrase that is not capitalized or defined shall be construed in accordance with the meaning commonly associated with such term or phrase within the State and region within which the Project is located.
- 1.4 Locations of Definitions of Capitalized Terms. For the convenience of the reader, the following table specifies the location in these General Provisions and other Contract Documents of the definitions of certain capitalized terms used in the Contract Documents. A capitalized term may be used in the Contract Documents in advance of the location in the Contract Documents where the term is defined. The following table may not include all such capitalized terms, whether intentionally or inadvertently, and, therefore, should not be considered to be a comprehensive list of all such defined terms.

Defined Term	Location of Definition	Location of Definition in Other
	in General Provisions	Contract Documents
Abnormal Weather	Subsection 11.3.2	
Accident Report	Section 9.6	
Actual Costs	Section 20.8	
Architect		Subsection 15(c), CSA
Architect Field Directive	Section 15.3	
Asbestos Remediation	Subsection 13.3.3	
As-Built Drawings and Specifications	Section 3.10	
Authorized Contractor Representative		Subsection 15(b), CSA
Authorized District Representative		Subsection 15(a), CSA
Cal-OSHA	Section 9.2	
CCD	Subsection 15.1.2	
Certification of Payment	Subsection 19.9.2	
Change Order	Section 15.2	
Change Order Cost	Section 15.7	
Change Order Request	Section 15.8	
Claim	Section 22.1	
Compensable Delay	Section 11.6	
Construction Change Directive	Subsection 15.1.3	
Construction Progress Payment	Section 19.11	
Contract		Subsection 3(a), CSA
Contract Documents	Section 4.1	
Contractor Guarantee	Section 18.2	
Contract Time	500000000000000000000000000000000000000	Subsection 5(b), CSA
County		Recital B, Site Lease
DAS	Subsection 10.6	Neortal By one Lease
Defective Work	Section 18.1	
DIR	000000000000000000000000000000000000000	Subsection 8(a), CSA
DIR Director	Section 10.2	
District Board	000000000000000000000000000000000000000	Recital D, Site Lease
District Indemnitee	Section 21.1	
Drawings		Subsection 3(b), CSA
DSA		Section 2, Site Lease
DVBE		Section 10, CSA
Early Work Schedule	Subsection 11.1.1	
Emergency	Section 9.5	
Final GMP		Subsection 6(c), CSA
General Provisions		Subsection 3(b), CSA
Guaranteed Maximum Price - GMP		Subsection 6(a), CSA
Guarantee Period	Section 18.4	V-11/
Guarantee Work	Section 18.7	
HCP	Section 13.2	
Inspector of Record	Section 2.3	
Job Superintendent	Subsection 3.4.1	
Leaseback Payment	22333311011011111	Section 4, LBA
zeasesaok i ayinene		Jestion ij Lbit

Defined Term	Location of Definition	Location of Definition in Other
	in General Provisions	Contract Documents
Legal Requirement		Subsection 6(d)(1), CSA
LLB Agreements		Recital C, Site Lease
Master Construction Schedule	Subsection 11.1.2	
MSDS	Section 13.2	
Non-Compensable Delay	Section 11.5	
Notice	Subsection 23.10.1	
Notice of Award		Subsection 4(a), CSA
Notice of Completion	Section 16.8	
Notice of Delay	Section 11.9	
Notice of Intent to Terminate for Cause	Subsection 20.1.2	
Notice to Proceed		Section 3, Site Lease
Notice of Termination for Convenience	Section 20.5	
NPDES Permit	Subsection 7.8.1	
OCIP		Section 14, LBA
OPSC		Section 2, Site Lease
OSHA	Section 9.2	
PCC Claims Procedures	Section 22.7	
Payment Bond	Section 6.1	
Performance Bond	Section 6.1	
Preconstruction Work		Exhibit A, CSA
Prevailing Wage Laws	Section 10.1	
Progress Payment Request	Subsection 19.6.1	
Progress Payment Review Meeting	Section 19.5	
Project		Recital B, Site Lease
Project Acceptance Date	Section 16.8	
Project Site		Recital B, Site Lease
Rain Day	Subsection 11.3.1	
Records of the Work	Section 3.12	
Recovery Schedule	Subsection 11.1.5	
Regular Working Hours		Section 12, CSA
Remaining Work	Section 16.4	
Request for Proposal - RFP	Section 15.7	
Required Commencement Date		Section 3, Site Lease
Required Completion Date		Section 3, Site Lease
Retention	Section 19.10	
RFI	Subsection 3.4.2	
SAB		Section 2, Site Lease
Safety Program	Section 9.2	
Schedule of Values	Section 19.1	
Specifications		Subsection 3(b), CSA
Specified Item	Section 12.1	
State		Recital B, Site Lease
Subcontract	Subsection 3.5.1	
Subcontractor	Subsection 3.5.1	

Defined Term	Location of Definition	Location of Definition in Other
	in General Provisions	Contract Documents
Substantial Completion Date	Section 17.1	
Surety Bond	Section 6.1	
SWPPP	Subsection 7.8.1	
Trench Safety Plan	Section 9.3	
Work		Recital D, Site Lease
Work by Others	Section 4.2	

PART 2 DISTRICT ADMINISTRATION OF THE CONTRACT

- 2.1 Status of Authorized District Representatives. The Authorized District Representatives (i.e., those who are authorized to represent the District in connection with the Contract and the Project are identified in Exhibit G to this CSA. The Contractor must not rely on any notice, order or other communication from the District that is not signed, given or directed by either the Authorized District Representative, and even then, the Contractor must be aware of the limitations on authority of any Authorized District Representative as set forth in these General Provisions. The District Board may have delegated to one or more Authorized District Representatives the authority to approve changes in the Work costing up to a specific dollar amount and/or an amount not in excess of the limitations set forth in Public Contract Code Section 20118.4. The Contractor shall be responsible for verifying whether the District Board has delegated any such approval authority, as well as any limitations on such approval authority. EXCEPT AS EXPRESSLY PROVIDED IN THESE GENERAL PROVISIONS WITH RESPECT TO ARCHITECT FIELD DIRECTIVES, NO DISTRICT OFFICER, STAFF MEMBER, CONSULTANT, CONTRACTOR OR OTHER PERSON (REGARDLESS OF WHETHER SUCH PERSON IS AN AUTHORIZED DISTRICT OFFICER) HAS THE ABILITY OR AUTHORITY TO ORDER ANY CHANGE IN THE WORK OR ANY OF THE DRAWINGS, SPECIFICATIONS OR OTHER CONTRACT DOCUMENTS ABSENT ACTION BY THE DISTRICT BOARD TO APPROVE THE CHANGE, WHETHER DIRECTLY OR BY DELEGATION OF AUTHORITY.
- **2.2 Services of Architect.** The Architect will provide services in accordance with the agreement between the District and the Architect, including, without limitation, interpreting and deciding matters related to performance of the Work and requirements of the Contract Documents. The Architect's decisions as to matters within its scope of authority, including, without limitation, matters relating to aesthetic effect, shall be final for the purposes of the Contractor proceeding with the Work.
- **2.3 Services of Inspector of Record.** A DSA-certified construction inspector ("Inspector of Record") will be responsible for continuously observing the Work in accordance with the requirements of Title 24 of the California Code of Regulations. The Contractor must not perform any portion of the Work absent knowledge by the Inspector of Record that the Work is to occur, and the Contractor must not perform any of the Work that is to occur on the Project Site if the Inspector of Record is not present on the Project Site. The Contractor may not perform any of the Work on any federal or State holiday observed by the Inspector of Record, regardless of whether the Contractor or its Subcontractors observe such federal or State holiday. The Contractor shall, on an ongoing basis, keep the Inspector of Record fully informed regarding the progress and manner of the Work and the type and character of materials and equipment incorporated into the Work. The Contractor shall provide written notice to the Inspector of Record not later than forty-eight hours in advance of each special or other inspection required in connection with the Work.
- **2.4 Authority to Reject Non-Conforming Work.** The District, Architect, and Inspector of Record each have the authority to reject Work that does not conform to any requirement of the Contract Documents. If any portion of the Work potentially does not or will not conform to any requirement of the Contract Documents, or if otherwise necessary or advisable to ensure compliance with requirements of the Contract Documents, the District may require additional inspection or testing of Work, whether such Work is then fabricated, installed, or completed. The costs of any such additional inspection or testing shall be paid as provided in Section 7.14 of these General Provisions. In no event shall the District, Architect or Inspector of Record be liable for any failure to identify and/or reject any Work that does not or will not conform to requirements of the Contract Documents, and no such failure shall be deemed or

construed to relieve the Contractor from its obligation to complete the Work in accordance with the Contract Documents.

- 2.5 Authority to Stop or Suspend Work. The District, Architect, and Inspector of Record each have the authority to stop or suspend some or all of the Work if: (i) they determine in their reasonable judgment that conditions are unsuitable for proceeding with the Work, including, without limitation, if harm or damage to persons or property reasonably may result if the Work were to proceed; (ii) the Contractor fails to correct Defective Work as directed; or (iii) the Contractor fails to carry out the Work in a manner that ensures it will be completed in accordance with the Contract Documents. The Contractor shall immediately comply with, and shall cause each Subcontractor to immediately comply with, each order to stop or suspend the Work. The Contractor shall not permit the stopped or suspended portion of the Work to resume until so ordered. In no event shall the District, Architect or Inspector of Record be liable for any failure to identify situations in which the Work arguably should or could have been stopped or suspended, and no such failure shall be deemed or construed to relieve the Contractor from its obligation to complete the Work in accordance with the Contract Documents.
- **2.6** Contractor Must Cooperate and Provide Information. The Contractor must cooperate with the District, Architect and Inspector of Record, and must promptly upon request furnish to the District, Architect and/or Inspector of Record such information, as may be necessary for any of them to fully and adequately perform their duties in connection with the Project.
- **2.7** Contractor Responsibility for Additional Professional Services. If, due to any request, act, omission, failure or default of the Contractor in connection with the Work or the performance under the Contract Documents by the Contractor or any of its officers, employees, agents, Subcontractors or other representatives, it is necessary for the District to provide or obtain professional services (including, without limitation, architectural, engineering, construction management, legal and/or inspection services) in addition to what otherwise would be required in connection with the administration of the Contract, the District shall be entitled to reimbursement from the Contractor for any and all costs of such additional services. The District may deduct such costs from any amounts otherwise due to the Contractor in accordance with the Contract Documents or, if such amounts are insufficient, the Contractor shall pay the difference to the District.
- **2.8 Damages Incurred by District Under Other Contracts.** Notwithstanding anything to the contrary, if the District is required, under any other contract entered into by the District in connection with the Project, to pay any damages (whether liquidated under such other contract or otherwise) and/or costs (whether fixed by a court of competent jurisdiction or otherwise), and the District would not have been responsible for such damages and/or costs but for an act or omission for which the Contractor is responsible, then the District may seek indemnification from the Contractor under Part 21 of these General Provisions and/or may pursue such other remedies as are permitted by law and/or the Contract.

PART 3 CONTRACTOR ADMINISTRATION OF THE CONTRACT

3.1 Public Works Project. The Project is being completed on the basis of a "Lease-Leaseback" construction delivery method as authorized under Education Code Section 17406. Nonetheless, the Project is a "public work" and "public project" within the meaning of various provisions of the Public Contract Code, Labor Code, Civil Code, and other applicable Legal Requirements. Therefore, the performance of the Work is subject to such requirements. The Contract Documents include various provisions relating to public works and public projects as provided by law, and the Contractor must thoroughly review and become familiar with the Contract Documents. However, the Contract Documents do not include comprehensive statements of all Legal Requirements applicable to public works and public projects, and the Contractor shall be deemed and construed to have acknowledged that fact by executing the LLB Agreements. In addition, in executing the LLB Agreements, the Contractor thereby represents and warrants that it is familiar and knowledgeable with respect to all Legal Requirements applicable to public works and public projects generally and to the Work specifically.

3.2 Status of Contractor.

- **3.2.1** Independent Contractor. For all purposes of the Contract, the Contractor shall be deemed and construed to be an independent contractor, not an officer, employee, partner, consultant, agent or other representative of the District. The District shall in no event be responsible or liable for any acts, omissions, liabilities or other obligations of the Contractor or any Subcontractor, materialman, or other person or entity that furnishes any labor, materials, services, goods or other things in connection with the Work.
- 3.2.2 Contractor Investments in Iran. Subject to certain exceptions, the Iran Contracting Act of 2010 (Public Contract Code Section 2200 et seq.) prohibits a party that engages in investment activities in Iran, as described in Public Contract Code Section 2202.5, from entering into any contract of \$1,000,000 or more for goods or services to be provided to a public entity. Concurrently with executing and delivering the LLB Agreements to the District, and as a condition to the LLB Agreements taking effect, and using the "Certification Regarding Iran Contracting Act" form, which is one of the Required Contract Forms, the Contractor must certify to the District that: (i) the Contractor is not identified on any list prepared by the California Department of General Services in accordance with subdivision (b) of Public Contract Code Section 2203; (ii) the Contractor is not a financial institution that, for 45 days or more, extends \$20,000,000 or more in credit to any other person or entity identified on any list prepared by the California Department of General Services in accordance with subdivision (b) of Public Contract Code Section 2203, if that person or entity uses or will use the credit to provide goods or services in the energy sector in Iran; (iii) the District has exempted the Contractor from the prohibition after making a public finding that, absent the exemption, the District will be unable to obtain the goods and/or services to be provided under the Contract; or (iv) the GMP does not exceed \$1,000,000.
- 3.3 Contractor Solely Responsible for Work. Except as expressly provided in the Contract Documents, the Contractor shall be solely responsible and liable for: (i) the construction means, methods, techniques, and procedures employed in connection with the Work; (ii) as applicable, the fabrication, procurement, quality, quantity, shipment, delivery, receipt and installation of any materials, equipment, work or services incorporated into the Work; (iii) safety precautions and/or safety programs required in connection with the Work; (iv) the failure of the Contractor or any Subcontractor to carry out Work in accordance with the Contract Documents; and (v) acts or omissions of the Contractor or any Subcontractor, materialman, or other person or entity that furnishes any labor, materials, services, goods

or other things in connection with the Work. Without limiting the foregoing, the Contractor shall be responsible for all damages to persons or property that occur as a result of its fault or negligence in connection with the performance of the Work, and all Work shall be solely at the Contractor's risk with the exception of damage to the Work in excess of five percent of the GMP caused by any tidal wave or earthquake in excess of 3.5 on the Richter Scale (which exception shall apply only if the damaged portion of the Work had been constructed or otherwise performed in accordance with the Contract Documents). The Contractor shall not be deemed or construed to be relieved, to any extent, from any responsibility for performance of any obligation under the Contract Documents because such obligation is being or will be performed by any Subcontractor.

3.4 On-Site Job Superintendent.

3.4.1 No Work if Not Present. At all times during which any portion of the Work is being performed, the Contractor shall have a superintendent for purposes of the Work ("Job Superintendent") present on the Project Site. No portion of the Work that is to occur on the Project Site may commence or continue to be performed if the Job Superintendent is not present at the Project Site, including, without limitation, at any time the Contractor is seeking approval of a replacement Job Superintendent. Prior to commencing any portion of the Work, the Contractor shall provide to the District, in writing, the name and qualifications of the person who will serve as the Job Superintendent. The District, in its reasonable discretion, may approve or disapprove of any Job Superintendent or may require that the Contractor replace any Job Superintendent. The Contractor also may request that the District permit the Prime Contractor to replace any Job Superintendent. The Contractor shall not replace a Job Superintendent unless: (i) the District requires replacement of a Job Superintendent; (ii) the District consents to a request by the Contractor to replace a Job Superintendent; or (iii) a Job Superintendent leaves or is terminated from his or her employment with the Contractor. If it becomes necessary to replace a Job Superintendent, the Prime Contractor shall provide in writing to the District the name and qualifications of a proposed replacement, who shall be subject to District approval as provided herein.

3.4.2 Responsibilities and Authority. The Job Superintendent shall be the Contractor's representative for all purposes of the Work, and shall oversee, direct, and be responsible for the Contractor's operations on the Project Site and the performance of the Work. The Contractor shall dedicate the Job Superintendent on a full-time and exclusive basis to the Work, and the Job Superintendent shall not provide services in connections with any other work or project. The Job Superintendent shall receive and accept all directives and other communications given to the Contractor by the District, Architect or Inspector of Record, which communications shall be binding on the Contractor as if given directly to the Contractor. During the course of the Work, only Authorized Contractor Representatives shall be permitted, as provided in Section 4.6 of these General Provisions, to submit any requests for information regarding the Work or the Contract Documents (each an "RFI"), regardless of whether the request originated with the Contractor or any Subcontractor, materialman, or other person or entity that furnishes any labor, materials, services, goods or other things in connection with the Work. Each and every act and decision of the Job Superintendent shall be deemed and construed to be an act and decision of the Contractor and shall bind the Contractor. Prior to commencing any portion of the Work, the Contractor must take whatever action(s) as may be necessary to authorize the Job Superintendent to the extent provided in this Subsection.

3.5 Selecting and Contracting with Subcontractors.

3.5.1 Written Subcontracts Required. For purposes of the Contract, each person and entity that will perform, under direct or indirect agreement with the Contractor (i.e., of any tier), any portion of the Work or supply, manufacture or fabricate any materials, equipment, and/or other things to

be specially designed, made and/or worked in connection with the Work is referred to in these General Provisions as a "Subcontractor." The Contractor shall enter into appropriate, written contracts (each a "Subcontract") with each first-tier Subcontractor. In addition to any other requirements therein, each Subcontract must provide that: (i) the District is an intended third-party beneficiary of the Subcontract; and (ii) upon termination of the Contractor's right to perform the Work, the District may, in its sole discretion, assume the Subcontract in order to continue the Work to be performed by such Subcontractor. No Subcontract or provision thereof shall be deemed or construed to constitute a limitation or waiver of any right of the District under the Contract Documents.

3.5.2 Selection of Subcontractors.

3.5.2.1 Informal Selection Process. The selection of Subcontractors shall comply with Education Code section 17406, subdivision (a)(4). Contractor must implement an informal bidding or proposal process for the selection of experienced, knowledgeable, reputable, and responsible Subcontractors for the Project to assist in fostering competition and ensuring reasonable pricing of the Work. Contractor must: (i) provide public notice of availability of work to be subcontracted in accordance with the publication requirements applicable to the competitive bidding process of the school district, including a fixed date and time on which qualifications statements, bids, or proposals will be due; (ii) establish reasonable qualification criteria and standards regarding Subcontractors; and (iii) award the subcontract either on a best value basis or to the lowest responsible bidder. The foregoing process may include pregualification (different from the pregualification requirements of Public Contract Code section 20111.6, which is also applicable to the Project) or short-listing. The foregoing process shall not apply to Subcontractors listed in the original proposal, if any. The Contractor must require that each bidder submits with its bid a noncollusion affidavit in accordance with Public Contract Code Section 7106. The Contractor shall inform all contractors from which it seeks bids or proposals that the District will not be a party to any contracts with the selected Subcontractors. The Contractor shall notify and permit the District to be present for opening of bids and/or proposals, to review the bids and/or proposals upon opening and at any reasonable time thereafter, and to make the bids and/or proposals available for public review. The Contractor shall review each bid and/or proposal and ensure that it fully encompasses the scope of work intended for the applicable portion of the Work.

3.5.2.2 Balance of Cost and Other Factors. The Parties intend that the process for selection of Subcontractors under this Subsection 3.5.2 shall take into consideration the cost of the Work and other factors important to timely and efficient completion of the Work. Therefore, if the Contractor proposes to use other than the contractor with the monetarily-lowest bid or proposal for any trade or other component of the Work, the Contractor shall document in reasonably-detailed writing to the District all reasons for using other than the contractor with the monetarily-lowest bid or proposal and, in conjunction therewith, shall provide to the District any documentation in support of such reasons. The Parties may agree to seek additional or new bids or proposals for any trade or other component of the Work if necessary or advisable in the best interests of the Project and the District, including, without limitation, if any selected contractor fails or refuses to enter into a Subcontract.

3.5.2.3 Contracting with Selected Subcontractors. Upon the Parties agreeing on the final selection of Subcontractors under this Subsection 3.5.2, the Contractor must: (i) provide to the District an updated list of all Subcontractors providing services in connection with the Project; (ii) complete, execute and submit a copy of the "Certification Regarding Contractor Registration" included in the Required Contract Forms that is applicable to all Subcontractors not covered by any such form previously submitted by the Contractor in connection with the Project; and (iii) to the extent Public Contract Code Section 20111.6 applies, ensure that the Subcontractors are prequalified by the District

prior to commencing any of the Work. Each Subcontractor that will perform any portion of the Work, to the extent required by any applicable Legal Requirement, must be duly and appropriately licensed by the Contractors State License Board prior to commencing such portion of the Work. Within five days of entering into any Subcontract, the Contractor must provide to the District a copy of the Subcontract, including, without limitation, any and all exhibits and/or attachments thereto.

3.5.2.4 Replacement of Subcontractors. All Subcontractors awarded construction subcontracts, no matter when they were selected, shall be afforded all the protections of the Subletting and Subcontracting Fair Practices Act (Chapter 4 (commencing with Section 4100) of Part 1 of Division 2 of the Public Contract Code). Upon the occurrence of one of the situations listed in Public Contract Code section 4107, the District shall permit the Contractor to replace any Subcontractor as provided in this Subsection. If the Contractor proposes to replace any Subcontractor, the Contractor must set forth in a reasonably-detailed written request to the District all reasons for replacing the Subcontractor and, in conjunction therewith, shall provide to the District any documentation in support of such reasons. The District shall not unreasonably deny, withhold, or condition its consent to replace any Subcontractor if such replace complies with the law. Except if the District in its discretion consents in writing, or except in the event of an Emergency that reasonably makes it impractical or infeasible, the Contractor shall select the replacement Subcontractor using the procedures set forth in the foregoing provisions of this Subsection 3.5.2.

- 3.6 Subcontractors Have No Contractual Privity with District. Nothing in the Contract Documents or any Subcontract shall be deemed or construed to create any contractual relationship between the District and any Subcontractor, materialman, or other person or entity that furnishes any labor, materials, services, goods or other things in connection with the Work. The Contractor is and shall remain fully responsible and liable to the District for all acts and omissions of any and all Subcontractors, materialmen, or other individuals or entities that furnish any labor, materials, services, goods or other things on behalf of the Contractor and in connection with the Work, and their respective contractors, employees, agents and other representatives.
- **3.7 Prohibition Against Unlawful Discrimination.** In connection with the LLB Agreements and performance of the Work, including, without limitation, in regard to their employment practices, the Contractor and each Subcontractor of every tier must comply with, and must not discriminate or provide preferential treatment in violation of, any and all applicable District policies or regulations, federal, State and other anti-discrimination laws, rules, regulations and requirements, as amended from time to time, including, but not limited to:
- (i) The Fair Employment and Housing Act (Cal. Gov. Code Section 12900 et seq.);
- (ii) The Unruh Civil Rights Act (Civil Code Section 51 et seq.);
- (iii) California Government Code Section 11135 et seq.;
- (iv) California Labor Code Section 1101 et seq.;
- (v) California Labor Code Section 1735;
- (vi) The Federal Civil Rights Act of 1964 (42 U.S.C. Section 2000e et seq.);

- (vii) The Americans With Disabilities Act of 1990 (42 U.S.C. Section 12101 et seq.);
- (viii) The Age Discrimination in Employment Act (29 U.S.C. Section 621 et seq.);
- (ix) The Rehabilitation Act of 1973 (29 U.S.C. Section 701 et seq.); and
- (x) Presidential Executive Order 11246.
- 2.8 Contractor Responsible for Subcontractor Compliance. Each Subcontractor and other person or entity on, at or in the vicinity of the Project Site on account of the Work must comply with all applicable provisions of these General Provisions (including, without limitation, standards of behavior), notwithstanding that: (i) various obligations set forth in these General Provisions are characterized as the Contractor's obligations; or (ii) the Contractor is expressly responsible in accordance with various provisions of these General Provisions for including certain obligations in its agreements with Subcontractors and others. The Contractor must ensure that each Subcontractor is aware of and understands: (i) the respective authority of the District, Architect, and Inspector of Record under the Contract Documents; and (ii) all general requirements set forth in these General Provisions and other Contract Documents that are applicable to the Subcontractor. In no event shall the failure of any such Subcontractor to have any such awareness and/or understanding be deemed or construed to constitute a basis or excusable cause for any extension of the Contract Time or increase in the GMP. The Contractor shall be responsible and liable for any and all costs arising from any failure by the Contractor to comply with the foregoing obligations.
- **3.9 Contractor Must Maintain Reference Materials at Project Site.** The Contractor shall maintain in good order at the Project Site: (i) the As-Built Drawings and Specifications, and copies of all addenda, Bulletins, Interpretations, Clarifications, Change Orders, Architect Field Directives, Construction Change Directives, and other documents that modify, illustrate or explain the Contract; and (ii) at least one copy each of all approved shop drawings, product data sheets, samples, and similar submittals of the Contractor. The Contractor must continuously update or otherwise maintain current versions of such reference materials. Upon request, the Contractor must make any or all such reference materials available to the District, Architect and/or Inspector of Record.
- 3.10 As-Built Drawings and Specifications. During the course of performing the Work, the Contractor must, in accordance with the Architect's instructions, carefully and accurately illustrate, locate, dimension, note or otherwise describe on one full-size set of the Drawings and Specifications ("As-Built Drawings and Specifications") any and all deviations, corrections, deletions, additions, enhancements, expansions and/or clarifications of, from, or to the Work initially prescribed or shown in the Drawings and Specifications. Not as a limitation on the foregoing, the As-Built Drawings and Specifications must illustrate and/or specify any and all: (i) Work performed or completed differently than as initially shown or required; (ii) changes ordered under Architect Field Directives, Construction Change Directives and/or Change Orders; (iii) authorized substitutions of Specified Items; (iv) final location of all electrical and mechanical equipment, utility lines, ducts, outlets, structural members, walls, partitions, and other significant elements of the Work; and (v) all existing improvements, including, without limitation, any existing substructures, encountered during the performance of the Work that were not demolished or otherwise removed or relocated. No portion of the Work shall be permanently sealed or covered until all required information relating to such portion of the Work has been recorded for purposes of preparing the As-Built Drawings and Specifications.

- 3.11 Updates to As-Built Drawings and Specifications. At each Progress Payment Review Meeting, the Contractor must submit, for approval, three copies of each sheet and/or page of the As-Built Drawings and Specifications showing all changes made since the Contractor submitted the immediately preceding Progress Payment Request. Each update of the As-Built Drawings and Specifications is subject to approval by the District, Architect and the Inspector of Record, and the Contractor must promptly revise the updated As-Built Drawings and Specifications if required. The approval of the updated As-Built Drawings and Specifications, and certification by the Contractor that the updated As-Built Drawings and Specifications fully and accurately reflect the Work as actually completed and in progress as of the end of the period covered by the associated Progress Payment Request, shall be conditions precedent to payment to the Contractor under such Progress Payment Request.
- **3.12** Contractor Must Maintain Records of the Work. The Contractor shall prepare, update and maintain on file, in its principal office, records of the Work containing all significant documentation related to the Work ("Records of the Work"), including, without limitation, copies of each Contract Document, Change Order, Architect Field Directive, Construction Change Directive, shop drawing, product data sheet, sample, submittal, approval, RFI, invoice, payment receipt, conditional waiver, unconditional waiver, punch-list, et cetera. The Contractor shall organize and maintain the Records of the Work in a logical manner, based on subject matter and/or portion of the Work and chronologically.
- 3.13 Review and Audit of Records of the Work. In accordance with Government Code Section 8546.7, the State has the right to examine, review, audit and/or copy the Records of the Work during the three-year period following final payment to the Contractor under the Contract. In addition, the District, DSA, SAB and OPSC each hereby has the right to examine, review, audit and/or copy the Records of the Work during the four-year period following final payment to the Contractor under the Contract. Therefore, the Contractor shall preserve and retain all such Records of the Work for a period of four years commencing upon final payment to the Contractor under the Contract or, if an examination, review or audit is commenced but not completed within such four-year period, until such examination, review or audit has been completed. The Contractor, upon request, shall make the Records of the Work available for the purposes described in this Section at all reasonable times during the period the Contractor is required to preserve and maintain the Records of the Work.

PART 4 THE CONTRACT AND CONTRACT DOCUMENTS

- that establish the entire understanding and agreement between the District and the Contractor, regardless of when prepared or entered into, including, without limitation, the documents described in Subsection (b) of Section 3 of the Construction Services Agreement. The Contract Documents are intended to be complementary and, as described herein, inclusive of the Work to be completed by the Contractor. The Contract Documents are to be construed collectively as a whole; therefore, any item of Work required by any one or more of the Contract Documents, but not required by others, shall be provided by Contractor as if specifically required by each of the Contract Documents. However, if there is an inconsistency in the requirements of the Contract Documents, then: (i) the Construction Services Agreement shall be deemed and construed to govern over the General Provisions; (ii) the General Provisions shall be deemed and construed to govern over the Drawings and Specifications; (iii) the Specifications shall be deemed and construed to govern over the Drawings as to materials, workmanship, and installation procedures; and (iv) a requirement in the Specifications that is more stringent, requires a higher quality and/or requires a greater quantity shall be deemed and construed to govern over other requirements in the Specifications.
- 4.2 Work to be Inferred from Contract Documents. The Drawings, Specifications, and other Contract Documents may not specifically illustrate or describe every item of Work required to complete the Contractor's scope of work and to deliver to the District a complete Project. The Contract Documents are intended to sufficiently describe the Work required so that the Contractor may determine the materials, labor, services and other things of any nature required for the proper execution and completion of the Work, including, without limitation, any requirements for materials, labor, services or other things that should be inferred from the Contract Documents. Therefore, in addition to completing all Work as expressly illustrated or described in the Contract Documents, the Contractor shall be responsible for providing all materials, labor, services, and other things of any nature necessary to complete the Work as may be inferred from the Contract Documents so that, in conjunction with any work in connection with the Project contracted for by the District or work accomplished using the District's own forces ("Work by Others"), the District receives a fully complete and operational Project as intended.
- **4.3 Geotechnical and Soils Reports.** The District or Architect may determine that, in connection with the Project, it is required or advisable to obtain one or more geotechnical and/or soils reports. In such event, the District, at its expense, may provide copies of such report(s) to the Contractor. However, if the District provides any such report to the Contractor, the report in no event shall be deemed or construed to be part of the Contract Documents, and the District shall be deemed and construed to have provided the reports solely for the convenience of, and as supplementary information to, the Contractor. The District shall not be deemed or construed to have made any representation or warranty as to the information or recommendations set forth in any geotechnical and/or soils reports provided by the District in connection with the Project, or as to the soils, groundwater, or other surface and subsurface conditions of the Project Site. The information included in any such report shall be deemed and construed as approximate only, and the Contractor shall be responsible for inspecting and verifying the actual conditions of the Project Site.
- **4.4** Contractor Must Know and Understand Applicable Legal Requirements. The Contractor is hereby required to, and shall be deemed for all purposes of the Contract to, be aware of, know and understand all Legal Requirements applicable to public works projects generally and to the Work

specifically. Therefore, each and every Legal Requirement that is, or is required to be, set forth or summarized in the Contract Documents that, for any reason, is omitted from, or is incorrectly set forth or summarized in, the Contract Documents, shall nonetheless be deemed to be included in the Contract Documents, and the Contract shall be construed and enforced as though such omitted or incorrect provision were properly and correctly set forth in the Contract Documents. All Legal Requirements that apply to public works projects generally and to the Work specifically, but which are not required to be listed or summarized in the Contract Documents, shall also be incorporated as if fully set forth herein.

4.5 Pre-Construction Review of Contract Documents. Prior to when required to initially proceed with the Work, the Contractor shall complete a careful and detailed review of all Contract Documents in order to: (i) determine and confirm all materials, labor, services, and other things of any nature required to fully complete the Work in accordance with the Contract Documents; (ii) determine whether there are any errors, inconsistencies, conflicts, ambiguities, omissions, or lack of sufficient detail or explanation in the Drawings, Specifications or other Contract Documents; (iii) determine whether the requirements of the Drawings, Specifications and other Contract Documents conform with all Legal Requirements applicable to the Work, including, without limitation, Title 21 and Title 24 of the California Code of Regulations, applicable building codes, and utility-company requirements. The Contractor must provide prompt written notice to the District and the Architect if the Contractor perceives any issues as described in the foregoing clauses (ii) or (iii). Neither the requirements of this Section, nor any delay by the Contractor in complying with such requirements, shall be deemed to relieve the Contractor from complying with requirements for commencing and/or completing the Work, and Contractor shall schedule and complete its pre-construction activities to accommodate the review required under this Section. Aside from its review obligations under this Section 4.5, nothing in the Contract Documents shall be deemed or construed to make the Contractor responsible for any errors or omissions in any portion of the Drawings and Specifications or other design documents for the Project prepared by or under the direction of the Architect.

4.6 Requests for Information.

4.6.1 Submittal By Contractor. To the extent reasonable, the Contractor may submit to one or more RFIs, in writing, as provided in this Section 4.6. The Contractor must submit each RFI to the Inspector of Record and the Architect, with a copy thereof to the District. The Contractor must submit an RFI sufficiently in advance as will avoid and/or prevent any delays in the Work or any Work by Others. Each RFI must: (i) identify the specific portion of the Contract Documents that is the subject of the RFI, including, without limitation, Drawing and detail number, Specification section, page number, et cetera; (ii) describe in reasonable detail what the Contractor does not understand, what the Contractor believes is not sufficiently detailed or provided for, or other matter or question that is the subject of the RFI; and (iii) describe the Contractor's interpretation or suggested resolution of the matter or question that is the subject of the RFI. Prior to submitting an RFI, the Contractor must review the RFI to ensure that it is clear (not ambiguous) and asks a question not previously answered.

4.6.2 Response By Architect. The Architect will respond in writing to each sufficiently-detailed RFI, taking into consideration any comments provided by the Inspector of Record, within fifteen days or such other reasonable time as will be dependent on the number of then-pending RFIs and other submittals, the complexity of the issues raised by any RFIs and/or other submittals, the relative importance or priority of the RFIs and other submittals, *et cetera*. In no event shall the Architect's response to an RFI be deemed or construed to constitute, accept, acknowledge, or acquiesce to any change to, or variance from, the Drawings, Specifications or other Contract Documents unless: (i) the RFI expressly and unambiguously specifies that the Contractor is seeking a change to, or variance from, the

D-14

Construction Services Agreement

requirements of the Contract Documents and the Architect's response to the RFI expressly acknowledges and consents to such change or variance; or (ii) the Architect initiates the change to, or variance from, the Contract Documents and expressly and unambiguously states in its response to the RFI that such response, or any particular portion thereof, constitutes a change to, or variance from, the Contract Documents.

- **4.7 Costs of Erroneous or Non-Conforming Work.** If the Contractor or any Subcontractor performs any portion of the Work that does not comply with the Contract Documents or applicable Legal Requirements, the Contractor shall bear any and all costs arising therefrom including, without limitation, the cost of correction, without increase or adjustment to the GMP or the time for performance.
- 4.8 Ownership and Rights to Contract Documents. For all purposes related to the Contract, the Drawings, Specifications, and other Contract Documents are deemed to be and shall remain the property of the District. Except for the District and, to the extent provided in its agreement with the District, the Architect, no person or entity (including, without limitation, the Contractor and anyone performing any work or services under the Contract on behalf of the Contractor) may own or claim a copyright in the Drawings, Specifications, or other Contract Documents. Neither the Contractor, nor any other person or entity other than the District, may use any of the Contract Documents for any purpose other than in connection with the Work. Upon final completion and acceptance of the Project in accordance with these General Provisions, the Contractor must return to the District all originals and copies of any and all of the Contract Documents, except that the Contractor may retain one complete set of the Contract Documents for the Contractor's records.

PART 5 SUBMITTALS

- Authorized District Representatives and Architect any and all shop drawings, samples, deferred approvals, and other submittals required under the Contract Documents, including, without limitation, any submittals prepared or otherwise provided by or on behalf of any Subcontractor. A Subcontractor may not provide a submittal directly to the Authorized District Representatives and Architect because the Contractor has responsibility for full compliance with the requirements of this Part 5. The Contractor must submit each submittal accompanied by a separate transmittal letter for each submittal item or group of items for which a submittal is required under the Contract Documents. Each submittal, and the transmittal letter and other information submitted with the submittal, must be legible in all respects. All shop drawings, samples, deferred approvals, and other submittals required under the Contract Documents shall be deemed and construed to become property of the District upon submittal under this Part 5. The Contractor may not use the submittal process, in lieu of complying with the procedures for substitution set forth in Part 12 of these General Provisions, to request substitution of any Specified Item.
- **5.2** Contractor Must Identify Deviations from Contract Requirements. If a shop drawing, sample, deferred approval or other submittal required under the Contract Documents does not conform in every respect with all requirements of the Contract Documents (including, without limitation, any qualification of, modification of, or other deviation from such requirements), such deviation must be expressly identified and explained in detail in the submittal, noted in the accompanying transmittal letter, and, as applicable, identified by "clouding" on the submittal.
- and other submittals required under the Contract Documents and, as applicable, request additional information, request correction and resubmittal, and/or approve the submittals. The Contractor must make all corrections requested by the Architect and must resubmit the revised submittal to the Architect within the time specified by the Architect. In no event shall the review and/or approval of any submittal be deemed or construed to: (i) constitute verification or approval of any specific dimensions, field conditions, or quantities; (ii) constitute a comprehensive analysis and verification that the submittal is free of all errors and deficiencies; (iii) relieve the Contractor from responsibility for any deviation in the submittal from the requirements of the Contract Documents that was not expressly stated in the submittal as being a deviation from such requirements; (iv) relieve the Contractor from responsibility for any error or deficiency in the submittal not within the scope of review and approval; (v) relieve the Contractor from responsibility for any deficiency in the Work, any failure to coordinate or ensure that the item(s) fit with other portions of the Work or any Work by Others, or any other failure to perform the Work in accordance with the Contract Documents; or (vi) relieve the Contractor from any responsibility for violation of any patent or other right of any person or entity.
- 5.4 No Work or Deliveries Permitted Absent Approved Submittals. In no event may the Contractor commence any portion of the Work that requires any shop drawing, sample or other submittal under the Contract Documents, unless and until the Architect has reviewed and approved the submittal in accordance with this Part 5. In addition, if DSA approval is required for any submittal, the Contractor may not commence the portion of the Work requiring the submittal unless and until: (i) the Architect has obtained the required DSA approval; or (ii) the Architect directs the Contractor, in writing, to commence such portion of the Work.

D-16

PART 6 PERFORMANCE AND PAYMENT BONDING REQUIREMENTS

- 6.1 Surety Bonds a Condition Precedent to Commencing the Work. Within seven days after receipt of the Notice of Award (if issued by the District) or prior to the Required Commencement Date, whichever is sooner, the Contractor must provide to the District: (i) a material and labor payment bond to ensure satisfaction of any claims of materials suppliers and of mechanics and laborers employed in connection with the Work ("Payment Bond"); and (ii) a bond to ensure faithful (including, without limitation, timely) performance by the Contractor of its obligations under the Contract Documents ("Performance Bond"). The Payment Bond and the Performance Bond (each a "Surety Bond") must be in the forms provided by the District as one of the Required Contract Forms or, in the District's sole discretion, may be in such forms as are approved by legal counsel for the District in advance of the Contractor commencing the Work. Neither of the Surety Bonds shall have a stated expiration date, and each shall remain in effect at all times that Contractor has any obligation under the Contract.
- **6.2 Penal Sums of Surety Bonds.** Each of the Surety Bonds initially must have a penal sum equal to 100% of the GMP. If the GMP is increased in accordance with the Contract Documents, then, within seven days after such increase, the Contractor must increase the amount of each of the Surety Bonds to equal the total increased GMP. In addition, the Contractor shall review and renew or amend either or both of the Surety Bonds within seven days after receiving notice from the District that either or both have become insufficient.
- and admitted to transact business in the State in accordance with Code of Civil Procedure Section 995.120. Each of the Surety Bonds must be signed by the duly-authorized representatives of both the Contractor and the surety, and the signatures must be notarized. In addition, for each of the Surety Bonds, the Contractor must provide to the District: (i) a print-out of information from the website of the Department of Insurance confirming that the surety is an admitted surety insurer; or (ii) a certificate from the Clerk of the County that the surety is an admitted surety insurer. Such print-out or certificate must be attached to the associated Surety Bond. Should any surety lose its status as a State-admitted surety, the Contractor shall immediately provide written notice thereof to the District, and the District shall make no further payments to the Contractor under the Contract Documents until such time as the surety regains its status or the Contractor obtains and the District qualifies and approves a substitute surety.

PART 7 PERFORMANCE OF THE WORK GENERALLY

- Parties entering into the LLB Agreements, the Contractor acknowledges that it has had opportunities to inspect the Project Site and become fully acquainted with the conditions under which the Work will be performed and to fully understand the facilities, difficulties, restrictions, and requirements attendant to the performance of the Work on and at the Project Site. By entering into the LLB Agreements, the Contractor shall be deemed and construed to have so inspected the Project Site and to know and understand the conditions of the Project Site existing as of the Effective Date. Therefore, the Contractor shall not be relieved of any obligation under the Contract as a result of any failure by the Contractor: (i) to know or understand all such existing conditions; (ii) to have discovered or learned of any such existing condition; or (iii) to have correctly ascertained the existence, extent, scope, magnitude, or limitations of any such existing condition. Except as expressly provided in the Contract Documents, the Contractor shall not be entitled to any additional compensation in connection with the Project as a result of any condition of, at, or in the immediate vicinity of the Project Site that differs from what the Contractor may have anticipated.
- 7.2 Contractor Must Furnish Everything Required. Except to the extent the Contract Documents expressly provide otherwise, the Contractor must obtain, furnish, or otherwise make available or have in effect, all at its own cost, any and all labor and other services, materials (including, without limitation, any that will not be incorporated into the Work), systems and building equipment, tools, construction equipment and machinery, heat, air conditioning, water, electricity, other utilities, transportation, temporary and permanent facilities, permits and licenses, layout, surveys and other things as are necessary or convenient for the Contractor to undertake and properly complete the Work in accordance with the Contract Documents, including, without limitation, all assemblies and systems specified in the Contract Documents and anything that is not specified in, but that reasonably may be inferred from, the Contract Documents as described in Section 4.2 of these General Provisions. The Contractor shall be solely responsible and liable for paying all applicable federal, State, and local taxes and other charges assessed or levied on or in connection with such labor, materials, and other services and things. For purposes of the Contract, the Work includes each and every obligation described or otherwise specified in the Contract Documents that must be undertaken, performed and completed by the Contractor in connection with the Project, including, without limitation: (i) any Work inferred as provided in Section 4.2 of these General Provisions; and (ii) any Work performed on behalf of the Contractor by any Subcontractor or other person or entity.
- 7.3 Contractor Responsible for Permits and Fees. Except for initial DSA approval of the Drawings and Specifications, if any, the Contractor shall not assume that the District or some other Party has obtained any and all permits, licenses, approvals, and inspections, or paid any and all fees and other costs, necessary for the Contractor to perform the Work. The Contractor must ascertain for itself whether any permits, licenses, approvals or inspections must be obtained, or any fees or other costs paid, in order for the Contractor to properly and legally perform the Work. Unless the Contract Documents expressly provide otherwise, the Contractor must complete the application processes (including, as required, obtaining the signature of an Authorized District Representative on any such application) and obtain all permits, licenses and similar authorizations that are necessary for or in connection with performance of the Work, including, without limitation, any construction, encroachment, or other permits necessary for off-site improvements and utility facilities. The Contractor shall be responsible for ensuring that it and each Subcontractor has applied for, obtained, and maintains in effect at all times during the performance

D-18

of the Work, any and all permits and licenses as are required by applicable Legal Requirements to be in effect in connection with performance of the Work. As between the District and the Contractor, the Contractor shall be solely responsible for paying the costs of all such permits, licenses, and authorizations, including, without limitation: (i) fees for any business, contractor's, or other license or permit required for the Contractor to conduct its business; (ii) any filing and plan-check fees for deferred approvals or other approvals that the Contractor must obtain in accordance with the Contract Documents; and (iii) any royalties, rental, and/or license fees arising from the use of any material, machine, method or process used in performing the Work.

- 7.4 Contractor Must Comply with Applicable Legal Requirements. At all times during and in connection with performance of the Work, the Contractor must fully comply with, give any and all notices required by, and undertake any and all actions required under, all applicable Legal Requirements, including, without limitation, applicable building, mechanical, plumbing, fire and other codes. The Contractor shall be solely responsible and liable for any failure to comply with such requirements in connection with the Work, and shall bear the cost of additional work arising from any such failure, including without limitation, any costs arising from any stop-work order issued by the DSA or other governmental entity as a result of the Contractor failing to perform the Work in compliance with applicable Legal Requirements.
- 7.5 Coordination of the Work. The Contractor must: (i) appropriately and adequately coordinate and time the various portions of the Work to ensure that all Work is properly completed within the time(s) required under the Master Construction Schedule; and (ii) determine when completed or in-progress portions of the Work are sufficient and ready for subsequent portions of the Work. The Contractor must coordinate, time, and perform all portions of the Work so as to avoid and/or prevent any interference with any Work by Others. The Contractor must at all times furnish a sufficient number of workers to ensure that the Work is efficiently and timely undertaken and completed in accordance with the Master Construction Schedule. The Contractor shall permit the Work to be performed only by workers who are appropriately qualified and skilled in the work assigned to them. Notwithstanding the foregoing, apprentices may perform portions of the Work if enrolled in an appropriate apprenticeship program and appropriately supervised at all times during performance of such portions of the Work.
- contracts to provide for Work by Others. The District, in its sole discretion, may: (i) enter into other contracts to provide for Work by Others in connection with the Project, or may perform any such Work by Others using its own forces and/or (ii) provide for phasing of various portions of the Project. The Contractor must ascertain for itself the overall scope and nature of the Project, the scope and nature of any Work by Others required to complete the Project and the potential impacts of any Work by Others on the coordination, timing, scheduling and performance of the Work, including, without limitation, the impacts of any phasing of the Project. The Contractor must accommodate all such potential impacts, and shall be deemed and construed to have accommodated all such potential impacts, when: (i) preparing the proposed Master Construction Schedule; (ii) coordinating, timing, and scheduling the Work; and (iii) performing the Work. If the Contractor believes that any delay in any Work by Others has caused or resulted in any unreasonable delay and/or unreasonably increases the cost of performing the Work, the Contractor may provide notice and a request for Change Order as provided in Sections 11.9 and 11.10, respectively, of these General Provisions.
- **7.7** Contractor Must Preserve Survey Monuments and Markers. At all times prior to completion of the Work, the Contractor must employ reasonable caution to avoid removing, dislocating, covering or otherwise disturbing any survey monuments, stakes, markers, devices, or implements. In the event any

such monument, stake, marker, device, or implement is disturbed by any person on, at or in the vicinity of the Project Site on account of the Work, regardless of whether due to negligence, accident or other cause, the Contractor shall be solely responsible and liable for all costs attributable to such disturbance, including, without limitation, costs of replacement and/or other correction by a licensed land surveyor or registered civil engineer and any necessary compliance with requirements to file records thereof with appropriate governmental authorities.

7.8 NPDES Permit and SWPPP.

7.8.1 District to Obtain Coverage. The District shall be responsible for obtaining any required coverage for the Project from the State Water Resources Control Board under the National Pollutant Discharge Elimination System General Permit for Storm Water Discharges Associated with Construction and Land Disturbance Activities ("NPDES Permit"), including, without limitation, filing the required "Permit Registration Documents," which includes a Notice of Intent, Storm Water Pollution Prevention Plan ("SWPPP"), and other compliance related documents. The District may contract with a consultant for the preparation and processing of some or all of such documents, including, without limitation, the SWPPP.

7.8.2 Implementation and Compliance. If NPDES Permit requirements are applicable to the Project, the Contractor must not commence work on the Project until the District has provided a copy of the NPDES Permit and the SWPPP to the Contractor, and it shall be the Contractor's responsibility to incorporate such requirements into all Subcontracts. The Contractor shall be responsible for implementing and complying with the provisions of the NPDES Permit and the SWPPP, including, without limitation, the standard provisions, monitoring and reporting requirements as required by the NPDES Permit. The Contractor shall provide copies of all reports and monitoring information to the District and Architect. The GMP shall be deemed and construed to include compensation to the Contractor for all costs of compliance with specified requirements of the NPDES Permit and SWPPP. The Contractor shall be responsible for removal and clean-up of all run-off and other control measures upon completion of the Work.

- **7.8.3** Consequences of Failure to Comply. Failure to comply with the NPDES Permit is in violation of federal and State law. The Contractor shall be solely responsible and liable for any and all costs and/or delays arising from any failure of the Contractor to comply with any of the requirements described in this Section.
- performance of the Work, the Contractor must comply with all federals, State, and local governmental and quasi-governmental requirements providing for control of erosion, run-off, drainage, and/or discharges into any storm drain systems and/or watercourses including, without limitation, all applicable requirements of any city, County and/or flood-control district, and any NPDES Permit and SWPPP applicable to the Work and/or the Project. The Contractor must ascertain for itself the requirements applicable to the Work and the Project. The Contractor shall be solely responsible and liable for any and all costs and/or delays arising from any failure of the Contractor to comply with any of the requirements described in this Section.
- **7.10** Contractor Must Not Disturb Run-Off Control Measures. Except for removal upon completion of all other Work, the Contractor must not remove, relocate, damage, destroy or otherwise disturb any on-site or off-site erosion and/or run-off control measures established in connection with the

Project. The Contractor must replace and/or restore any such control measures disturbed by any person or entity on, at, or in the vicinity of the Project Site on account of the Work. The Contractor shall be solely responsible and liable for any and all costs and/or delays arising from any failure of the Contractor to comply with any of the requirements described in this Section.

7.11 Contractor Must Protect Site and Existing Improvements. At all times prior to completion of the Work, the Contractor must adequately secure, preserve and protect all: (i) completed and inprogress portions of the Work; (ii) materials and equipment used in connection with the Work; (iii) all Work by Others and other existing improvements in the immediate vicinity of, affected by, or receiving any portion(s) of, the Work; and (iv) the portion(s) of the Project Site in the vicinity of the Work. The foregoing shall be deemed to require, among other things, that the Contractor: (i) install barricades around any shrubs or trees that the Contract Documents or the District requires to be preserved or protected, but that may be adversely affected by performance of the Work; (ii) drain, remove or otherwise mitigate any water, mud, dust, debris, et cetera, as necessary for proper performance of the Work; (iii) avoid driving over, parking on, and placing any loads (including, without limitation, stabilization legs of cranes or other equipment) on, any curbs, gutters and sidewalks on, at or in the vicinity of the Project Site; and (iv) avoid overloading any completed or partially completed building, structural members or elements, or other improvements, with workers, materials and/or equipment. The Contractor must promptly repair, replace or otherwise correct any damage to property caused by the Contractor or any other person or entity performing any of the Work or, alternatively, the District may cause any such damaged property to be repaired, replaced or otherwise corrected by another party at the Contractor's expense. The Contractor shall be solely responsible and liable for any costs and/or delays arising from any failure of the Contractor to provide adequate security, preservation and protection as provided herein.

7.12 Requirements for Cuts, Patches and Attachments. The Contractor must obtain the written consent of the Architect prior to: (i) cutting and/or patching any completed portions of the Work, any Work by Others, or existing improvements on, at or in the vicinity of the Project Site; and/or (ii) cutting, boring or drilling into, or attaching anything onto, any structural members or elements, including, without limitation, columns, shear walls, trusses, et cetera. The Contractor must resolve any doubt regarding whether something is a structural member or element by consulting with the Architect. The Architect may impose reasonable conditions on any such approval, including, without limitation, conditions relating to the appearance, quality, functionality, integrity and/or safety of the Work and any Work by Others. The Contractor shall be solely responsible and liable for any and all costs arising from any action within the scope of this Section that is performed without consent of the Architect, performed contrary to direction or instructions of the Architect, or is otherwise defective or improperly performed, including, without limitation, any costs arising from the need to obtain Architect, DSA and/or other approvals.

7.13 Testing and Inspection. The Contractor must provide written notice to the District, Architect and Inspector of Record sufficiently in advance of when it is necessary for any required testing or inspection to occur, in order to: (i) permit the District to arrange for such testing or inspection, including, without limitation, if materials and/or equipment are required to be inspected at the place of manufacture or other source of supply; and (ii) avoid and/or prevent any delays in performing the Work and any Work by Others. The District, Architect, Inspector of Record and/or entity performing any test shall select each sample of material that must be tested, not the Contractor. If the Contractor incorporates into the Work any materials and/or equipment that are required to be tested and/or inspected prior to such materials and/or equipment having been tested and approved, or closes or otherwise makes inaccessible any assembly or assemblies that must be tested prior to being closed or otherwise becoming inaccessible, the District may require that the Contractor open, uncover or otherwise deconstruct that

portion of the Work in order to permit the required inspection(s) to occur. In such event, the Contractor shall be solely responsible and liable for any costs arising from the need to open, uncover, or deconstruct such portion of the Work and to thereafter replace or otherwise correct such portion of the Work.

- 7.14 Special or Additional Testing or Inspection. If necessary or advisable, the District may require special or additional testing and/or inspection, not otherwise required, of any portion of the Work and/or any materials and/or equipment incorporated or to be incorporated therein. To the extent required, the Contractor must open, uncover, or deconstruct any portions of the Work or any assemblies incorporated therein in order to reveal hidden portions of the Work. If the special or additional testing and/or inspection reveals that the Work conforms to all requirements of the Contract Documents, the District shall be responsible for the cost of the testing and/or inspection and for costs incurred by the Contractor, as required, to open, uncover, or deconstruct any portions of the Work and, thereafter, correct such Work and any Work by Others disturbed by the additional testing and/or inspection. If the special or additional testing and/or inspection reveals that the Work does not substantially conform to all requirements of the Contract Documents, the Contractor shall be solely responsible for the cost of the testing and/or inspection and for costs incurred by the Contractor, as required, to open, uncover or deconstruct any portions of the Work and, thereafter, correct such Work and any Work by Others disturbed by the additional testing and/or inspection.
- **7.15** Contractor Must File Verified Reports. From time to time during the performance of the Work and as otherwise required by the DSA, the Contractor and each other person or entity performing any portion of the Work that is required by any applicable Legal Requirement, must prepare and file with the DSA any and all verified reports required under Education Code Section 17309, Section 36 of Title 21 of the California Code of Regulations, and Section 4-366 of Title 24 of the California Code of Regulations.
- 7.16 Contractor Must File Daily Reports. For each day on which any portion of the Work is performed, regardless of the scope or amount of such Work, the Contractor must provide to the District, Architect, and Inspector of Record a report describing all activities in connection with performance of the Work that day, listing all workers performing any Work that day and their respective trades, experience levels and employers, and any equipment or materials delivered and/or installed or incorporated into the Work that day. The above-described report shall include a daily sign-in sheet for Contractor's employees and the employees of any Subcontractor and/or supplier who are on the Project Site for any given work day. For each day on which the Contractor conducts any safety meeting, the Contractor must submit minutes and the sign-in or attendance sheet for the safety meeting with the daily report for that day.
- 7.17 Contractor Must Control Noise and Dust. At all times during performance of the Work, the Contractor must ensure that all construction equipment used in connection with the Work are properly fitted with appropriate and adequate noise-reduction devices or mechanisms, and that all such devices and mechanisms are properly and well maintained. The Contractor also must take all such actions as are necessary to prevent dust and/or debris attributable to the Work from blowing into, or otherwise creating a nuisance or polluting, any other areas on or off the Project Site. If the Work is to be performed at any time school is in session, the risk of the District stopping the Work on account of any such disturbance shall be deemed and construed as foreseeable, and the Contractor in no such event shall be entitled to an extension of the Contract Time or an increase in the GMP.
- **7.18** Contractor Must Keep Project Site Clean. The Contractor must at all times keep the Project Site free of (and not less than daily must collect and remove) any and all trash, debris, dust, excess water, excess materials, unused equipment, et cetera attributable to the Work, and shall maintain the Project

Site and all structures or other improvements in a clean and orderly condition. The Contractor must comply with any and all reasonable directions of the District, Architect and Inspector of Record with respect to clean-up of the areas in which the Work is or was being performed, and any other areas affected by such Work (including, without limitation, adjacent streets, sidewalks, gutters, et cetera), break areas, lunch areas, et cetera. If the Contractor fails at any time to keep such areas clean and free of trash, debris, et cetera, the District may cause such areas to be cleaned by others, and the cost thereof shall be charged to the Contractor and/or deducted from amounts otherwise payable to the Contractor under the Contract.

PART 8 PROJECT SITE DECORUM

Performance of Work at Operating School Facilities. This Section 8.1 shall be applicable to the Contract if the Work or any portion thereof is to be performed at any existing and operating school facility. The Contractor and each Subcontractor and other person or entity on, at or in the vicinity of the Project Site on account of the Work must: (i) become informed of and take into account the age and maturity of the students on the Project Site; (ii) control their behavior accordingly; (iii) avoid or, if avoidance is not possible, minimize interactions with any students who are not accompanied by an adult; (iv) coordinate, schedule and perform any Work that may cause inconvenience, interference or other disturbance of any classes or other school operations during times that will eliminate or minimize the disturbance; (v) enclose the area in which the Work is to occur with a substantial barricade or take other safety precautions as directed by the District; and (vi) comply with directions from the District regarding the timing and performance of the Work that are intended to avoid unnecessarily disturbing school operations. Each person at, on, or in the vicinity of the Project Site on account of the Work must remain in the immediate vicinity of the portion of the Work he or she is to perform, and must not stray to other areas of the Project Site; provided that the foregoing shall not be construed to prohibit any such person from accessing, as reasonably necessary, his or her work area(s), restrooms, any designated lunch area, or any designated parking area.

8.2 Procedures to Prevent Contact with Students.

8.2.1 Significance of Requirements. This Section 8.2 shall be applicable to the Contract only if so provided in Section 11 of the Construction Services Agreement. If this Section 8.2 is applicable, it is because the District has determined that persons assigned to the Work or who otherwise will be present at, on or in the vicinity of the Project Site on account of the Work may have more than "limited contact" with minor-aged students.

8.2.2 Criminal-History Background Checks. Unless the Contractor is a sole proprietor, as discussed in Section 11 of the Construction Services Agreement, the Contractor, in conformance with Education Code Section 45125.1, shall require and be responsible for ensuring that each person who will be at, on or in the vicinity of the Project Site on account of the Work shall comply with all California Department of Justice guidelines and requirements relating to fingerprinting and criminal-history background checks. The Contractor shall certify in writing to the District, using the "Certification Regarding Employee Background Checks" form, which is one of the Required Project Forms, that no person assigned to the Work or who otherwise will be present at or on the Project Site has been convicted of any serious or violent felonies (as described in Education Code Section 45122.1). The Contractor must attach to the executed Certification of Employee Background a list of all persons to whom the certification applies. The Contractor shall prohibit and prevent each and every person who will be at, on or in the vicinity of the Project Site on account of the Work (including not only all persons assigned to the Work directly by the Contractor, but also all persons assigned to the Work by any Subcontractor, materialman, or other person or entity that furnishes any labor, materials, services, goods or other things in connection with the Work) from being present at, on or in the vicinity of the Project Site unless and until the Contractor provides the required certification including such person to the District.

8.2.3 Responsibility for Subcontractor Compliance. The Contractor shall require in each Subcontract that, if the Subcontractor will assign any person to the Work or otherwise will cause or permit any person to be present at or on the Project Site, the Subcontractor must cooperate in regard to, and fully comply with, the requirements of this Section 8.2. The Contractor may on that basis delegate

responsibility for compliance with this Section 8.2 to any such Subcontractor; however, the Contractor at all times retains full responsibility and/or liability for such compliance or lack thereof.

- 8.2.4 Alternatives to Fingerprinting and Background Checks. Upon request of the Contractor with respect to any particular situation and/or duration of time, the District in its sole discretion may consent to the Contractor implementing measures intended to protect the District's minoraged students, which measures would be in lieu of the Contractor complying with Subsections 8.2.2 and 8.2.3 herein. Subject to District approval, such alternative measures might include, but are not necessarily limited to: (i) installing a physical barrier to limit contact between students and the employees and other representatives of the Contractor, Subcontractors, and others present on or at the Project Site on account of the Work; (ii) providing for the continuous supervision and monitoring of such employees, representatives, and others by an employee of the Contractor who has received fingerprint clearance from the California Department of Justice; or (iii) providing for the surveillance of such employees, representatives, and others by a District employee. The Contractor must implement any such approved alternative measures at no additional cost to the District, and the Contractor shall be responsible for ensuring compliance with such alternative measures by or with respect to all persons assigned to the Work or who otherwise will be present at, on, or in the vicinity of the Project Site on account of the Work.
- **8.2.5** Consequences of Non-Compliance with Requirements. Due to the possible adverse consequences of contact with students and other minor-aged individuals, any failure by the Contractor to ensure compliance with the requirements of this Section 8.2 shall be deemed and construed to constitute a material breach of the Contract, upon which the District, in its sole discretion, may immediately terminate the Contract without any further compensation to Contractor and/or pursue all other rights and remedies it may have against the Contractor under law or the Contract.
- **8.3** Courteous and Professional Conduct and Appearance. The District may require that the Contractor remove and either temporarily or permanently bar from the Project Site any person on or in the vicinity of the Project Site on account of the Work who: (i) does not act in a courteous and professional manner; (ii) does not wear clothing appropriate for the Work being performed and the situation in which the Work is being performed; (iii) wears any clothing or other thing with any obscene, violent, derogatory, or other inappropriate depiction or message, or that depicts or advertises any drug, alcohol, tobacco, or other controlled substance prohibited to minors.
- **8.4 Requirements for Use of Restrooms.** If the Contract Documents do not require that the District make restroom facilities available at the Project Site, the Contractor shall be responsible for providing portable restroom facilities (including, without limitation, hand-washing area and soap) for use by all persons at, on, or in the vicinity of the Project Site on account of the Work. In no circumstances may any such person urinate, defecate or expectorate (i.e., spit) anywhere at, on or in the vicinity of the Project Site other than, as appropriate, into a urinal or toilet inside a restroom facility.
- **8.5** Control of Break-Time Activities. The Contractor must control the break-time activities of all persons present on the Project Site on account of the Work. The Contractor shall ensure that break areas are cleaned after each break and that all refuse (including, without limitation, soda cans, food wrappers and containers, plastic bottles, and uneaten food) are placed in appropriate trash or recyclable-materials receptacles. No person may bring onto, or keep in the vicinity of, the Project Site any glass bottle or other glass food or beverage container.

- **8.6 Prohibition Against Drugs (including Alcohol) and Tobacco.** All District properties, including, without limitation, the Project Site, are "drug-free" workplaces and, therefore, the Contractor is hereby made subject to the requirements of Government Code Sections 8350 *et seq.*, the Drug-Free Workplace Act of 1990. In addition, all District properties, including, without limitation, the Project Site, are "tobaccofree" workplaces. No person at, on or in the vicinity of the Project Site on account of the Work may: (i) engage in the unlawful manufacture, dispensation, possession or use (including being under the influence) of any illegal or controlled substance; (ii) possess or use any alcoholic beverage; (iii) use any legal substance that results or likely will result in serious or significant impairment of normal abilities; or (iv) smoke, inhale, chew, or otherwise use or consume tobacco products, including via devices known as ecigarettes or "vape pens." Within seven days after receipt of the Notice of Award (if issued by the District) or prior to the Required Commencement Date, whichever is sooner, the Contractor must submit to the District an executed copy of the "Certification Regarding Drug-Free Workplace" and "Certification Regarding Tobacco-Free Workplace" forms included in the Required Contract Forms.
- **8.7 Parking Restrictions.** Each person who brings a personal or work vehicle or equipment to the Project Site must fully comply with any parking controls and/or program established by the District. If the District has not designated an on-site or off-site parking area for use in connection with the Work, then no personal or work vehicles or equipment: (i) may be parked on the Project Site except as expressly authorized by the District; or (ii) may be parked in the vicinity of the Project Site in any manner that is illegal or otherwise creates a safety hazard or nuisance.
- 8.8 Work Vehicles and Equipment. Any work vehicles and/or equipment used on the Project Site must be removed from the Project Site or, if applicable, removed to a designated location, promptly after completion of the portion of the Work for which the vehicle or equipment is required. The Contractor shall be responsible for damage to the Project Site or any necessary repair or other work at the Project Site resulting from use of any such vehicle or equipment. The Contractor and/or the individual owners of any personal or work vehicles or equipment at, on or at the Project Site shall be and remain responsible for the security, safety and condition of such vehicles and equipment. None of the District, Architect or Inspector of Record shall be responsible or liable for any theft or damage that occurs to any vehicles and/or equipment at, on, or in the vicinity of the Project Site.

PART 9 PROJECT SITE SAFETY

- 9.1 General Safety-Related Responsibilities. Without limiting any other provision of the Contract Documents, the Contractor has, and shall at all times retain, the ultimate responsibility for: (i) keeping the areas at, on, and in the vicinity of the Project Site free of safety hazards arising from performance of, or in any way connected with, the Work; (ii) performing the Work in a manner that ensures the safety of persons and property at, on, or in the vicinity of the Project Site; (iii) providing all physical safety measures required in connection with the Work to adequately protect persons and property at, on or in the vicinity of the Project Site, whether required by the Contract Documents, any Legal Requirement, or because dictated by conditions in which the Work will be performed; (iv) complying with all Legal Requirements applicable to job-site and worker safety, including, without limitation, posting of all required information regarding protection of workers and giving of any and all required notices, warnings and disclosures; (v) protecting the Work and all materials, equipment and other things to be incorporated into the Work or used in connection with the Work, whether on or off the Project Site; and (vi) protecting property at, on or in the vicinity of the Project Site that may be affected by the Work, including, without limitation, structures, streets, sidewalks, gutters, paved areas, utilities, trees, shrubs and lawns that will not be removed or replaced in connection with the Project.
- Compliance with Safety Requirements. The Contractor shall prepare in writing any and all safety, loss-prevention and/or injury and illness prevention plans and/or programs required in connection with the Work, whether required by the Contract Documents or by any applicable Legal Requirement, including, but not limited to, Cal-OSHA requirements (each a "Safety Program"). The Contractor shall provide a copy of each Safety Program to the District prior to commencing any portion of the Work. An insurance loss-prevention agent and/or other District representative, including, without limitation, in connection with an applicable OCIP, may implement a Safety Program in connection with the Project, in which case, the Contractor and each Subcontractor shall fully comply with all requirements of such Safety Program. As directed by the District, the Contractor shall coordinate its Safety Program procedures and requirements with those of any other Safety Programs applicable to the Project. The Contractor shall be responsible for implementing and ensuring compliance with any and all procedures, training and other requirements of each applicable Safety Program, including, without limitation, training of workers in regard to security of the Project Site and the transportation, storage and use of hazardous materials. The Contractor shall also be responsible for compliance with any and all applicable Legal Requirements relating to job-site and work safety, including, without limitation, requirements for fully-stocked and adequate first-aid supplies at the Project Site consistent with U.S. Department of Labor, Occupational Safety and Health Administration ("OSHA") and California Department of Industrial Relations, Division of Occupational Safety and Health ("Cal-OSHA") requirements.
- 9.3 Trench Safety Plans. If the GMP exceeds \$25,000, then, prior to undertaking the excavation of any trench that will or reasonably might be five feet or more in depth, the Contractor must submit to the District a detailed plan showing the design of the shoring, bracing, sloping or other mechanisms for protection of workers and others from collapse or cave-in of the trench ("Trench Safety Plan"). Without limiting any other requirement applicable to contents of a Trench Safety Plan, each Trench Safety Plan must also: (i) provide for safe means for workers, as necessary, to enter and exit the trench; (ii) consider active and surcharge loads, and specify minimum required distances between the trench edges and adjacent buildings, embankments, subsurface utilities, construction equipment, spoils, et cetera; and (iii) provide for daily inspection of the trench by a competent person prior to each work shift and after any change in existing conditions. Each Trench Safety Plan must be prepared by an appropriately skilled,

experienced and licensed civil or structural engineer, who must certify that the Trench Safety Plan complies with minimum requirements of all applicable Construction Safety Orders of Cal-OSHA. If a Trench Safety Plan varies from the standards established by applicable Construction Safety Orders, the Contractor must obtain Cal-OSHA approval of the Trench Safety Plan. The Contractor shall not commence the excavation of a trench within the scope of this Section until the Contractor has provided (and the District has accepted) the applicable Trench Safety Plan and a copy of the excavator's current and valid Cal-OSHA Construction Activity Permit. Neither anything in this Section nor any review and/or acceptance by the District of any Trench Safety Plan shall be deemed or construed to: (i) impose any tort liability on the District; or (ii) relieve the Contractor from responsibility for providing shoring, bracing, sloping, or other mechanisms adequate to protect workers and others on, at or in the vicinity of the Project Site in connection with any trench to be excavated by or on behalf of the Contractor.

- 9.4 Notice and Correction of Non-Compliance. Upon being notified of any actual or potential hazard to the health or safety of any person or property at, on or in the vicinity of the Project Site, the Contractor must immediately undertake action to correct, cure or otherwise eliminate the hazard. If the Contractor fails to eliminate, or fails to implement reasonable efforts to eliminate, the actual or potential hazard within a reasonable time, not to exceed twenty-four hours in any case, the District may, but shall not be obligated to, eliminate such hazard, and the cost thereof shall be charged to the Contractor and/or deducted from amounts otherwise payable to the Contractor under the Contract. In no event shall the failure by the District or any other party to provide notice to the Contractor of any actual or potential hazard be deemed or construed to relieve the Contractor from any responsibility or liability whatsoever attributable to such hazard. Unless and only to the extent set forth in the Contract Documents, the District assumes no responsibility or liability for the physical condition or safety of the Project Site, any ongoing or completed construction thereon, or any equipment, supplies or materials present at, on or in the vicinity of the Project Site.
- 9.5 Contractor Response to Emergency Situations. In the event of an Emergency that endangers or reasonably might endanger the health, safety or welfare of any person(s) or property, the Contractor shall take such action as the Contractor, in its reasonable discretion, determines is necessary to prevent the threatened damage or injury to person(s) or property, including, without limitation, rendering first aid to any injured person. For purposes of these General Provisions, an "Emergency" is defined as a sudden, unexpected occurrence that creates a clear and imminent danger and that requires immediate action to prevent or mitigate any injury to any person (including death) or any damage to, or loss of, property or essential public services.
- 9.6 Accident Reports. The Contractor must file a written report with the District as provided in this Section (each an "Accident Report") in each case in which, due to any cause: (i) any significant accident occurs (regardless of whether any injury or damage occurs), any person is injured (including death), or any property on, at or in the vicinity of the Project Site is damaged, in connection with the performance of the Work or by any person at, on or in the vicinity of the Project Site on account of the Work, and regardless of whether such portion of the Work is being or to be performed by the Contractor or any Subcontractor; or (ii) the Contractor renders aid to any person as described in Section 9.5 of these General Provisions. The Contractor shall provide an Accident Report to the District within twenty-four hours of the incident detailed in the Accident Report.

PART 10 COMPLIANCE WITH LABOR LAWS

- 10.1 Contractor Must Comply with Prevailing Wage Laws. The Contractor must be, and shall be deemed and construed to be, aware of and understand the requirements of California Labor Code Sections 1720 et seq. and 1770 et seq., and Title 8 of the California Code of Regulations, Section 16000 et seq. (collectively, "Prevailing Wage Laws"), which require the payment of prevailing wage rates and the performance of other requirements on certain "public works" and "maintenance" projects. The Project is a "public works" project, as defined by the Prevailing Wage Laws, and the Contractor must perform all of the Work as a public works project. The Contractor must fully comply, and must ensure full compliance by all Subcontractors and other persons and entities as required, with all applicable Prevailing Wage Laws.
- **10.2 Copies of Prevailing Wage Rates.** Prior to commencing any portion of the Work, the Contractor must: (i) obtain from the Director of Industrial Relations ("DIR Director") the general prevailing rate of per-diem wages and the general prevailing rate for holiday and overtime work in the locality in which the Work is to be performed for each craft, classification, or type of worker needed to perform the Work, in effect as of the date the District sought proposals for the Work; and (ii) provide copies of such prevailing wage rates to the District. The Contractor also must make copies of the prevailing rates of perdiem wages for each craft, classification or type of worker needed to perform the Work available to interested parties upon request, and must post copies at the Contractor's principal place of business and at the Project Site.
- 10.3 Penalties for Violations of Prevailing Wage Laws. In accordance with Labor Code Section 1775, the Contractor and any Subcontractor shall forfeit, as a penalty to the District, not more than \$200 and, subject to limited exceptions, not less than certain amounts specified by law, for each calendar day, or portion thereof, for each worker paid less than the prevailing wage rates as determined by the DIR Director for the work or craft in which the worker is employed. The Contractor or the applicable Subcontractor shall pay to each worker the difference between such prevailing wage rates and the amount paid to the worker for each calendar day or portion thereof for which the worker was paid less than the applicable prevailing wage rates.
- **10.4** Labor-Compliance-Related Requirements. The Contractor and each Subcontractor must comply with all labor-related requirements, as described in Section 9 of the Construction Services Agreement, at no extra cost to the District.
- 10.5 Prohibition Against Debarred Subcontractors. No Subcontractor may perform any portion of the Work if the Subcontractor is ineligible to perform work on a public works project under Section 1777.1 or Section 1777.7 of the Labor Code. Any contract relating to a public works project entered into by the Contractor and any such "debarred" party is void as a matter of law, and a debarred party may not receive any public money for performing work as a contractor or subcontractor on any public works project. The Contractor must refund to the District any public money that has been paid to a debarred party in connection with the Work. The Contractor shall be responsible for the payment of wages to workers of any debarred party that is allowed to perform any of the Work.

10.6 Employment of Apprentices; Skilled and Trained Workforce.

10.6.1 Apprentice-Related Requirements. The Contractor and each Subcontractor shall be responsible for compliance with the provisions of law relating to employment of apprentices, including, without limitation, Sections 1777.5, 1777.6, and 1777.7 of the California Labor Code. As provided by Labor

Code Section 1777.7, violations of Labor Code Section 1777.5 may result in forfeiture not to exceed \$100 for each full calendar day of non-compliance. Information regarding apprenticeship standards, wage schedules, and other requirements may be obtained from the DIR Director, ex officio the Administrator of Apprenticeship, San Francisco, California, or from the DIR's Division of Apprenticeship Standards ("DAS") and its branch offices.

10.6.2 Skilled Workforce-Related Requirements. To the extent Project is subject to Education Code Section 17407.5, the Contractor shall be solely responsible and liable for ensuring compliance by it and each of its Subcontractors at every tier with the requirements of Section 17407.5, and Public Contract Code Section 2600 et seq., to perform the Work using a "skilled and trained workforce," including, without limitation, as such requirements progressively take effect over time or are amended. Each month during the Site Lease Term, and until the District Board acceptance of the Project under Section 16.8 of these General Provisions, the Contractor must submit a written report to the District that demonstrates compliance with the requirements of Public Contract Code Section 2600 et seq. by the Contractor and its Subcontractors at every tier. Each such report shall be a public record and available for public inspection. IF THE CONTRACTOR FAILS TO SUBMIT ANY MONTHLY REPORT REQUIRED UNDER THIS SECTION OR PUBLIC CONTRACT CODE SECTION 2600 ET SEQ., OR SUBMITS A REPORT THAT DOES NOT DEMONSTRATE COMPLIANCE WITH PUBLIC CONTRACT CODE SECTION 2600 ET SEQ., THE DISTRICT: (i) SHALL IMMEDIATELY CEASE MAKING ANY PAYMENTS THAT OTHERWISE ARE PAYABLE OR BECOME PAYABLE TO THE CONTRACTOR IN ACCORDANCE WITH THE LLB AGREEMENTS AS PROVIDED IN PUBLIC CONTRACT CODE SECTION 2602; AND (ii) WILL NOT RESUME MAKING ANY SUCH PAYMENTS UNLESS AND UNTIL THE CONTRACTOR CURES ITS NONCOMPLIANCE. THE CONTRACTOR, AND EACH SUBCONTRACTOR AT EVERY TIER, SHALL CONTINUE PERFORMING WORK ON THE PROJECT WITH NO WORK STOPPAGES. THE DISTRICT SHALL NOT BE LIABLE WHATSOEVER FOR THE CONTRACTOR'S FAILURE. OR ANY SUBCONTRACTOR'S FAILURE. TO COMPLY WITH PUBLIC CONTRACT CODE SECTION 2600 ET SEQ. THE CONTRACTOR SHALL DEFEND, INDEMNIFY, AND HOLD HARMLESS THE DISTRICT FOR ANY FAILURE TO COMPLY WITH EDUCATION CODE SECTION 17407.5 OR PUBLIC CONTRACT CODE SECTION 2600 ET SEQ.

each Subcontractor shall not permit any person performing any of the Work to work more than eight hours during any one calendar day or more than forty hours during any one calendar week. The Contractor and any Subcontractor shall forfeit, as a penalty to the District, \$25 for each worker employed in the execution of the Work by the Contractor or the Subcontractor who is required or permitted to work more than eight hours in any one calendar day or forty hours in any calendar week in violation of Sections 1810 through 1815, inclusive, of the Labor Code. However, notwithstanding the foregoing, in accordance with Labor Code Section 1815, the Contractor or a Subcontractor may permit a worker to work in excess of eight hours per day, or forty hours per week, if all work in excess of such limits is compensated at a rate not less than one and one half times the worker's basic rate of pay.

10.8 Payroll Records. The Contractor and each Subcontractor must comply with all applicable provisions of Labor Code Sections 1776 and 1812, which relate to preparing and maintaining accurate payroll records, and making such payroll records available for review and copying by the District, the DIR's Division of Labor Standards Enforcement, and the DAS. The payroll records must be certified, maintained at the principal offices of the Contractor, and made available as required under Labor Code Section 1776. The Contractor must inform the District of the location at which the payroll records are located, including the street address, city, and county, and must, within five working days, provide a notice of any change of location and address. The Contractor and any Subcontractor that fails to timely comply with requests for

certified public records shall forfeit, as a penalty to the District, \$100 for each calendar day, or portion thereof, for each worker, until strict compliance is effectuated, and, in addition to penalties as provided by law, may be subject to debarment under Labor Code Section 1771.1.

10.9 Contractor Must Know and Comply with All Labor Laws. Notwithstanding anything to the contrary, the Contractor shall be responsible and liable for ascertaining, knowing, understanding and complying with all Legal Requirements applicable to the matters addressed in this Part 10, including, without limitation all such requirements specifically cited in this Part 10. If any provision of this Part 10 conflicts with, or is not a complete statement of, any provision of an applicable Legal Requirement, such Legal Requirement shall be deemed, respectively, to govern over or to expand on such provision of this Part 10, and the Contractor and each Subcontractor shall be required to comply in all respects with the applicable Legal Requirement.

PART 11 SCHEDULING OF WORK AND DELAYS

11.1 Early Work Schedule and Master Construction Schedule.

11.1.1 Early Work Schedule. Within seven days after receipt of the Notice of Award (if issued by the District) or prior to the Required Commencement Date, whichever is sooner, the Contractor must submit to the District and the Architect, for review and approval, a proposed schedule that specifies the Contractor's Project-related activities for the first 60 days of the Work ("Early Work Schedule"). If the District or the Architect requires revisions to the Early Work Schedule, the Contractor must correct and resubmit the Early Work Schedule promptly and within such time as will avoid any delay in the Work.

the Required Commencement Date, the Contractor must submit to the District and the Architect a schedule for completion of all of the Work ("Master Construction Schedule") that: (i) provides for a logical and orderly progression of the Work to completion within the Contract Time; and (ii) specifies time for accomplishing all activities and events needed for completion of the Project, including, without limitation, order deadlines for long-lead-time items, delivery dates for critical or special equipment and/or materials, and time for completion of commissioning and close-out activities. The Contractor must submit the Master Construction Schedule to the District and the Architect for review and approval. If the District or the Architect requires revisions to the Master Construction Schedule, the Contractor must correct and resubmit the Master Construction Schedule promptly and within such time as will avoid any delay in the Work. Upon approval of the Master Construction Schedule, it shall supersede the Early Work Schedule, and it may be changed only as provided in the Contract Documents. The Master Construction Schedule shall be developed using Primavera P6 format showing one defined critical path and illustrating the duration, start, and finish of any milestones.

11.1.3 Modifications. The Master Construction Schedule shall be subject to reasonable modification by the District from time to time during the course of construction of the Project, due to changes in the Project, circumstances beyond the control of the District, *et cetera*, and such modifications may result in the Master Construction Schedule not including any float that previously had been included in the Master Construction Schedule.

11.1.4 Updates and Revisions. Not less than once per month (starting with the second month after the Work commences), the Contractor must update and, as necessary, revise the Master Construction Schedule to accommodate authorized changes in the Work and/or Contract Time, as well as any other authorized modifications to the Master Construction Schedule. Each update of the Master Construction Schedule must also specify the actual commencement and completion dates for the scheduled activities. The Contractor must submit a narrative explaining any schedule modifications along with each monthly schedule update. The Contractor must submit copies of each such updated and/or revised Master Construction Schedule and narrative to the District and Architect as provided in Subsection 19.6.2 of these General Provisions.

11.1.5 Schedule Compliance and Recovery from Delays. Time is of the essence with respect to the performance of the Work. If the performance of any portion of the Work falls behind what is required by the Master Construction Schedule, the Contractor must, at its cost, accelerate and/or do all other things as necessary to complete the Project within the time specified in the Master Construction Schedule. In such event, the Contractor must submit to the District and the Architect a proposed revision to the Master Construction Schedule that clearly specifies how the Contractor will bring the Work back

into conformance with the Master Construction Schedule ("Recovery Schedule"). Each Recovery Schedule is subject to approval by the District and the Architect.

11.2 Work Outside Regular Working Hours.

11.2.1 Requested by Contractor. If the Contractor desires to perform any portion of the Work on, at or in the vicinity of the Project Site on any days or at any times other than during Regular Working Hours, or for more than eight working hours per day, the Contractor must obtain the written consent of the District, the Inspector of Record and/or any special inspector, and, if necessary, any city, County and other governmental agencies having competent jurisdiction. The Contractor shall in each such case be responsible for paying any and all additional or increased management, supervision, inspection and other costs incurred by the District on account of such work, and such costs shall be charged to the Contractor and/or deducted from amounts otherwise payable to the Contractor under the Contract.

11.2.2 Required by Contract or District. The Contractor must perform portion(s) of the Work on any days or at any times other than Regular Working Hours if such requirement is set forth in the Contract Documents. The Contractor also must perform portion(s) of the Work on any day or at times other than Regular Working Hours if required by the District for any reason that is not the fault of, caused by, or otherwise the responsibility of, the Contractor, in which case the District shall be responsible for paying any and all additional or increased management, supervision, inspection and other costs that it incurs.

11.3 Delays Resulting from Abnormal Weather.

and in the vicinity of the Project are, for all purposes relating to the Work, hereby deemed foreseeable. Therefore, the Contractor must accommodate in the proposed Master Construction Schedule the anticipated number of work days during which performance of the Work cannot occur or continue due to normal seasonal weather conditions (each a "Rain Day"). The number of Rain Days included in the Master Construction Schedule shall be determined based on weather data compiled by the National Oceanic and Atmospheric Administration that establishes the normal seasonal weather conditions for the general location of the Project Site and the time(s) of the year(s) during which the Work will be performed. The Contractor shall bear the risk attributable to all normal seasonal weather conditions, including, without limitation, precipitation, temperatures, winds, amount of daylight, et cetera. The Contractor shall include anticipated Rain Days in the Master Construction Schedule and as part of the critical path of the Project. Any Rain Days not used in any given month may be saved and used in any other month as necessary. Under no circumstances, however, can any saved and unused Rain Day be utilized to extend the Contract Time. Contractor is only entitled to a weather-related extension of time once all Rain Days have been used, and as provided in Section 11.3.2 below.

extension of time to complete the Work only if the performance of the Work is delayed by inclement weather in an amount, frequency, or duration in excess of the number of Rain Days determined under Subsection 11.3.1 of these General Provisions ("Abnormal Weather"). The District shall grant to the Contractor an extension of time for performance of the Work subject to the Contractor establishing, based on sufficient proof, that: (i) the weather conditions constituted Abnormal Weather, i.e., were in excess of "normal" Rain Days; (ii) the delay was caused by the Abnormal Weather; and (iii) the delay affected the critical-path activities of the Work.

- 11.4 Delays Arising from Hazardous Materials and Existing Structures. If any portion of the Project involves or relates to the presence, repair, modification, rehabilitation, reconstruction, demolition, removal or other work on or involving any existing structures, utilities, or other improvements, regardless of whether the Contractor is to perform such work, the District and the Contractor shall be deemed and construed, for purposes of Public Contract Code Section 7102 and otherwise, to have anticipated that asbestos, lead, petroleum distillates and/or other hazardous materials may exist or be discovered during the course of the Work that may delay or otherwise disrupt the performance of the Work. In such event, the Contractor's sole remedy for any delay attributable to the investigation, analysis, removal, abatement, decontamination and/or other actions necessary to correct such condition shall be an extension of time, unless the delay continues for more than ninety days past the date the hazardous materials were discovered, in which case the Contractor may be entitled to additional compensation, as provided in Section 11.6 of these General Provisions, for the period of delay in excess of ninety days.
- shall be required to pay any additional compensation to the Contractor or shall otherwise be liable for any costs attributable to a delay (each a "Non-Compensable Delay") if: (i) the cause of the delay was beyond the control of and without the fault of the District; (ii) the delay was reasonable under the circumstances involved; or (iii) the delay was within the contemplation of the District and the Contractor. For purposes of the Contract, delays within the contemplation of the District and the Contractor shall be deemed and construed to include, without limitation, delays attributable to: (i) normal seasonal weather conditions; (ii) coordination of the Work with any Work by Others; (iii) work on the Project that must be completed prior to some or all of the Work being commenced or completed; (iv) discovery of hazardous materials (including, without limitation, asbestos) if the Work or the Project involves or relates to the presence, repair, modification, rehabilitation, reconstruction, demolition, removal or other accommodation of existing structures, utilities, or other improvements; (v) the location, time of year and other conditions in which the Work is to be performed; and (vi) other matters typically attendant to construction projects of the same general type and scope as the Project. The Contractor's sole and exclusive remedy in the event of a Non-Compensable Delay shall be to seek an extension of time for performance of the Work.
- 11.6 Compensable Delays. The Contractor shall be entitled to compensation from the District on account of a delay in the performance of the Work (each a "Compensable Delay") only if: (i) the District caused or otherwise was responsible for the delay; (ii) the delay was unreasonable under the circumstances involved; and (iii) the delay was not within the contemplation of the District and Contractor. A delay shall not be considered to be a Compensable Delay to the extent the delay was caused, contributed to, or continued by the Contractor or any Subcontractor or other party or entity under the control or direction of, or otherwise performing any work or services on behalf of, the Contractor. A delay shall be considered a Compensable Delay only if and to the extent the delay adversely impacts a portion of the Work that is a critical-path item, and the District shall not be required to pay any compensation whatsoever to the Contractor (including, without limitation, any extended overhead, general conditions costs, impact costs, and/or out-of-sequence costs) in the absence of any such adverse impact on the critical-path of the Work. Subject to the Contractor's compliance with applicable requirements of this Part 11, and based on sufficient proof provided by the Contractor or otherwise obtained by or provided to the District, additional compensation to the Contractor for a Compensable Delay shall be set forth in and authorized by a Change Order or Construction Change Directive.
- **11.7 Contractor-Caused Delays.** Without limiting anything else in the Contract, The District shall be entitled to compensation from the Contractor on account of delays in the performance of the Work or

any Work by Others caused by, the fault of, or otherwise attributable to the Contractor or any employee, agent, Subcontractor or other person or entity acting on behalf of the Contractor. The Contractor shall not be deemed to be at fault for delays resulting from any tidal waves or any earthquakes in excess of 3.5 magnitude; provided that the portion of the Work thereby damaged had been constructed or otherwise performed in accordance with the Contract Documents. Compensation payable to the District on account of a delay for which the Contractor is responsible shall be at the rate provided in Section 13 of the Construction Services Agreement and per Section 11.8 below.

11.8 Liquidated Damages for Contractor-Caused Delays. The District and the Contractor hereby agree that it is impracticable and extremely difficult to ascertain the actual damages and costs the District will incur on account of a delay in completion of the Project or a phase or milestone thereof for which the Contractor is responsible. Therefore, to the extent the Contractor is responsible for any delay in meeting any milestone or final completion date applicable to the Project, or is otherwise responsible for compensating the District in connection with a delay in completing the Project or a phase or milestone thereof, the Contractor shall pay to the District such liquidated damages as provided in the Construction Services Agreement for each day (or portion thereof) of such delay. The District and the Contractor each hereby expressly agree that such liquidated damages amount constitutes fair and reasonable compensation to the District for such delay, regardless of whether such amount is at any time determined not to constitute actual full-compensation. The District may deduct any liquidated damages from amounts otherwise payable to the Contractor in accordance with the Contract and/or may pursue such other remedies as are permitted by law and/or the Contract. Nothing in this Section shall be deemed or construed to limit or preclude any right of the District to recover additional or other damages or costs if such right is allowed by law and/or expressly set forth elsewhere in any of the Contract Documents, including, without limitation, as provided in Section 2.7, Section 2.8, Section 7.4, Subsection 13.3.3, Section 13.6, and Subsection 15.1.4 of these General Provisions.

Mandatory Notice of Delay. Not later than five days after the start of any delay in the performance of the Work, regardless of the cause of the delay, and regardless of whether the delay is then ongoing, the Contractor must provide written notice of the delay to the District and the Architect ("Notice of Delay"). In any case that the Contractor fails to timely provide a Notice of Delay: (i) if it thereafter gives a Notice of Delay, the Contractor shall be deemed and construed to have waived and released any and all rights to an extension of time or additional compensation with respect to any time more than five days prior to the date the District actually receives the Notice of Delay; and (ii) the District may hold the Contractor responsible for any delays and/or increased costs that the District reasonably might have mitigated had the District received a timely Notice of Delay. A Notice of Delay must set forth the cause(s) of the delay and be accompanied by documentation reasonably evidencing and supporting the Contractor's position with respect to the cause(s) of the delay. The District or the Architect may request that the Contractor provide any additional or more detailed information regarding the delay, which the Contractor must provide within five days of request. THE GIVING OF A NOTICE OF DELAY IN CONNECTION WITH A DELAY SHALL BE DEEMED AND CONSTRUED AS A MANDATORY PREREQUISITE FOR ANY EXTENSION OF TIME AND/OR ADDITIONAL COMPENSATION TO THE CONTRACTOR ON ACCOUNT OF SUCH DELAY.

11.10 Requests for Additional Time and/or Compensation for Delays. If the Contractor desires to receive an extension of time or additional compensation on account of a delay, the Contractor must submit an associated Change Order Request within fourteen days of providing the Notice of Delay for that delay, regardless of whether the delay is still ongoing. If the District grants the Contractor an extension of

D-35

time for performance of the Work, the extension shall be proportionate to the actual delay in the performance of the Work, e.g., a half-day extension for a half-day delay. THE CONTRACTOR'S COMPLIANCE WITH THE FOREGOING PROCEDURAL REQUIREMENTS IN CONNECTION WITH A DELAY SHALL BE DEEMED AND CONSTRUED AS A MANDATORY PREREQUISITE FOR ANY EXTENSION OF TIME OR ADDITIONAL COMPENSATION TO THE CONTRACTOR ON ACCOUNT OF SUCH DELAY, AND FOR FILING OF A RELATED CLAIM IN ACCORDANCE WITH PART 22 OF THESE GENERAL PROVISIONS.

11.11 Contractor Claims Arising from Delays. IF THE CONTRACTOR DISPUTES ANY DETERMINATION MADE BY OR ON BEHALF OF THE DISTRICT IN REGARD TO THE CAUSE OF, RESPONSIBILITY FOR, EXTENSION OF TIME ATTRIBUTABLE TO, OR ADDITIONAL COMPENSATION ATTRIBUTABLE TO, ANY DELAY IN THE PERFORMANCE OF THE WORK, THEN, SUBJECT TO COMPLIANCE WITH THE REQUIREMENTS OF SECTIONS 11.9 AND 11.10 OF THESE GENERAL PROVISIONS, THE CONTRACTOR MAY FILE A CLAIM IN ACCORDANCE WITH PART 22 OF THESE GENERAL PROVISIONS.

PART 12 SUBSTITUTION OF SPECIFIED ITEMS

12.1 Requests for Substitution of Specified Items. Except as the Contract Documents expressly provide, any material, product, service or thing described in the Contract Documents as being required in connection with the Work and designated by specific brand or trade name (each a "Specified Item") shall be deemed and construed to set forth the minimum requirements for such Specified Item and to be followed by the words "or equal." To the extent permitted by this Part 12, the Contractor may offer in place of any Specified Item any substitute item that the Contractor can demonstrate is equal or better in all material respects to the Specified Item and that will adequately and fully accomplish the intended aesthetics, purposes and/or functions of the Specified Item. The District in no event shall be required to permit substitution of any Sole-Source Item designated in accordance with Public Contract Code Section 3400. The Contractor must request a substitution sufficiently in advance to avoid or prevent any delay in the Work or any Work by Others. In no event shall the Contractor substitute any item in place of a Specified Item unless and until the Contractor obtains written approval in accordance with this Part 12.

12.2 Conditions for Requesting Substitution.

12.2.1 Items That Become Commercially Unavailable. Unless the Contractor suggests a substitution as a means of implementing a value-engineering reduction in the GMP, the Contractor may request substitution of a Specified Item only if the Specified Item becomes commercially unavailable during the course of the Work. The Contractor in such event must provide documentation to the Architect that reasonably evidences that the Specified Item is no longer commercially available. An increase in the cost of a Specified Item shall not be deemed or construed to have made the Specified Item commercially unavailable. The Architect and/or the District may independently verify whether the Specified Item is no longer commercially available.

12.2.2 Items Still Commercially Available. The District, in its sole discretion, may, but is not required to, consider a request for substitution of a Specified Item that is still commercially available. However, in any such case, the Contractor shall be solely responsible and liable for: (i) any delays arising from the need to obtain approval of a proposed substitute item from the DSA and/or other governmental entity with competent jurisdiction; (ii) any costs of professional design services necessary to process and obtain any required approvals by the DSA and/or any other governmental entity with competent jurisdiction; and (iii) any costs of professional design services necessary to redesign or otherwise ensure coordination of the proposed substitute item with other portions of the Work or any Work by Others.

providing to the District and the Architect such documents, samples and other information as will reasonably substantiate that a proposed substitute item: (i) is equal or better in all material respects to the Specified Item; and (ii) will adequately and fully accomplish the intended aesthetics, purposes and/or functions of the Specified Item. Factors that the District and Architect will consider in determining whether a proposed substitute item satisfies the foregoing requirements may include, without limitation, whether the proposed substitute item will or will not: (i) have comparable durability and expected useful life; (ii) be consistent with the design and intended aesthetics of the Work and the Project; (iii) fit with, or require any change in the construction of, the Work and/or the Project; (iv) result in the District incurring more or less operations, maintenance and/or other costs; (v) have replacement parts and service available at least to the same extent as the Specified Item; and (vi) require any increase in the GMP and/or extension of the Contract Time. The Contractor must provide any and all information that would tend to indicate that a proposed substitute item is not suitable as a substitute for the Specified Item, and the Contractor

shall have breached its obligations under the Contract if it intentionally or negligently fails to provide all of such information. The types or forms of information submitted by the Contractor in connection with a request for substitution of a Specified Item must include, without limitation, all illustrations, specifications, catalog cut-sheets, manufacturer's brochures and other documentation that describe the characteristics, quality, and aesthetics of the proposed substitute item.

- **12.4 Certification of Requests for Substitution.** The Contractor must certify each request for substitution, on a form either provided by or approved by the District, and subject to penalty of perjury, that the Contractor: (i) has made all reasonable efforts to obtain all information relevant to the request for substitution of the Specified Item; (ii) has provided all of such information to the District and the Architect; and (iii) reasonably and in good faith believes that the proposed substitute item is equal or better in all material respects to the Specified Item, will adequately and fully accomplish the intended aesthetics, purposes and/or functions of the Specified Item, and otherwise satisfies all requirements of this Part 12.
- 22.5 Approval of Requests for Substitution. The District, in its sole discretion and after consultation with the Architect, shall determine whether a proposed substitute item: (i) is equal or better in all material respects to the Specified Item; (ii) will adequately and fully accomplish the intended aesthetics, purposes and/or functions of the Specified Item; and (iii) otherwise satisfies all requirements of this Part 12. Upon determining that a proposed substitute item satisfies all of the foregoing conditions, the District shall approve the request for substitution. The District, in its sole discretion, may, but is not required to, approve a request for substitution of a Specified Item despite the proposed substitute item not fully satisfying all of the foregoing conditions. The District also may impose conditions on the approval of any request for substitution, including, without limitation, the Contractor's agreement to provide an extended warranty on the substitute item or some other assurance of the compatibility, fitness, quality, durability, or performance of the substitute item. If the District approves a request for substitution of a Specified Item, the approval and all terms and conditions thereof shall be set forth in a Change Order, and execution of such Change Order by the Contractor shall be a condition to it taking or having any effect.
- 12.6 Disapproval of Requests for Substitution. If the District disapproves a request for substitution of a Specified Item that is commercially available, the Contractor must provide the Specified Item without any extension of the Contract Time or increase in the GMP. If the District disapproves a request for substitution of a Specified Item that is no longer commercially available, then, within such time as will avoid or prevent any delay in the Work or any Work by Others, the Contractor must submit an alternate proposal for substitution of the Specified Item in accordance with this Part 12.

PART 13 HAZARDOUS EQUIPMENT, MATERIALS AND SUBSTANCES

- 13.1 Use of Hazardous Materials. If the Contractor reasonably must use explosives, toxic solvents, or other hazardous equipment, materials or substances in order to adequately complete the Work, the Contractor must provide written notice to, and obtain the consent of, the District at least seven days prior to bringing such hazardous items onto the Project Site. The District may impose conditions on any such approval for use of any such hazardous items to ensure safety and protect the Work, Project and Project Site, and any such use must comply in all respects with applicable Legal Requirements. If any hazardous items must be stored at the Project Site, the Contractor must coordinate the location and means of storage with the District and local public officials having jurisdiction over such matters. The Contractor shall be solely responsible and liable in all respects for the safe, appropriate and lawful use, handling, and storage in connection with the Work of any hazardous equipment, materials and/or substances.
- Hazard Communication Program. The Contractor must comply with all requirements of the Safe Drinking Water and Toxic Enforcement Act of 1986 (Health and Safety Code Section 25249.5 et seq., commonly referred to as "Proposition 65") that are applicable to a "person in the course of doing business." Prior to commencing any portion of the Work, and as required by and in accordance with Cal-OSHA regulations and other applicable Legal Requirements, the Contractor must develop a written hazard communication program ("HCP") that specifies, among other required matters, criteria for: (i) labeling and/or other forms of warning in regard to hazardous substances to be used in connection with performance of the Work; (ii) making available all required material safety data sheets ("MSDS") for such substances; and (iii) informing and training employees in regard to dangers and proper handling of such substances. The Contractor must implement, maintain, and enforce its HCP at all times prior to full completion of the Work. The Contractor also must: (i) develop and provide to the District a complete list of hazardous substances brought onto or kept at the Project Site; (ii) make required MSDS available in a readily-accessible place at the Project Site; (iii) comply with requirements for giving notice to all persons who may be exposed to any chemical known to the State to cause cancer, including, without limitation, ensuring that any such substances brought onto or kept on the Project Site properly labeled; and (iv) ensure that all persons working with or in the vicinity of any such substances are informed of applicable hazards and trained in proper use and handling of such substances.

13.3 Asbestos and Asbestos-Containing Materials.

13.3.1 Contractor Awareness of Asbestos. The Contractor acknowledges that, if the Work involves any repair, modification, rehabilitation, reconstruction, demolition, removal or other work on or involving any existing structures, utilities, or other improvements, then asbestos or asbestos-containing materials likely will be present on or at the Project Site. The Contractor shall be deemed and construed for all purposes of the Contract to have undertaken the Work with full knowledge of the currently-accepted standards, hazards, risks, and liabilities associated with asbestos and asbestos-containing materials. The Contractor shall be solely responsible and liable for safely and appropriately performing any Work that may require demolition or removal, may uncover, reveal or otherwise expose, or may otherwise involve or relate to, asbestos or asbestos-containing materials.

13.3.2 Use of Asbestos Prohibited. Notwithstanding anything to the contrary, in no circumstances may the Contractor use or incorporate into the Work any asbestos or asbestos-containing materials, or use or employ in connection with the Work any equipment, tools, clothing or other things that contain or incorporate asbestos or asbestos-containing materials. For purposes of the Contract: (i)

"asbestos" means any naturally occurring fibrous hydrated mineral silicate, including, without limitation, chrysotile, crocidolite, amosite, fibrous tremolite, fibrous anthophyllite, and fibrous actinolite; and (ii) "asbestos-containing materials" means materials or products formed by mixing asbestos fibers with other materials, such as cement, rock wool, plaster, cellulose, clay, vermiculite, perlite, adhesive, et cetera. Within seven days after receipt of the Notice of Award (if issued by the District) or prior to the Required Commencement Date, whichever is sooner, the Contractor must provide to the District an executed copy of the "Certification Regarding Asbestos" form, which is included in the Required Contract Forms.

13.3.3 Liability for Asbestos. If the Contractor violates the prohibition set forth in Subsection 13.3.2 of these General Provisions or otherwise is responsible for asbestos contamination on, at or in the vicinity of the Project Site, the Contractor shall be solely responsible and liable for any and all costs attributable to: (i) correction of the Work; (ii) any and all investigations, analyses, removals, abatements, decontaminations or other actions necessary to correct the violation ("Asbestos Remediation"), including, without limitation, costs incurred by the District for additional administrative and professional services and for laboratory services, consultants, and contractors; and (iii) any injury to any person and/or damage to any property arising or alleged to have arisen from the violation. The District shall arrange for performance of any necessary Asbestos Remediation in accordance with all applicable Legal Requirements.

- 13.4 Discovery of Hazardous Materials. If, during performance of the Work, the Contractor encounters any materials that the Contractor reasonably believes to be hazardous materials that have not been rendered harmless, the Contractor must continue the Work in unaffected areas reasonably believed to be safe, but must immediately cease all Work in the area of the suspected hazardous materials and report the condition, in writing, to the District. The District will as necessary obtain the services of an independent and qualified person or company to identify the suspected hazardous materials, and to determine whether removal or some other corrective measures are required to render the materials harmless. The Contractor may resume work on the affected area only upon a determination that: (i) the materials identified by the Contractor are not hazardous and removal or other measures are not necessary; or (ii) removal or other measures necessary to render the materials harmless have been completed in accordance with applicable Legal Requirements. Any required asbestos-related work (defined in Health and Safety Code Section 25914.1) and/or hazardous substance removal (defined in Business and Professions Code Section 7058.7) that is not disclosed in the Contract Documents shall be performed under a separate District contract.
- 13.5 Discovery of Hazardous Waste in Excavations Deeper Than Four Feet. Section 14.7 of these General Provisions shall apply in the case of any suspected hazardous waste discovered in any trench or other excavation that extends deeper than four feet below the surface.
- 13.6 Contractor Responsibility for Releases of Hazardous Substances. If any person on, at or in the vicinity of the Project Site on account of the Work dumps, pours, spills, buries, places, discharges or otherwise releases any hazardous materials, waste or substances into or onto the Project Site or property in the vicinity of the Project Site, whether intentionally or otherwise, the Contractor shall be solely responsible and liable for any and all costs attributable to such release, including, without limitation: (i) costs of any necessary correction of the Work; (ii) any and all investigations, analyses, removals, abatements, decontaminations or other actions necessary to correct such release, including, without limitation, costs incurred by the District for additional administrative and professional services and for laboratory services, consultants, and contractors; and (iii) any injury to any person and/or damage to any property arising or alleged to have arisen from the violation. The Contractor's indemnification and other

D-40

obligations under Part 21 of these General Provisions shall apply with respect to any and all costs and other liabilities, excluding attorneys' fees, arising from any such release.
D-41

PART 14 EXCAVATIONS AND UTILITIES

- 14.1 Underground Utilities and Installations. Except as the Contract Documents expressly indicate otherwise in any particular case, the existence, locations and, if available, depths of underground utilities, facilities and/or installations indicated in the Contract Documents as being on, at or in the vicinity of the Work and/or the Project Site are: (i) based on records available to the District, not on surveys or excavations prepared or performed by the District; and (ii) are to be considered approximate, not exact. The District may, but shall not be required to, indicate in the Contract Documents the location of service laterals and/or appurtenances within the vicinity of the Work if the presence of such utilities can be inferred from the presence of other visible facilities, such as buildings, meters, junction boxes, et cetera. Prior to commencing any excavation in connection with the Work or any other activity that reasonably might damage underground utilities, facilities or installations, the Contractor must: (i) thoroughly inspect the vicinity of the Work for above-ground facilities, such as buildings, meters, junction boxes, et cetera, that might indicate the presence of underground service mains, trunklines, laterals or appurtenances; (ii) determine the exact location of any underground utilities, facilities and installations within an indicated approximate location, using air-vacuum excavation (i.e., potholing) techniques (or an underground-utility locating service to perform such services); (iii) immediately report to the District any utilities, facilities or installations located in the vicinity of the Work that are not indicated in the Contract Documents or are in a location materially different from the location indicated in the Contract Documents; and (iv) provide a written report to the District describing the exact location of all underground utilities, facilities and installations within the vicinity of the Work, including a diagram thereof, if necessary to adequately describe such exact locations. The Contractor must coordinate with and obtain the consent of the District prior to any potholing on, at or in the vicinity of the Project Site. Upon consent of the District, the Contractor may use a potholing technique other than air-vacuum excavation if that reasonably is not appropriate for the applicable soil type or conditions.
- 14.2 Regional Notification Center. If the Work involves any trenching, boring, tunneling, digging or other excavation, the Contractor shall be solely responsible and liable for compliance with all applicable requirements of Government Code Sections 4216 through 4216.9, and with all requirements of the Contractors State License Board relating to such Government Code provisions. The Contractor must, as required, obtain from the Regional Notification Center an Underground Service Alert identification number and must provide such identification number to the District. Prior to it expiring, the Contractor must contact the Regional Notification Center for any necessary revalidation of the identification number.
- Government Code Section 4215, if the Contractor, while performing the Work, discovers utility facilities not identified by the District in the Contract Documents, the Contractor shall immediately provide written notice to the District and the applicable utility company. The public utility, if it owns the utility facilities, shall have the sole discretion to perform any necessary repairs or relocation work or to permit the Contractor to do such repairs or relocation work at a reasonable price. The District shall not be required to indicate the presence of existing service laterals or appurtenances whenever the presence of such utilities on the Project Site can be inferred from the presence of other visible facilities, such as buildings, meter and junction boxes, on or adjacent to the Project Site. The District shall be responsible for the timely removal, relocation, or protection of existing main or trunkline utility facilities located on the Project Site, if such utilities are not identified by the District in the Contract Documents. The District shall compensate the Contractor for the costs of locating, repairing damage not due to the failure of the Contractor to exercise reasonable care, and removing or relocating such main or trunkline utility facilities not indicated

in the Drawings and Specifications with reasonable accuracy, and for equipment used in connection with the Work necessarily idled during such work. The Contractor shall not be assessed liquidated damages for delay in completion of the Work, when such delay was caused by the failure of the District or the owner of the utility to provide for removal or relocation of such main or trunkline utility facilities. The provisions of this Section shall not be deemed or construed to preclude the District from requiring changes in the Work that will eliminate the need to remove and/or relocate any utility facilities.

- **14.4 Coordination with Utility Companies.** Sufficiently in advance in order to avoid and/or prevent any delay in the Work or any Work by Others, the Contractor must coordinate each portion of the Work involving or requiring construction of, or connection to, utility facilities (including, without limitation, any relocation of facilities) with the applicable utility company. In any case that the owner of a utility has the option of performing any required work, but such owner permits the Contractor to perform such work, the Contractor must perform such work in compliance with all requirements of such owner.
- **14.5 Excavation Permits.** The Contractor must not commence any excavations required in connection with the Work until the Contractor or the appropriate Subcontractor has: (i) applied for and obtained all necessary permits for the excavations, including, without limitation, any OSHA and Cal-OSHA permits; (ii) provided a copy of each such permit to the District; and (iii) posted each such permit in a prominent location on the Project Site.
- **14.6** Safety Plans for Trenches Deeper Than Five Feet. If the GMP exceeds \$25,000, Section 9.3 of these General Provisions, which relates to Trench Safety Plans, shall apply in the case of any and all trenches that will or reasonably might be excavated to a depth of five feet or more.
- **Differing Conditions in Excavations Deeper Than Four Feet.** If the Work involves digging any trenches or other excavations that extend deeper than four feet below the surface, the Contractor shall promptly, and before the following conditions are disturbed, notify the District, in writing, of any: (i) material discovered during such excavation that the Contractor believes may be hazardous waste, as defined in Section 25117 of the Health and Safety Code, that is required to be removed to a Class I, Class II, or Class III disposal site in accordance with provisions of existing law; (ii) subsurface or latent physical conditions at the Project Site differing from those for which the Contractor is responsible in accordance with Section 7.1 of these General Provisions; or (iii) unknown physical conditions at the Project Site of any unusual nature, different materially from those ordinarily encountered and generally recognized as inherent in work of the character provided for in the Contract Documents. The District shall promptly investigate any such conditions identified by the Contractor. If the District determines that such conditions exist and cause a decrease or increase in the Contractor's cost of, or the time required for, performance of the Contract, the District shall issue a Change Order or Construction Change Directive in accordance with the procedures set forth in these General Provisions. In the event a dispute arises between the District and the Contractor in regard to whether any such condition actually exists, or causes a decrease or increase in the Contractor's cost of, or time required for, performance of the Contract, the Contractor shall not be excused from completing all Work in accordance with the milestones set forth in the Master Construction Schedule, but shall proceed with all Work to be performed under the Contract Documents. However, the Contractor shall retain all rights in such regard as provided by law or the Contract, and the Contractor may file a Claim in accordance with the Contract Documents.

PART 15 CHANGES IN THE WORK

15.1 Authorization Required for Changes in the Work.

strict accordance with the Drawings and Specifications and other Contract Documents approved and adopted for the Project, as those may from time to time be amended, supplemented, or otherwise modified in accordance with the Contract Documents. Except for the District and except to the extent of the Architect's authority to issue Architect Field Directives, no person or entity (including, without limitation, the Architect and Inspector of Record) has the unilateral authority to order any changes in the Work or to make any changes in the Drawings, Specifications, or other Contract Documents. Absent a duly-authorized Change Order, Construction Change Directive or Architect Field Directive, the Contractor must not change the Work, or permit any changes in the Work to occur, including, without limitation, any substitution, addition, omission, deviation or other change. Notwithstanding anything to the contrary, in order to be valid and enforceable against the District, each Change Order and Construction Change Directive must have been approved by the District Board directly or, as described in Section 2.1 of these General Provisions, the District Board must have delegated to the Authorized District Representative, in writing, the authority to approve such Change Order or Construction Change Directive.

approve changes to the Work, the Architect will be responsible for obtaining such DSA approval. Although subject to change, the DSA presently may approve changes in the Work by means of a Construction Change Document ("CCD") process. The Architect may seek CCD approval of any Change Order, Construction Change Directive, Architect Field Directive, or other document describing the Work or changes in the Work. Because such documents may be grouped together for purposes of obtaining DSA approval, a CCD as approved by the DSA will not necessarily correspond in reference-number or content to Change Orders or other documents issued under the Contract Documents.

15.1.3 Direction to Proceed With Changes. An Authorized District Representative may determine that it is necessary, in order to prevent delays in the Work, to direct the Contractor to proceed with changes in the Work that are included in a proposed Change Order, but the Change Order has not yet been approved or ratified by the District Board. In such event, the Authorized District Representative, to the extent so authorized, may issue written instructions to the Contractor to implement and proceed with such changes (each a "Construction Change Directive"). However, in no event may the Contractor receive any payment on account of any Work performed under a Construction Change Directive until the District Board has approved or ratified the corresponding Change Order.

15.1.4 Responsibility for Unauthorized Changes. The Contractor shall be solely responsible for any and all costs attributable to any unauthorized change in the Work from what is required under the Contract Documents, including, without limitation, all costs of any replacement or other correction of the Work and all costs incurred by the District for additional administrative and professional services in connection with such correction. In addition, the Contractor shall be deemed and construed to have waived any and all rights to any compensation for any change in the Work that was not duly authorized prior to being commenced or otherwise implemented.

15.2 Changes Required by Change Order. Subject to the provisions of this Part 15 and other requirements of the Contract Documents, the District may at any time issue an amendment to the Contract for purposes of ordering change(s) in the Work to be performed under the Contract,

adjustment(s) in the GMP and/or Contract Time, or other change(s) in the requirements of the Contract (each a "Change Order"). As provided in this Part 15, a Change Order may be unilateral if issued without approval by the Contractor or mutual if the District and the Contractor have both approved the Change Order. The Architect shall prepare each final Change Order and, upon taking effect, a Change Order shall constitute one of the Contract Documents and shall be subject to all other applicable provisions of the Contract Documents as if originally included therein. In no event shall any Change Order be deemed or construed to invalidate the Contract. The Contractor must implement the changes specified in a Change Order promptly or by such time as specified in the Change Order. The Contractor must perform all work necessary to complete any change(s) specified in a Change Order in accordance with all provisions of the Contract, except as any such provisions are expressly modified by the Change Order.

- 15.3 Changes Required by Architect Field Directive. At any time and without invalidating the Contract, the Architect may issue a written directive requiring any minor changes in the Work that are consistent with the intent of the Contract Documents and that do not require an adjustment in the GMP or Contract Time (each an "Architect Field Directive"). An Architect Field Directive must be signed by the District and the Architect in order to be valid and binding on the Contractor. Upon taking effect, an Architect Field Directive shall constitute one of the Contract Documents and shall be subject to all other applicable provisions of the Contract Documents as if originally included therein. The Contractor's approval of an Architect Field Directive shall not be required, and the Contractor must implement each minor change specified in an Architect Field Directive promptly or by such time as specified in the Architect Field Directive. IF THE CONTRACTOR BELIEVES THAT IT IS ENTITLED TO AN INCREASE IN CONTRACT TIME AND/OR GMP ON ACCOUNT OF AN ARCHITECTURAL FIELD DIRECTIVE, THE CONTRACTOR MUST PROVIDE NOTICE AS PROVIDED IN SECTION 15.5 OF THESE GENERAL PROVISIONS.
- 15.4 Changes Required by Unilateral Change Order. At any time and without invalidating the Contract, the District may issue a Change Order to require any changes in the Work that are within or consistent with the general scope of the Contract and/or the Project. In order to be binding on the Contractor, each such unilateral Change Order must be signed by the District and the Architect. The Contractor's approval of a unilateral Change Order shall not be required, and the Contractor must implement all changes specified in a unilateral Change Order promptly or by such time as specified therein. The District may, but shall not be required to, issue a unilateral Change Order in any case that the District and the Contractor have been unable to agree on the terms of a requested mutual Change Order. A unilateral Change Order may direct that any Work under the Change Order or otherwise be performed on a time-and-materials, lump-sum, or unit-pricing basis. IF THE CONTRACTOR DISAGREES WITH ANY OF THE TERMS OF A UNILATERAL CHANGE ORDER, THE CONTRACTOR MUST PROVIDE NOTICE AS PROVIDED IN SECTION 15.5 OF THESE GENERAL PROVISIONS.
- 15.5 Notice of Disagreement Regarding GMP or Time. The Contractor must provide written notice to the District and the Architect if the Contractor: (i) reasonably believes that the implementation of any Bulletin, Interpretation or Clarification will require an adjustment to the GMP and/or Contract Time that is not set forth in a corresponding Change Order; (ii) reasonably believes that it is entitled to an increase in the GMP and/or Contract Time on account of a change required under an Architect Field Directive; or (iii) reasonably disagrees with the adjustment to the GMP and/or the Contract Time, if any, set forth in a unilateral Change Order. Any such notice must set forth in reasonable detail all bases asserted by the Contractor in support of its position that it is entitled to an adjustment of the GMP and/or Contract Time, or that any specified adjustment of the GMP and/or Contract Time is not adequate. THE CONTRACTOR MUST PROVIDE SUCH NOTICE PRIOR TO COMMENCING ANY WORK OR OTHERWISE IMPLEMENTING THE APPLICABLE BULLETIN, INTERPRETATION, CLARIFICATION, ARCHITECT FIELD

D-45

DIRECTIVE OR UNILATERAL CHANGE ORDER, OR WITHIN THREE DAYS OF THE ISSUANCE OF SUCH DOCUMENT, WHICHEVER IS SOONER.

- under Section 15.5 of these General Provisions either prior to commencing any work or otherwise implementing any change required under a Bulletin, Interpretation, Clarification, Architect Field Directive or unilateral Change Order, or within three days of the issuance of such document, whichever is sooner, the Contractor shall be deemed and construed to have waived any and all rights to any adjustment in the GMP and/or Contract Time on account thereof. THE GIVING OF AN APPLICABLE NOTICE UNDER SECTION 15.5 OF THESE GENERAL PROVISIONS SHALL BE A CONDITION PRECEDENT TO THE CONTRACTOR HAVING ANY RIGHT, WHETHER UNDER A CLAIM FILED IN ACCORDANCE WITH PART 22 OF THESE GENERAL PROVISIONS OR OTHERWISE, TO ANY ADJUSTMENT (OR FURTHER ADJUSTMENT) OF THE GMP AND/OR CONTRACT TIME ON ACCOUNT OF THE APPLICABLE BULLETIN, INTERPRETATION, CLARIFICATION, ARCHITECT FIELD DIRECTIVE OR UNILATERAL CHANGE ORDER.
- 15.7 Changes Requested by the District. The District may at any time request that the Contractor propose any adjustments to the GMP and/or Contract Time attributable to any change in the Work or other requirements of the Contract desired by the District (each a "Request for Proposal" or "RFP"). The Architect shall prepare each RFP in writing and submit it to the Contractor with all information reasonably necessary to permit the Contractor to determine the nature and scope of the proposed change, including, if necessary or convenient, any Drawings and Specifications that illustrate or explain the changes. Within seven days of receipt of an RFP, and without additional compensation, the Contractor must provide to the District and the Architect a written proposal setting forth any proposed adjustments to the GMP and/or Contract Time that the Contractor reasonably believes are appropriate considering the nature and scope of the proposed change. Each proposal that includes a proposed adjustment to the GMP must be accompanied by an estimate of the effect (whether additive or deductive) of the change(s) on the amount of the GMP ("Change Order Cost") detailed in accordance with any requirements imposed under Section 15.9 of these General Provisions. Each proposal that includes a proposed adjustment to the Contract Time must set forth the impact of the proposed change(s) on any milestones and on the critical path of the Work as set forth in the Master Construction Schedule, not just specify an increase in the number of days desired for completion of all Work. The District may accept the Contractor's proposal for a Change Order, may attempt to negotiate terms for a Change Order that are different from those proposed by the Contractor, or may determine not to further pursue the change originally desired by the District.
- District issue a Change Order to provide for adjustments to the GMP and/or Contract Time attributable to any change in the Work or other requirements of the Contract desired by, or required of, the Contractor (each a "Change Order Request"). The Contractor may, among other reasons, base a Change Order Request on a Claim asserted by the Contractor. The Contractor must prepare each Change Order Request in writing and must submit it to the District and the Architect with: (i) all information reasonably necessary to permit the District and the Architect to determine the nature and scope of the proposed change; and (ii) the proposed adjustments to the GMP and/or Contract Time, if any, that the Contractor reasonably believes are appropriate considering the nature and scope of the proposed change. The Contractor must submit each Change Order Request a sufficient time in advance of when the change must be implemented in order to avoid and/or prevent any delay in the Work or the Project. Each such Change Order Request that includes a proposed adjustment to the GMP must be accompanied by an estimate of the Change

D-46

Order Cost detailed in accordance with any requirements imposed under Section 15.9 of these General Provisions. Each Change Order Request that includes a proposed adjustment to the Contract Time must set forth the impact of the proposed or other change(s) on any milestones and on the critical path of the Work as set forth in the Master Construction Schedule, not just specify an increase in the number of days desired for completion of all Work. The District may agree to the terms set forth in a Change Order Request, may attempt to negotiate terms for a Change Order that are different from those proposed by the Contractor, or may determine not to agree to the Change Order Request. If the District and the Contractor are able to agree on all terms of a Change Order Request, the Architect will prepare the Change Order to include all such terms.

Change Order Request, the Contractor must, at no additional cost, prepare a written estimate of the Change Order Cost, which must include a complete itemization of all materials, labor and other costs, whether additive or deductive, that affect the GMP, including, without limitation, estimates of hours of labor required, wage rates, material quantities, unit prices, et cetera. The District and/or the Architect reasonably may require that, in addition to typical line-item costs, one or more specific cost components or information be included in any cost estimate prepared by the Contractor in connection with any response to an RFP or any Change Order Request, and the estimate prepared by the Contractor must include such cost components. The District in its sole discretion may require that the Contractor provide any estimate of the Change Order Cost, or some portion thereof, on: (i) a "time and materials" basis as described in Section 15.10 of these General Provisions; (ii) a lump-sum basis as described in Section 15.11 of these General Provisions; and/or (iii) a unit-price basis as described in Section 15.12 of these General Provisions. Upon request, the Contractor must furnish to the District and the Architect such information as reasonably substantiates wage rates, bond premiums, or other amounts included in an estimate.

15.10 Determining Change Order Cost Based on Time and Materials. In the event the District requests that an estimate of a Change Order Cost, or portion thereof, be prepared on a time-and-materials basis, the Contractor must provide an itemization of the estimated costs of all time, materials and equipment necessary to complete the required change(s). Each time and materials estimate must include the cost components and conform to all associated requirements specified in Section 15.13 of these General Provisions, except that the Contractor must make reasonable and good-faith efforts to estimate and include in the estimate the maximum number of hours of labor in the various job classifications, and the maximum number of units of materials, required to complete the required change(s). Upon request of the District, the Parties shall determine a not-to-exceed amount for the maximum number of hours, the maximum number of units of materials, or both.

15.11 Determining Change Order Cost Based on Lump Sum Proposal. In the event the District requests that an estimate of a Change Order Cost, or portion thereof, be prepared on a lump-sum basis, the Contractor must provide an itemization of the costs of all time, materials and equipment necessary to complete the proposed change(s). Each lump-sum estimate must include the cost components and conform to all associated requirements specified in Section 15.13 of these General Provisions. Upon being duly approved by the Parties, a lump-sum estimate shall in all circumstances be deemed and construed to be the agreed Change Order Cost regardless of the total time, number of hours of labor, quantities of materials, et cetera, actually required to complete or otherwise implement the proposed change(s).

15.12 Determining Change Order Cost Based on Unit Pricing. In the event the District requests that an estimate of a Change Order Cost, or portion thereof, be prepared on the basis of unit-prices, the Contractor must provide an itemization of the costs of all time, materials and equipment necessary to

complete each logical, defined, or specified unit of the proposed change(s). The Contractor must make reasonable and good-faith efforts to estimate the maximum number of each logical, defined or specified unit required to complete the proposed change(s) on a unit-price basis. Each unit-price estimate must include the cost components and conform to all associated requirements specified in Section 15.13 of these General Provisions. Upon being approved by the Parties, a unit-price estimate shall in all circumstances be deemed and construed to be the Change Order Cost for each unit, regardless of the number or cost of such units actually required to complete or otherwise implement the proposed change(s).

15.13 Cost Components to be Included in All Estimates. The District may require that reasonable additional or modified cost components or information be included in any necessary cost estimate, but, otherwise, each estimate prepared by the Contractor in response to an RFP or in connection with a Change Order Request must include the following cost components and conform to all associated requirements specified below:

(i) Labor Costs:

Itemize all job classifications for labor necessary to complete the proposed change(s), direct hourly wage rates, and the estimated total number of hours in each job classification required to complete the change(s). Separately itemize any employer-paid payroll taxes, insurance, benefits and other costs attributable to such labor. Do not include off-site management, supervision and/or administration in this cost component, as the compensation for such costs shall be deemed to be included within the Contractor's general markup.

(ii) Materials Costs:

Itemize (in sufficient detail to identify) all materials necessary to complete the proposed change(s), quantities required, taxes, and any delivery costs. The amounts itemized in this cost component must be reduced by the full amount of any credits and/or discounts given in connection with obtaining the materials, as described in Section 15.15 of these General Provisions.

(iii) Equipment Costs:

Itemize all equipment necessary to complete the proposed change(s), hourly costs of rental or operations, and total number of hours required. Separately itemize any rented or leased equipment from any owned equipment. Separately itemize any equipment cost that is based on a per-load amount. Do not include in this cost component any hand tools, equipment with a value of less than \$1,000, or equipment with a daily rental rate of less than \$500, as the compensation for such items shall be deemed to be included within the Contractor's general markup. Also do not include in this cost component the rental of any equipment if other suitable equipment already is available at the Project Site, unless the use of such equipment would unreasonably delay the Work or Work by Others.

(iv) Subtotal:

Calculate the sum total of the labor, materials, and equipment costs determined in accordance with the foregoing clauses (i), (ii) and (iii).

(v) General Markup:

Specify an amount, in no event in excess of ten percent of the subtotal amount calculated in accordance with the foregoing clause (iv), which shall be deemed and construed to fully compensate the Contractor for overhead, profit and all other direct and indirect costs (other than bond markup)

attributable to the proposed change(s), including, without limitation, any and all costs of research; negotiations; preparation of estimates and other documents; insurance; home-office overhead; on-site and off-site supervision; interference, delay, acceleration and other impacts on the Work; guarantees; protection facilities; materials handling; supplies; safety equipment; and hand tools, equipment with a value of less than \$1,000, and equipment with a daily rental rate of less than \$500. Notwithstanding the foregoing, any portion of the work necessary to complete the proposed change(s) that is to be performed by any Subcontractor must not include a markup by the Subcontractor in excess of ten percent, or a markup by the Contractor in excess of five percent, of the total labor, materials and equipment included within such subcontracted work.

(vi) Bond Markup:

Specify an amount, in no event in excess of one percent of the subtotal amount calculated in accordance with the foregoing clause (iv), to compensate the Contractor for any additional bonding costs incurred in connection with the work necessary to complete the proposed change(s). Do not include any such amount if no additional bonding costs will be incurred.

(vii) Change Order Cost:

Calculate the total Change Order Cost, which shall be the sum total of the subtotal amount calculated in accordance with the foregoing clause (iv), the general markup specified in accordance with the foregoing clause (v), and any bond markup specified in accordance with the foregoing clause (vi).

15.14 Deductive or Reduced Change Order Costs. Any RFP or Change Order Request may propose any reduction in the amount and/or scope of the Work, regardless of whether the RFP or Change Order Request also proposes any additional or increased amount and/or scope of the same or other portions of the Work. In such event, the estimate prepared by the Contractor in response to the RFP or in connection with the Change Order Request must include the same cost components and conform to all associated requirements specified in Section 15.13 of these General Provisions, except that the estimate must determine the deduction from the GMP attributable to the reduction in the amount and/or scope of the Work. If an RFP or Change Order Request specifies only deductive changes(s), the Change Order Cost in its entirety will represent a deduction from the GMP. When both deductive change(s) and additive change(s) are specified in an RFP or Change Order Request, the Change Order Cost shall be based on the net effect on itemized costs, including, without limitation, general markup and bond markup.

15.15 Discounts and Refunds Deducted from Change Order Costs. The Contractor must make reasonable efforts to obtain or otherwise secure any and all discounts, rebates, refunds and/or offsets that may be available with respect to materials, equipment and supplies necessary, or no longer necessary, in connection with any change(s) in the Work or other requirements of the Contract. The Contractor must include in each estimate prepared in accordance with Section 15.13 of these General Provisions any such discounts, rebates, refunds and/or offsets as reasonably may be available. In the case of any change(s) completed on a time-and-materials basis or a unit-price basis, the Contractor must document any and all discounts, rebates, refunds and/or offsets as provided in Subsections 15.17.2 and 15.17.3 of these General Provisions.

15.16 Substantiation of Subcontractor Pricing Included in Estimates. If an estimate includes any work by a Subcontractor of any tier or materials provided by any materialman, the Contractor must furnish to the District and the Architect: (i) a detailed estimate, prepared and signed, as applicable, by the

Subcontractor or materialman, of the cost for labor, material, equipment, markup, et cetera; and (ii) such information as reasonably substantiates wage rates, bond premiums or other amounts included in the estimate, including, without limitation, any markup by the Subcontractor.

15.17 Substantiation of Time and Materials and Unit-Price Costs.

15.17.1 Requirement for Notice. The Contractor must not commence performance of any portion of the Work authorized to be performed on a time-and-materials basis or a unit-price basis unless the Contractor gives notice at least twenty-four hours in advance to the District, Architect and Inspector of Record that such Work will be commencing, so that they may be present during performance of such Work.

Requirements for Daily Time and Materials Reports. The Contractor must obtain the Inspector of Record's signature on a copy of the "Daily Time and Materials Report" form included in the Required Project Forms for each day during the performance of the Work, specifying: (i) the identification number assigned to that portion of the Work; (ii) the location and description of such Work; (iii) the job classifications, names and social security numbers of the workers performing such Work; (iv) the materials used in performing such Work; and (v) the equipment used in performing such Work, other than tools and equipment included within the general markup. The Contractor must: (i) obtain the Inspector of Record's signature the day Work described on a Daily Time and Materials Report is performed, or, if agreed by the Inspector of Record, the following day; and (ii) provide a copy of each Daily Time and Materials Report to the Inspector of Record upon obtaining his or her signature, or allow the Inspector of Record to copy the Daily Time and Materials Report at such time. Upon request, the Contractor must also submit any other relevant information as the District may require, including, without limitation, copies of wage rates as included in certified payroll records, receipts, payment invoices, shipping invoices, bills of lading, et cetera. If the Contractor fails to provide documentary evidence or other information sufficient to substantiate the amount and/or costs of Work performed on a time-andmaterials basis or unit-price basis, the District, in its reasonable discretion, may determine such amounts and/or costs.

15.17.3 Requirements for Separate Accounting Records. If the Contractor performs any Work (whether under the original Contract, any Change Order, or otherwise) on a time-and-materials basis or a unit-price basis, the Contractor must adequately document all labor, materials and equipment used and/or consumed in connection with such Work. The Contractor must prepare and maintain separate cost-accounting records, in accordance with generally-accepted accounting standards and principles, for each portion of the Work performed on a time-and-materials basis or unit-price basis, and shall make such accounting records available to the District, the State, and other parties to the same extent as required under the Contract Documents for other accounting records related to the Work.

15.18 Change Orders Include Full and Final Compensation. Except as expressly set forth in any particular Change Order, each Change Order shall be deemed and construed to include all adjustments to the GMP and/or Contract Time attributable to the work and/or other change(s) required under the Change Order, including, without limitation, any and all extensions of time and overhead, acceleration costs, profit, general conditions costs, expenses, and other direct and indirect costs and expenses of such work and/or changes. In addition, each Change Order shall be deemed and construed to include all necessary adjustments attributable to cumulative impacts of that and any and all preceding Change Orders, whether such impacts relate to scheduling, productivity or other matters. By signing a Change

Order, the Contractor shall be deemed and construed to have waived any and all Claims and rights to any adjustments to the GMP and/or Contract Time other than as are set forth in the Change Order, and the Contractor may not thereafter attempt to hold the District responsible for any interference, delay, acceleration, or other impact on the Work and/or additional costs attributable to the change(s) required under the Change Order.

15.19 Correction of Non-Conforming Work. The Contractor shall be responsible and liable, at its cost, for correction of Work necessary because of any portion of the Work was negligently performed, is defective, or otherwise does not conform with requirements of the Contract Documents, including, without limitation, because the Contractor failed to properly coordinate, schedule or supervise the incorrect portion of the Work.

PART 16 FINAL INSPECTION AND COMPLETION OF WORK

- 16.1 Contractor Must Determine When Work is Complete. The Contractor shall be solely responsible for determining when the Work or any portion thereof is complete, and the Contractor must base any such determination on adequate reviews and inspections of the Work by the Contractor's own forces, rather than relying on representations by any Subcontractor or others. At such time as the Contractor reasonably believes that all of the Work is substantially complete, the Contractor may request an inspection of the Work in accordance with Section 16.2 of these General Provisions. For purposes of the Contract Documents, "substantially complete" and "substantial completion" shall mean that the Project has been completed to such extent and to such condition that: (i) the District may take beneficial occupancy and use of, and may have unrestricted access to, all of the Project, and the District may commence operations therein, including, without limitation, educational programs; (ii) only minor cleaning, adjustment or similar corrective items that will not significantly interfere with such occupancy and use by the District, commonly referred to as "punch list" items, remain to be completed in order for the Work to be deemed fully completed; (iii) not as a limitation on the foregoing, all mechanical, electrical, plumbing and other building and site systems must, as applicable, be fully operational, tested and balanced, and start-up completed; and (iv) any and all approvals for which the Contractor is responsible in connection with the construction of the Project have been received from each governmental authority having jurisdiction over the construction, other than DSA close-out approval and final acceptance by the District.
- 16.2 Initial Request for Inspection. At such time as the Contractor reasonably believes that the Work is substantially complete, the Contractor must provide to the District, Architect and Inspector of Record: (i) written notice that all Work has been substantially completed and is ready for inspection; and (ii) a completed and executed copy of the "Certification Regarding Readiness for Inspection" form, which is one of the Required Project Forms (with original provided to the Architect). The District, Architect and Inspector of Record will perform such inspection within ten days following receipt of notice from the Contractor, and the District will notify the Contractor regarding the date and approximate time such inspection is to commence. The Contractor must conduct the examinations of the Work in a logical and sequential manner in order to facilitate an efficient and thorough inspection.
- 16.3 Re-Inspection of Work. If it is determined after the inspection described in Section 16.2 of these General Provisions, or after any subsequent re-inspection of the Work under this Section 16.3, that the Work is not substantially complete, the District and/or the Architect will provide written notice to the Contractor describing the incomplete and/or unsatisfactory portions of the Work. The Contractor must complete and/or correct all such Work within a reasonable time, not to exceed any time limit specified in the notice to the Contractor. Upon completing and/or correcting all such Work, Contractor must submit a new notice and a new Certification Regarding Readiness for Inspection as described in Section 16.2 of these General Provisions, and the re-inspection of the Work shall occur in accordance with the procedures set forth in that Section. Nothing shall be deemed or construed to require that any re-inspection of the Work be limited to only any incomplete and/or unsatisfactory portions of the Work previously identified and/or other portions of the Work affected thereby. Nothing in this Section or other provisions of this Part 16 shall be deemed or construed to constitute or serve as the basis for an extension of the Contract Time.
- **16.4 Determination that Work is Substantially Complete.** If it is determined after any inspection or re-inspection of the Work that it is substantially complete, the District and/or the Architect will provide

written notice of such determination to the Contractor. In such event, the District, Architect and/or Inspector of Record will also prepare and provide to the Contractor a punch-list of any minor items of the Work that must be completed and/or corrected in order for the Work to be fully completed and accepted by the District ("Remaining Work").

- and all Remaining Work within fourteen days of receiving the punch-list described in Section 16.4 of these General Provisions. If the Contractor fails to complete and/or correct the Remaining Work within the permitted fourteen-day period, the District may: (i) withhold from the final payment to Contractor under Section 19.12 of these General Provisions an amount equal to 150% of the Architect's estimate of the total cost to correct and/or complete all Remaining Work; and (ii) cause such Remaining Work to be completed and/or corrected and, thereafter, deduct the costs thereof from the amount withheld.
- 16.6 Contractor Request for Final Walk-Through. At such time as the Contractor reasonably believes that all Remaining Work has been adequately completed, the Contractor must provide a written request to the District, Architect and Inspector of Record for a final walk-through inspection. The District will coordinate and schedule the final walk-through inspection. The purpose of the final walk-through inspection will be to confirm that all Remaining Work has been completed in accordance with the Contract Documents or as otherwise required. At such time as the District, Architect and Inspector of Record determine that all Remaining Work has been completed in accordance with the Contract Documents or as otherwise required, and any and all other requirements of the Contract Documents have been satisfied, the District will provide written notice thereof to the Contractor. The District will recommend that the District Board accept the Work as complete only as provided in Sections 16.8 of these General Provisions.
- 16.7 Contractor Responsibility for Inspection Costs. The Contractor shall be solely responsible and liable for any and all costs (including, without limitation, costs of Architect and/or Inspector of Record services, administrative costs, consultant fees, transportation costs, et cetera) if: (i) the Contractor at any time requests an inspection of the Work without having a good-faith belief, based on its own knowledge, that the Work has been fully and properly completed and is ready for inspection; and/or (ii) the Contractor at any time requests an inspection of the Work and it thereafter becomes apparent that the Contractor knew or reasonably should have known that the Work was not fully and properly completed and ready for inspection. Not as a limitation on the foregoing, the Contractor shall be responsible for such costs if Work noted as incomplete and/or unsatisfactory during a prior inspection is not reasonably complete and/or satisfactory upon re-inspection, or, in the reasonable opinion of the District, Architect or Inspector of Record, the Contractor is using the inspection as a means to define the scope of any uncompleted portions of the Work or accelerate the Work of any Subcontractor. Any such costs shall be charged to the Contractor and/or deducted from amounts otherwise payable to the Contractor under the Contract.
- 16.8 Acceptance of the Completed Project. After the entirety of the Project has received final walk-through inspection approval, the completed Project must be accepted by action of the District Board. The District will cause a "Notice of Completion" for the Project to be recorded within fifteen days after the date the District Board takes such action to accept the Project (herein, the "Project Acceptance Date").

PART 17 CLOSE-OUT OF THE WORK

- 17.1 Close-Out Submittals are Prerequisite to Final Payment. Within ten days of receiving notice in accordance with Section 16.4 of these General Provisions that the Work is substantially complete ("Substantial Completion Date"), the Contractor must submit to the District all documents and other things required under this Part 17 that are within or associated with the scope of the Work, or are otherwise to be provided in connection with the Work. The requirements of this Part 17 shall be deemed and construed to be in addition to, and not in lieu of, any and all other close-out requirements set forth in the Contract Documents. However, if any matter addressed in this Part 17 is also addressed elsewhere in the Contract Documents, such provisions shall be construed to require compliance with all requirements of both such provisions or, if a conflict between such provisions exists, the provision located elsewhere in the Contract Documents shall govern. Notwithstanding anything to the contrary, the District shall not be required to make the final payment to the Contractor under Section 19.12 of these General Provisions unless and until the Contractor submits to the District all documents and other things required under this Part 17.
- **17.2** As-Built Drawings and Specifications. The Contractor must obtain final approval of the As-Built Drawings and Specifications from the District, Architect and Inspector of Record.

17.3 Equipment Operations and Maintenance Manuals.

sets, in three-ring binders, of: (i) manufacturer's manuals and/or instructions for operation, maintenance and repair of any and all systems, equipment, assemblies, and similar things incorporated into the Work, including, without limitation, any parts lists; (ii) any and all reports and other information resulting from any required commissioning, testing, balancing, et cetera, of such systems, equipment and other things; (iii) information (including, without limitation, contractor's license numbers, business license numbers, current addresses, telephone and facsimile numbers, and email addresses) identifying any and all Subcontractors and/or vendors associated with the purchase, installation, commissioning, testing and/or guarantee of such systems, equipment and other things; (iv) manufacturer guarantees and/or warranties for such systems, equipment, assemblies and other things; (v) special guarantees and warranties required by the Contract Documents; (vi) assignments of all guarantees and warranties from Subcontractors, materialmen and other persons and entities that furnished any labor, materials, services, goods or other things in connection with the Work; and (vii) any other associated information required by the Contract Documents. The Contractor must label, tab and otherwise organize such information so that the systems, equipment and/or other things to which they apply can easily and unambiguously be identified.

17.3.2 Required Correction and Certification. The Contractor must certify in writing that each set of binders is complete, accurate, and covers all of the Work, and such certification must be included immediately after the index page in each binder, and must be separately tabbed and indexed. Upon receipt, the District shall review the binders. If the District determines that the binders are not complete or otherwise must be corrected, the District will return all sets of the binders to the Contractor for correction. The Contractor must correct, re-certify, and return all sets of binders to the District within five days.

17.4 Log of Identification Tags and Labels. The Contractor must submit to the District a log or schedule of any and all equipment, valves, pipes, connections, meters and/or other things that, in accordance with the Contract Documents or applicable Legal Requirements, must be tagged or labeled.

The Contractor must label, tab and otherwise organize such information so that the systems, equipment and/or other things to which they apply can easily and unambiguously be identified.

- Documents in any particular case, the Contractor must submit to the District two sets of keys for each door, access panel, cabinet, gate, equipment cover, and other thing having any locking mechanism and/or capable of being locked. Each key must be securely attached to a fob or have some other means of securely attaching a label, other than attaching a label directly to the key. The description on the fob or label of each key must adequately identify the key, and all keys must be logged on a keying schedule or index that will permit the District to easily and unambiguously identify the doors, equipment and/or other things that are opened or operated by the keys. If the Specifications provide for electronic "key card" or similar systems, the Contractor must comply with all requirements of the Contract Documents or the District relating to identifying, logging or scheduling, and providing key cards or similar items to the District.
- 17.6 Tools, Spare Parts, Et Cetera. The Contractor must submit to the District any and all tools, spare parts, "attic" stock, et cetera, required under the Contract Documents. The Contractor must label, tab and otherwise organize such items so that the systems, equipment and/or other things to which they apply can easily and unambiguously be identified. In the event any spare parts, attic stock, or other item(s) reasonably would be too large, too heavy or in too great a quantity, to physically submit to the District, the Contractor must, subject to approval by the District, specify, on a log or schedule, all such items and their respective storage locations on the Project Site, and provide four copies of such log or schedule to the District.
- **17.7** Close-Out Materials. The Contractor must promptly provide to the District and Architect any and all forms, records and/or other documents within the Contractor's possession, control or scope of responsibility that the District and/or Architect reasonably request in connection with the close-out of the Work and/or the Project, including, without limitation, DSA close-out, if applicable.
- 17.8 Other Contract Document Close-Out Requirements. The Contractor, to the extent not already described in this Part 17, must as applicable obtain, complete and/or prepare, and must submit to the District, any and all close-out submittals and/or information required under any other requirement of the Contract Documents.
- **17.9 Contractor Guarantee.** The Contractor must provide to the District an executed copy of the "Contractor Guarantee" form, which is one of the Required Project Forms.

PART 18 CONTRACTOR GUARANTEE OF WORK

- 18.1 No Waiver of District Rights. In no event shall any payment to the Contractor, any provision of the Contract Documents, a Notice of Completion, or the use or occupancy of any portion of the Work or the Project, be deemed or construed: (i) to relieve the Contractor of any responsibility and/or liability for any defective or improper systems, equipment, materials or other things incorporated into the Work, faulty workmanship, or Work not performed in accordance with the Contract Documents and all applicable Legal Requirements ("Defective Work"); (ii) to constitute acceptance by the District, without recourse, of any Defective Work; or (iii) to constitute a waiver of any right the District has to hold the Contractor responsible and/or liable for any Defective Work.
- warranties of the Contractor under the Contract Documents, the Contractor hereby guarantees that, during all applicable Guarantee Periods: (i) all Work shall have been performed in accordance with all requirements of the Contract Documents and applicable Legal Requirements, and the Work and the Project are otherwise free of any Defective Work; and (ii) if any Defective Work is discovered during the applicable Guarantee Period, the Contractor shall at its cost repair, replace or otherwise correct the affected portion of the Work as provided in this Part 18 ("Contractor Guarantee"). Without limiting the foregoing, the Contractor Guarantee shall be deemed and construed to guarantee against any and all defects that may arise from any error or fault in any design(s) for which the Contractor was responsible in accordance with the Contract Documents, including, without limitation, any design provided by or through any Subcontractor or other person or entity in connection with the Work. No failure by the Inspector of Record or any other party to inspect or properly inspect any portion of the Work shall be deemed or construed to limit or otherwise condition the Contractor's responsibilities and/or liabilities under the Contractor Guarantee.
- **18.3 Limitations on Contractor Guarantee.** The Contractor Guarantee does not guarantee against damage to the Work: (i) caused by the District or any persons or entities other than the Contractor or other than any Subcontractor or other person or entity on, at, or in the vicinity of the Project Site on account of the Work; (ii) resulting from a lack of reasonable maintenance after the Substantial Completion Date; or (iii) resulting from changes to the Work performed by any persons or entities for whom the Contractor is not directly or indirectly responsible, unless the changes were performed in accordance with the Contract Documents and/or instructions or directions provided by the Contractor.
- 18.4 Applicable Guarantee Periods. Except as the Contract Documents otherwise provide, the Contractor Guarantee shall be and remain in effect at all times during the period ("Guarantee Period") that commences on the Substantial Completion Date and ends on the sooner of: (i) the date that is one year after the Project Acceptance Date; or (ii) the date that is two years after the Substantial Completion Date. The foregoing definition of the Guarantee Period shall be deemed and construed to apply to the Work generally, and shall not be deemed or construed to supersede or limit any provision of the Contract Documents that specifically requires a longer Guarantee Period. In no event shall any applicable Guarantee Period serve as a limitation with respect to latent defects in the Work, which remain subject to applicable statute(s) of limitation.
- **18.5** Specific Guarantee Periods for HVAC and Roofing. The Guarantee Period for all heating, ventilation and air conditioning equipment, controls, et cetera, shall start when such portion of the Work first commences and shall end on the sooner of: (i) the date that is two years after the Project Acceptance

Date; or (ii) three years after the Substantial Completion Date. The Guarantee Period for all roofing materials, membranes, sheet-metal, *et cetera*, shall start when such portion of the Work first commences and shall end on the date that is five years after the Substantial Completion Date.

- 18.6 Manufacturer and Other Third-Party Guarantees. The Contractor Guarantee shall in no event be deemed or construed to limit, in any manner, any manufacturer or other third-party guarantee or warranty (including, without limitation, any that have a longer applicable Guarantee Period). No such manufacturer or other third-party guarantee or warranty shall be deemed or construed to relieve the Contractor from its responsibilities and/or liabilities under the Contractor Guarantee. At all times while the Contractor Guarantee is in effect during an applicable Guarantee Period, but not thereafter, the Contractor must assist the District in processing any manufacturer and other third-party guarantee or warranty claims with respect to systems, equipment, materials and/or other things incorporated into the Project as part of the Work.
- 18.7 Guarantee Work by Contractor. Within ten days after receipt of written notice from the District, the Contractor, at no cost to the District, must repair, replace or otherwise correct: (i) any Defective Work that is discovered or revealed during an applicable Guarantee Period; and (ii) any systems, equipment, materials and/or other things damaged, destroyed or otherwise disturbed as a consequence of the repair, replacement or other correction of the Defective Work ("Guarantee Work"). All such Guarantee Work must result in a repair, replacement or other correction that satisfies all requirements of the Contract Documents or otherwise must be completed in accordance with District requirements. The Contractor must coordinate all Guarantee Work with the District in order to avoid interfering with District operations and/or endangering anyone at the Project Site. If the Contractor timely commences any Guarantee Work, but the Guarantee Work reasonably cannot be completed within ten days of notice from the District, the District shall determine a reasonable time within which the Contractor must complete the Guarantee Work. The Contractor must provide written notice to the District upon completing any Guarantee Work.
- 18.8 District Performance of Guarantee Work. The District may at any time cause any required Guarantee Work to be performed, as reasonably determined by the District, if: (i) the Contractor fails to undertake and/or complete the Guarantee Work within the time permitted under Section 18.7 of these General Provisions; or (ii) as reasonably determined by the District, an Emergency situation exists and the delay that would result from providing notice to the Contractor and permitting the Contractor to perform the Guarantee Work would endanger or further endanger any person(s) or property. The costs incurred by the District in connection with such Guarantee Work shall be assessed against the Contractor, and shall accrue interest at the maximum legal rate permitted in the State. In no event shall the District causing any Guarantee Work to be performed in accordance with this Section be deemed or construed to limit or otherwise condition the responsibilities and/or liabilities of the Contractor under the Contractor Guarantee.
- 18.9 Warranty of Title to Work. The Contractor further warrants that title to all systems, equipment, materials and other things incorporated into Work will pass to the District upon receipt of payment by Contractor, free and clear of all claims, liens, stop notices, security interests, charges, et cetera (each, for purposes of this Section, a "lien"). The Contractor's indemnification and other obligations under Part 21 of these General Provisions shall apply with respect to any and all such liens. The foregoing shall not be deemed or construed to: (i) prohibit the Contractor from asserting any Claim in accordance with the Contract Documents; or (ii) require that the Contractor deliver to the District title to any utility metering devices or similar equipment owned by any public or private utility company or service and installed as part of the Work for purposes of providing permanent utilities or services to or for the Project.

PART 19 PROGRESS PAYMENTS AND FINAL PAYMENT

Schedule of Values. Within ten days after receipt of the Notice of Award (if issued by the District) or prior to the Required Commencement Date, whichever is sooner, the Contractor must submit to the District, Architect and Inspector of Record a proposed schedule of the values allocated to the various portions of the Work ("Schedule of Values"), which must, among other things: (i) list the true actual cost (in dollars) of each separate activity and item included within the Work for which payment will be requested; (ii) specify the dollar amounts of overhead, profit, "general conditions" costs, and similar cost-items allocated to each activity and item included in the Work; (iii) separately itemize rough and finish work for the basic trades; (iv) specify individual dollar amounts for "large dollar" purchases, including, without limitation, systems, equipment, materials and/or other things to be incorporated into the Work; (v) allocate a value amount equal to not less than five percent of the total GMP to the closeout activities specified in Part 17 of these General Provisions; and (vi) consistent with the approved Master Construction Schedule, specify the projected total dollar amounts payable to the Contractor each month during the course of the Work. The Schedule of Values must also list as separate line items the costs for mobilization of the Project, bonds, insurance (if applicable), the Project Contingency, Contractor Contingency, allowances, and punch list items. The Schedule of Values need not specify the Leaseback Payments that will become due following District Board acceptance of the Project under Section 16.8 of these General Provisions, but must account for all of the Work. The Contractor shall not "front-load" the Schedule of Values by allocating increased or otherwise false dollar amounts to activities and/or items required to be performed in the early stages of the Work. The dollar amounts allocated for overhead, profit, general conditions costs, and similar cost-items must, as noted above, be allocated in the Schedule of Values to each particular activity; provided that the District, in its sole discretion, may consent to general conditions costs being allocated over the time for completion of the critical-path construction of the Project, as such critical path is set forth in the Master Construction Schedule. However, at any time the District reasonably determines that the progress of the critical-path portion of the Work is insufficient in comparison to what is specified in the Master Construction Schedule, as most recently updated, or in any approved recovery schedule, the District may adjust the payment of general conditions costs to match the percentage of the Project then complete, including, without limitation, retaining appropriate portion(s) of payments otherwise due to the Contractor. The District, in its sole discretion, may use the values allocated in the Schedule of Values as costs of activities and/or items that are eliminated from the scope of the Work. The Contractor must submit its proposed Schedule of Values with such documentary or other supporting evidence as reasonably supports and substantiates the allocations of values to the various portions of the Work. The Contractor must submit the Schedule of Values on such forms or software programs (e.g., Microsoft Project, Prolog, Expedition or Primavera) as approved by the Architect.

19.2 Contractor Must Obtain Approval of Schedule of Values. The primary purpose of the Schedule of Values is to serve as one basis for reviewing each Progress Payment Request submitted by the Contractor. Therefore, the Schedule of Values is subject to reasonable approval by the District, Architect and Inspector of Record, and the District shall not be required to make any payment to the Contractor unless and until the Contractor has obtained approval of an acceptable Schedule of Values. The District and/or Architect may require that the Contractor modify the Schedule of Values as they determine reasonably necessary. If the Schedule of Values is returned to the Contractor for any such modifications, the Contractor must modify and resubmit the Schedule of Values within two days. If the Contractor objects to any such required modification, the Contractor must, within such two-day period, provide any additional information to the District, Architect and Inspector of Record that the Contractor

believes supports and substantiates that the modification is not necessary. The Architect's decision regarding any such disputed modification shall be final and binding.

- 19.3 Contractor Must Monitor and Update Schedule of Values. The Contractor must update the Schedule of Values from time to time as reasonably necessary during the course of the Work, but in no event less than once per month or less than fourteen days prior to when amounts allocated in the Schedule of Values are to become due and payable. In each such update, the Contractor must identify and account for any changes in the Schedule of Values arising from delays, changes in the Work, changes in the Master Construction Schedule, et cetera. The Contractor must submit each update to the Schedule of Values for approval in accordance with the requirements set forth in Section 19.2 of these General Provisions that are applicable to the initial proposed Schedule of Values.
- 19.4 Materials Not Incorporated into Work. Notwithstanding anything to the contrary, the District shall not be required to pay the Contractor for any systems, equipment, materials or other things that have been purchased by the Contractor and stored on, at or in the vicinity of the Project Site, but not incorporated into the Work. The Contractor shall retain full responsibility and liability for any such things at all times, and for completing the Work in accordance with all applicable requirements of the Contract Documents.
- 19.5 Contractor Must Arrange Progress Payment Review Meetings. The Contractor must arrange to meet at the Project Site with the District, Architect and Inspector of Record in advance, but not more than five days in advance, of submitting each Progress Payment Request (other than the final Progress Payment Request), to review and discuss the Work for which the Contractor intends to seek payment in such Progress Payment Request (each a "Progress Payment Review Meeting"). The Contractor must provide to the meeting attendees, at the beginning of each Progress Payment Review Meeting, a comprehensive list of the Work that the Contractor intends to include within the scope of the Progress Payment Request. Upon request, the Contractor must show such Work to the meeting attendees and respond to relevant questions. If the District, Architect or Inspector of Record requests additional documentary or other support of the proposed Progress Payment Request, the Contractor must submit such documentary or other support with the Progress Payment Request submitted in accordance with Section 19.6 of these General Provisions. Such documentary or other support may include, without limitation, purchase invoices, rental receipts, delivery slips, certified payroll records, et cetera.

19.6 Contractor Submittal of Progress Payment Requests.

19.6.1 Timing and Content of Progress Payment Request. Not later than the seventh day of each month during the course of the Work, the Contractor must submit to the Architect a written request for payment on account of the Work completed during (or, if not previously compensated, prior to) the immediately preceding month (each a "Progress Payment Request"). Each Progress Payment Request must be submitted using a copy of the "Progress Payment Request" form, which is one of the Required Project Forms. In addition to any other information required by the Progress Payment Request form, the Contractor must itemize, in writing: (i) the total payment amount requested on account of the Work covered by the Progress Payment Request; (ii) the portions of such payment amount allocated to the various activities and items included in such Work; (iii) the portions of such payment amount attributable to each Subcontractor, materialman, and other person or entity that has furnished labor, material and/or equipment in connection with such Work; and (iv) the total balance due to each such Subcontractor, materialman and other person or entity after payment is made to them on account of the Progress Payment Request. With each Progress Payment Request, as additional information intended to assist in review of Progress Payment Requests, the Contractor must provide a reasonable, good-faith

D-59

892.00134

written estimate of the percentage, in relation to all Work required under the Contract Documents, of each of: (i) the total Work (i.e., on a cumulative basis) completed as of the end of the payment period covered by the Progress Payment Request; and (ii) the portion of the GMP allocable to such total Work; and (iii) each of the overhead, profit, general conditions costs, and similar cost items allocable to such total Work. No Progress Payment Request shall be deemed or construed to constitute a complete and valid request for progress payment unless and until: (i) the Contractor submits it with all required materials specified in Subsection 19.6.2 (or, if applicable, Subsection 19.6.3) of these General Provisions; and (ii) the Contractor provides the copies of the Progress Payment Request and all such required materials as specified in Subsection 19.6.4 of these General Provisions.

19.6.2 Materials to be Submitted with Progress Payment Request. The Contractor must submit, with each Progress Payment Request, all of the following: (i) an updated Master Construction Schedule showing changes since the Contractor provided the immediately prior update to the District; (ii) the updated Schedule of Values covering the payment period, as adjusted on account of any updates to the Master Construction Schedule made during the payment period; (iii) copies of all permits or other governmental licenses and approvals relating to the Work that were obtained or issued during the payment period; (iv) all information required under Section 3.11 of these General Provisions relating to As-Built Drawings and Specifications; (v) copies of any certified payroll records requested by the District; and (vi) conditional and unconditional waivers and releases as required under Section 19.7 of these General Provisions.

19.6.3 Final Progress Payment Request. Notwithstanding the requirement set forth in Subsection 19.6.1 of these General Provisions for submitting Progress Payment Requests by the seventh day of each month, the Contractor may submit a final Progress Payment Request to the Architect only after the District Board has accepted all of the Work in accordance with Section 16.8 of these General Provisions. The Contractor must specify in the final Progress Payment Request that Contractor is seeking final payment in the form of a release by the District of all Retention. The Contractor must, as applicable, submit with the final Progress Payment Request all of the materials specified in clauses (iii) through (vii), inclusive, of Subsection 19.6.2 of these General Provisions.

19.6.4 Copies of Progress Payment Requests and Supporting Materials. Concurrently with submitting any Progress Payment Request and other information to the Architect as required by this Section 19.6, the Contractor must submit copies of all such information to the District and Inspector of Record.

19.7 Requirements for Progress Payment Waivers and Releases.

19.7.1 Conditional Waivers and Releases for Contractor. With each Progress Payment Request that it submits in accordance with this Part 19 (other than the final Progress Payment Request), the Contractor must also submit an executed copy of the "Conditional Waiver and Release (Progress Payment)" form, which is one of the Required Project Forms. Such Conditional Waiver and Release (Progress Payment) must conditionally waive all lien and stop notice rights the Contractor may have against the District, the Project Site and the Project, with respect to all payments to be made to the Contractor on account of the Progress Payment Request. Each Conditional Waiver and Release (Progress Payment) must be duly executed and must contain an original signature of an Authorized Contractor Representative.

19.7.2 Unconditional Waivers and Releases for Contractor. With each Progress Payment Request that it submits in accordance with this Part 19, the Contractor must also submit an

executed copy of the "Unconditional Waiver and Release (Progress Payment)" form, which is one of the Required Project Forms. Such Unconditional Waiver and Release (Progress Payment) must unconditionally and irrevocably waive all lien and stop notice rights the Contractor may have against the District, the Project Site and the Project, with respect to all payments actually made to the Contractor on account of any prior Progress Payment Request. The Contractor must submit an Unconditional Waiver and Release (Progress Payment) in connection with a Progress Payment Request only if it was paid any funds on account of any prior Progress Payment Request for which it has not already submitted an Unconditional Waiver and Release (Progress Payment). Each Unconditional Waiver and Release (Progress Payment) must be duly executed and must contain an original signature of an Authorized Contractor Representative.

19.7.3 Unconditional Waivers and Releases for Subcontractors and Others. With each Progress Payment Request that it submits in accordance with this Part 19, the Contractor must deliver to the District an executed copy of the "Unconditional Waiver and Release (Progress Payment)" form, which is one of the Required Project Forms, for each Subcontractor, materialman and other person or entity that provided any labor, services, materials or equipment in connection with the Work. Any such person or entity must provide an Unconditional Waiver and Release (Progress Payment) in connection with a Progress Payment Request only if it was paid any funds on account of any prior Progress Payment Request for which it has not already submitted an Unconditional Waiver and Release (Progress Payment). Each Unconditional Waiver and Release (Progress Payment) must be duly executed and must contain an original signature of the person or entity's authorized representative.

19.7.4 Waiver and Release for Final Payment to Contractor. As a condition precedent to the District's obligation to make the final payment to the Contractor under Section 19.12 of these General Provisions, the Contractor must submit with its final Progress Payment Request an executed copy of the "Conditional Waiver and Release (Final Payment)" form, which is one of the Required Project Forms. The Conditional Waiver and Release (Final Payment) must be duly executed and must contain an original signature of an Authorized Contractor Representative.

19.7.5 Waivers and Releases for Final Payment to Subcontractors and Others. As a continuing obligation of the Contractor after final payment to the Contractor under Section 19.12 of these General Provisions, the Contractor must deliver to the District, within fourteen days following such final payment, an executed copy of the "Unconditional Waiver and Release (Final Payment)" form, which is one of the Required Project Forms, for each Subcontractor, materialman and other person or entity that provided any labor, services, materials or equipment in connection with the Work. Each such Unconditional Waiver and Release (Final Payment) must be duly executed and must contain an original signature of the person or entity's authorized representative.

19.8 Summary of Public Contract Code Section 20104.50. The State Legislature enacted Public Contract Code Section 20104.50 to ensure that contractors on certain public works projects are timely paid for their services on such projects. If a local public agency fails to pay an undisputed and properly submitted payment request within thirty days, the agency must pay interest at the legal rate set forth in subdivision (a) of Section 685.010 of the Code of Civil Procedure. Each agency must review payment requests as soon as practicable to determine if they are proper and suitable for payment. If a payment request is not proper, the agency must return it to the contractor within seven days, specifying in writing the reasons why it is not proper. If the agency returns an improper payment request to the contractor more than seven days after receipt, the number of days available to the agency to make payments without incurring interest will be reduced by the number of days by which the agency exceeds the seven-day

return requirement. The provisions of this Part 19 encompass, among other things, the requirements of Public Contract Code Section 20104.50

19.9 Architect Decisions Regarding Progress Payment Requests.

19.9.1 Notice of Approval or Disapproval. Within seven days after receipt from the Contractor of a complete and valid Progress Payment Request, or as soon thereafter as reasonably practicable, and after having considered input from the District, and the Inspector of Record, the Architect will provide written notice to the District, Inspector of Record, and the Contractor regarding whether the Architect approves or disapproves the Progress Payment Request, in whole or in part. The Architect's decision regarding whether any justifiable reason exists to disapprove all or a portion of a Progress Payment Request shall be final and binding.

Payment Request in whole or in part, the Architect will concurrently submit to the District and a written certification of the amount(s), based on the percentage of completion of the Work as of the date of the Progress Payment Request, that is payable to the Contractor on account of such Work ("Certification of Payment"). Each Certification of Payment will certify, to the best of the Architect's knowledge, information, and belief, and based on the Architect's observations and information in the Architect's possession, that all Work covered by the Progress Payment Request has been completed in accordance with the Contract Documents.

19.9.3 Rejection and Resubmittal of Progress Payment Requests. If the Architect disapproves all or any portion of any Progress Payment Request, then, with the notice required under Subsection 19.9.1 of these General Provisions, the Architect will return the Progress Payment Request to the Contractor with a written statement setting forth the reason(s) why it was rejected. The Contractor may correct the deficiencies in any disapproved Progress Payment Request and resubmit it without delay, but otherwise in accordance with Section 19.6 of these General Provisions. In applicable cases, the Architect will provide to the District with the Certification of Payment for a resubmitted Progress Payment Request, a written notice specifying the number of days by which the Architect exceeded the seven-day return requirement.

19.10 Retention from Construction Progress Payments. In addition to any amount(s) withheld from any Construction Progress Payment in accordance with the Contract or applicable Legal Requirement, the District shall withhold from each Construction Progress Payment (other than the final Construction Progress Payment) an amount equal to five percent of the total payment amount specified in the applicable Certification of Payment as security for adequate performance under the Contract ("Retention"). Notwithstanding the foregoing, after the Work is at least fifty percent complete, if the District Board determines that the Work is satisfactorily progressing, the District Board, in its sole discretion, may pay some or all of the remaining Construction Progress Payments (other than the final Construction Progress Payment) in full to the Contractor. Subject to the District's right to withhold some or all of the Retention as provided by the Contract or applicable Legal Requirement, the District will pay the Retention to the Contractor, as the final Construction Progress Payment, under Section 19.12 of these General Provisions. The District shall not be required to pay any interest on any Retention withheld under this Section 19.10. Upon request and at the sole cost and expense of the Contractor, the District will permit substitution of securities in lieu of the District withholding Retention, as provided in Public Contract Code Section 22300. Subject to any restrictions in Public Contract Code Section 22300, the District shall have the right to direct or approve any or all such securities.

- Certification of Payment for a Progress Payment Request submitted by the Contractor, the District shall pay to the Contractor an amount equal to the amount certified in the Certification of Payment less any duly withheld amounts (each a "Construction Progress Payment"). The District shall pay each such Construction Progress Payment (other than the final Construction Progress Payment) to the Contractor within thirty days after receipt from the Contractor of the complete Progress Payment Request; provided, however, that the District shall make the Construction Progress Payment to the Contractor an appropriate number of days less than thirty days if the Certification of Payment relates to a complete Progress Payment Request that was disapproved, but returned to the Contractor more than seven days after receipt. If the District fails to timely pay an undisputed Construction Progress Payment to the Contractor, the unpaid amount shall accrue interest, at the legal rate specified in Code of Civil Procedure Section 685.010, for each day the payment is late. The District will pay the final Construction Progress Payment to the Contractor as provided in Section 19.12 of these General Provisions.
- 19.12 Final Payment to Contractor of Retention. The District shall release to the Contractor, as the final Construction Progress Payment, any and all Retention, less any amount(s) the District deems necessary to withhold as provided by the Contract or applicable Legal Requirement, not sooner than 35 days after a Notice of Completion for the Work is recorded, but not later than 60 days after the first to occur of: (i) the District records a Notice of Completion for the Work; or (ii) "completion" of the Work is deemed to have occurred in accordance with Public Contract Code Section 7107. If some or all of any Retention is held in the form of securities, the District shall release such securities in accordance with the foregoing. In the event the District releases Retention to the Contractor because completion is deemed to have occurred in accordance with Public Contract Code Section 7107, the Contractor shall not thereby be deemed or construed to have been released from its obligations under the Contract, but the Contractor may terminate the Contract for cause at any time after the prerequisites set forth in Section 20.9 of these General Provisions have been satisfied.
- 19.13 District Issuance of Joint Checks. The District, in its sole discretion, may determine that it is necessary or advisable to issue any payment to the Contractor in the form of a joint check made payable to the Contractor and any of its Subcontractors, materialmen, or other persons or entities. The joint check payees shall be responsible for the allocation and disbursement of such funds between them. Except as may be required by applicable Legal Requirement, the District shall have no obligation to pay, or to ensure the payment of, any Subcontractor, materialman, or other person or entity that has furnished any labor, materials, services, goods or other things in connection with the Work. In no event shall the issuance by the District of any joint check be deemed or construed to: (i) constitute an understanding or agreement between the District and any such Subcontractor, materialman, or other person or entity, regardless of the number of joint checks issued; (ii) constitute an obligation of the District to any such Subcontractor, materialman, or other person or entity; or (iii) constitute or create any cause of action against the District by the Contractor or any such Subcontractor, materialman, or other person or entity.
- 19.14 District Does Not Waive Rights by Inspecting, Approving or Paying. In no circumstances shall any inspection and/or approval of any portion of the Work, any error or inaccuracy in any estimate or Schedule of Values, any processing of any Progress Payment Request, any issuance of a Certification of Payment, or any payment to the Contractor of any Construction Progress Payment (including, without limitation, the final Construction Progress Payment), be deemed or construed to constitute: (i) a release by the District of the Contractor from any of its obligations under the Contract; (ii) a representation by the District that any information submitted by the Contractor to substantiate any payment amount is accurate or was verified; (iii) a representation by the District that payments to Subcontractors, materialmen, and

others as required were verified or confirmed; (iv) a representation by the District that the Work complies in all respects with the requirements of the Contract Documents; (v) a representation by the District that the means, methods, techniques, sequences, or procedures used in performing the Work were reviewed and/or approved; (vi) an approval by the District of any means, methods, techniques, sequences or procedures used in the Work; (vii) an approval by the District of the use of any invention, appliance, process, article, device, or material, in violation of any royalty, patent or other rights of any person or entity; (viii) an acceptance by the District of any Work that does not conform to the requirements of the Contract Documents; (ix) a waiver by the District of any right(s) it has to enforce the Contractor's obligations under the Contract, whether or not prior to final payment under Section 19.12 of these General Provisions; or (x) a release by the District of the Contractor or its surety or sureties from responsibility for damages arising from the Work.

19.15 Stop Notices and Liens. The Contractor has sole responsibility and liability for ensuring that no person or entity files or otherwise imposes or causes to be imposed any stop notice or lien on or in relation to the Project Site or any portion of the Work or the Project on account of any labor, services, materials, equipment or other thing that the Contractor furnishes or causes to be furnished in connection with the Work. If any such stop notice or lien is filed or otherwise imposed, and if the stop notice or lien has merit or is valid, the Contractor shall be responsible and liable for all costs and expenses incurred by the District in connection with the stop notice or lien, excluding attorneys' fees. Any such costs and expenses shall be charged to the Contractor and/or deducted from amounts otherwise payable to the Contractor under the Contract.

PART 20 TERMINATION AND OTHER REMEDIES

20.1 Termination of Right to Perform Work for Cause.

20.1.1 Cause for Termination. The Contractor shall be in default of its obligations under the Contract, and the District may terminate the Contractor's right to perform the Work for cause, if:

- (i) the Contractor or any Subcontractor refuses or fails to accomplish the Work or any portion thereof in accordance with the Contract Documents or with such diligence as will ensure substantial completion within the Contract Time;
- (ii) the Work, or any portion or phase of the Work, is not, or reasonably will not be, Substantially Completed within the Contract Time;
- (iii) the Contractor or any Subcontractor unreasonably, persistently or repeatedly refuses or fails to supply enough properly skilled workers and/or proper materials to the Project;
- (iv) the Contractor or any necessary Subcontractor is unreasonably, persistently or repeatedly absent from the Project Site;
- (v) the Contractor fails to timely and fully pay any Subcontractors, materialmen, or other individuals or entities the funds to which they are entitled in connection with the Work;
- (vi) the Contractor or any Subcontractor, materialman or other individual or entity that furnishes labor, materials, services or other things in connection with the Work disregards or otherwise fails to comply with any applicable Legal Requirement;
- (vii) the Contractor becomes the subject of voluntary or involuntary bankruptcy proceedings and/or a court determines that the Contractor is bankrupt or otherwise unable to timely pay its bills and/or comply with its contractual obligations;
- (viii) the Contractor makes a general assignment for the benefit of its creditors, or a receiver is appointed on account of the Contractor's insolvency; or
- (ix) the Contractor or any Subcontractor fails to perform any of its obligations as and when required by the Contract, or otherwise violates any of the material provisions of the Contract.
- **20.1.2 Notice of Intent to Terminate for Cause.** In any situation described in Subsection 20.1.1 of these General Provisions, the District may provide written notice to the Contractor and its surety of the District's intention to terminate the Contractor's right to perform the Work ("Notice of Intent to Terminate for Cause"), stating in reasonable detail the reasons for the termination. The Contractor's right to perform the Work shall terminate ten days after receipt of a Notice of Intent to Terminate for Cause unless the reasons for termination have been resolved, corrected or cured, or arrangements satisfactory to the District have been made for correction or cure. The Contractor and its surety shall remain liable in accordance with the Contract Documents for all Work performed prior to such termination.
- **20.2** Surety and District Rights to Perform Work After Termination for Cause. In the event the District issues a Notice of Intent to Terminate for Cause, the Contractor's surety shall have the right to

take over and perform the Contract. If the surety does not (i) give the District written notice of surety's intention to take over and commence performance of the Contract within ten days of receipt of the Notice of Intent to Terminate for Cause and (ii) commence performance of the Contract within twenty days after receipt of the Notice of Intent to Terminate for Cause, then the District may elect to take over and proceed to complete the Work, by separate contract or by any other means or method the District deems advisable, for the account of and at the expense of the Contractor and/or the surety.

Notice of Intent to Terminate for Cause shall the Contractor remove from the Project Site, or suffer or permit the removal from the Project Site of, any tools, equipment, vehicles, materials, supplies, appliances, plants, or other items owned, leased or rented by the Contractor and used or employed in connection with the Work. In the event the District elects to take over the Work as specified in Section 20.2 of these General Provisions, then: (i) the District may, without liability for so doing, take possession of and use in completing the Work all such tools, equipment, vehicles, materials, supplies, appliances, plants, or other items on or at the Project Site used or employed in connection with the Work and owned, leased or rented by the Contractor; (ii) at the District's election, any or all of the Subcontracts shall be deemed to have been assigned to the District; and (iii) the Contractor and its surety shall be liable to the District for any costs or other damages incurred by the District attributable to or arising from the District taking over the Work as provided in this Section. THE CONTRACTOR MUST ENSURE THAT EACH OF ITS CONTRACTS WITH ITS SUBCONTRACTORS AND SURETIES INCLUDES SUCH PROVISIONS AS ARE NECESSARY TO EFFECTUATE THE REQUIREMENTS OF THIS SECTION 20.3, HOWEVER, NO FAILURE TO DO SO WILL INVALIDATE SUCH REQUIREMENTS.

20.4 Contractor Compensation After Termination for Cause. In the event the Contractor's right to perform the Work is terminated under Section 20.1 of these General Provisions, the Contract shall remain in effect. However, in such event, the Contractor shall not be entitled to receive any further payment under the Contract: (i) until the Project has been fully completed and accepted in accordance with Section 16.8 of these General Provisions; and (ii) only if the cost to the District to complete the Work is less than the unpaid portion of the GMP (including, among other costs, the Leaseback Payments, but not any interest thereon) as adjusted in accordance with the Contract. If the cost incurred by the District to complete the Work exceeds such unpaid portion of the GMP, the Contractor (or its surety) shall pay the difference to the District within thirty days of receiving an invoice for such amount from the District.

20.5 Termination of Right to Perform Work for Convenience. The District, for any reason and without need for cause, may terminate the Work in whole or, from time to time, in part, regardless of the circumstances resulting in termination. In any such event, the District shall provide written notice to the Contractor of the termination ("Notice of Termination for Convenience"). The Contractor shall remain liable in accordance with the Contract Documents for all Work performed prior to any such termination, and the Contractor's surety shall not be deemed or construed to have been relieved of its obligations in regard to any just claims arising out of or relating to such Work.

20.6 Contractor Must Cease Work Upon Termination for Convenience. Upon receipt of a Notice of Termination for Convenience, the Contractor shall immediately proceed to: (i) stop all Work to the extent specified in the Notice of Termination for Convenience; (ii) as specified in the Notice of Termination for Convenience, complete any Work that must be finished off prior to termination in a least-cost/shortest-time manner while still maintaining the quality required under the Contract Documents; (iii) leave the terminated portion of the Work (to the extent already completed) in a safe and clean condition so that it will not pose any threat to the health or safety of any persons; (iv) terminate all Subcontracts to

the extent those provide for the terminated portions of the Work, except such Subcontracts as specified in the Notice of Termination for Convenience that shall be deemed to be assigned to the District; and (v) place no further orders and make no further contracts with Subcontractors, except as necessary to complete portions of the Work not terminated.

20.7 Documenting Costs After Termination for Convenience. Within fourteen days after the effective date of a termination for convenience, the Contractor must submit to the District all documentation required under Section 19.6 of these General Provisions to substantiate all costs incurred by the Contractor for labor, materials and equipment through the effective date of the termination. In addition, within twenty-one days after the effective date of any such termination, the Contractor must submit to the District such documentation as reasonably details and substantiates costs reasonably incurred by the Contractor solely as a result of the termination for convenience. All of the foregoing documentation must: (i) describe the costs incurred with particularity; and (ii) be conspicuously identified as "Termination Costs Resulting from Termination for District Convenience."

20.8 Compensation After Termination for Convenience. In the event of a termination for convenience, subject to receipt of reasonable substantiating documentation as described in Section 20.7 of these General Provisions, the District shall pay to the Contractor an amount equal to the total of: (i) the actual costs incurred by the Contractor in accordance with the provisions of the Contract Documents attributable to the terminated portion of the Work that was satisfactorily completed by the Contractor, but not previously paid by the District ("Actual Costs"); (ii) a reasonable allowance for profit on the Actual Costs, provided that the Contractor establishes to the satisfaction of the District that it is reasonably probable the Contractor would have made a profit on such portion of the Work, but in no event shall such allowance exceed ten percent of the Actual Costs; and (iii) a reasonable allowance for administrative and demobilization costs incurred by the Contractor, but in no event shall such allowance exceed five percent of the Actual Costs. Notwithstanding anything to the contrary, in no event shall the total amount payable to the Contractor under this Section exceed the portion of the GMP that is proportionate to the portion of the Work satisfactorily completed prior to termination, consistent with the then-most recent approved Schedule of Values.

20.9 Termination by the Contractor for Cause. The Contractor may terminate the Contract, by providing written notice to the District, in the event: (i) as a result of factors beyond the Contractor's control and not arising from any fault, omission, act or negligence of the Contractor or any Subcontractor, materialman or other person or entity that has furnished or is to furnish any labor, materials, services or other things in connection with the Work, there has been a cessation or suspension of all Work for a period of more than one-hundred and twenty consecutive days, and the District has not provided to the Contractor within such period a notice to resume the Work, a Notice of Intent to Terminate for Cause, or a Notice of Termination for Convenience; or (ii) after notice and reasonable opportunity to cure, the District fails to pay the Contractor any substantial undisputed sums due to the Contractor in accordance with the Contract Documents. In the event of a termination by the Contractor under this Section, and subject to Section 20.10 of these General Provisions, the Contractor shall be entitled to compensation as provided in Section 20.8 of these General Provisions for a termination for convenience, including, without limitation, compensation not in excess of the proportionate amount of the GMP plus any accrued interest on unpaid, but due and payable, Leaseback Payments. In the event of termination by the Contractor under this Section, the Contractor shall have no Claim(s) against the District except with respect to Work that the Contractor performed prior to termination. The Contractor and its surety shall remain liable in accordance with the Contract Documents for all Work performed prior to such termination.

20.10 Documenting Costs After Contractor Termination for Cause. Within fourteen days after the effective date of a termination by the Contractor for cause under Section 20.9 of these General Provisions, the Contractor must submit to the District all documentation required under Section 19.6 of these General Provisions to substantiate all costs incurred by the Contractor for labor, materials and equipment through the effective date of the termination. All of the foregoing documentation must: (i) describe the costs incurred with particularity; and (ii) be conspicuously identified as "Termination Costs Resulting from Contractor Termination for Cause."

Remedies for Default Other Than Termination. In the event the Contractor is in default of its obligations under the Contract in any one or more of the ways specified in Subsection 20.1.1 of these General Provisions, then, in its sole discretion and after notice as provided in this Section, the District shall have the right to cure or otherwise correct the default(s), by any reasonable means or method the District deems advisable, without terminating the Contract or the Contractor's right to perform the other portions of the Work. If the District so intends to cure or otherwise correct any default(s) by the Contractor, the District shall provide written notice to the Contractor and its surety. If, within five days of receiving such notice (or, in the event of an Emergency or situation involving an existing or potential safety hazard, within such shorter period of time, if any, as set forth in the notice), the Contractor fails to adequately cure or otherwise correct the default(s), or fails to make arrangements satisfactory to the District for such adequate cure or other correction, the District shall have the right to cure or otherwise correct the default(s), by separate contract or by any other reasonable means or method the District deems advisable (including, without limitation, supplementing the workforce of the Contractor or any Subcontractor with additional workers and/or equipment), for the account of and at the expense of the Contractor. If the District is required to expend funds in connection with any such cure or other correction, such amounts shall accrue interest at the maximum legal rate from the date(s) expended to the date(s) the District is reimbursed, whether reimbursed directly by the Contractor or its surety. Such amounts shall be charged to the Contractor and/or deducted from amounts payable to the Contractor under the Contract. The rights of the District under this Section are in addition to, not in lieu of, any other rights and remedies the District may have under law, equity or contract, and in no circumstances shall the rights of the District under this Section be deemed or construed to limit or constitute a waiver by the District of any such rights or remedies. Neither the Contractor nor its surety shall have any recourse against the District with respect to any cure(s) and/or other correction(s) undertaken by the District in good faith.

20.12 Effect on other Projects with District. Contractor may be under contract with the District with regard to other projects that are being constructed simultaneously, or in a series, with this Work. In such an event, and without limitation or penalty to the District, any exercise by the District of the termination rights, or other rights, under this Part 20, or any other clause or provision in the LLB Agreements, may, at the District's sole written discretion, apply to any or all of the other projects or portions thereof. For example, if the District awards two simultaneous/serial projects to the Contractor, but the District is dissatisfied with the performance on any project or portion thereof, the District shall be able to terminate any or all of the other projects, or portion thereof, in addition to the first project, without penalty or any additional compensation.

PART 21 INDEMNIFICATION OF DISTRICT AND OTHERS

- 21.1 Indemnification of District and its Representatives. Subject to the provisions of Section 21.4 of these General Provisions, the Contractor shall indemnify, defend, and hold-harmless the District, the District's project manager for the Project, if any, and the Inspector of Record (each, including the District, a "District Indemnitee"), and each of them, from and against any and all claims, demands, causes of action, actions, other proceedings, liens, judgments, damages, losses, costs, expenses (including attorneys' fees), and other liabilities of any nature arising from, or directly or indirectly related to, the performance of the Work or Project and/or the Contractor's or any Subcontractor's acts and/or omissions directly or indirectly relating to the Contract Documents, including, without limitation, liabilities arising from: (i) personal injury (including death) or property damage; (ii) any act or omission by the Contractor, any Subcontractor, or any other person or entity that furnishes any labor, materials, services, goods or other things in connection with the Work; and (iii) infringement or alleged infringement of any patented or unpatented invention, process, article, or appliance manufactured or used in the performance of the Contract, including its use by the District, unless specifically stated otherwise in the Contract Documents. The obligations of the Contractor under this Section are in addition to, and not in lieu of, any other obligations the Contractor has under the Contract Documents and/or applicable Legal Requirements.
- Contractor Defense of District and its Representatives. Without prior demand from the District, the Contractor, at its sole cost and expense, shall defend each and every claim, demand, action, and other proceeding within the scope of Section 21.1 of these General Provisions as to which any District Indemnitee is a party, regardless of whether the District Indemnitee may be the sole party named in the claim, demand, action or other proceeding. Any such defense must be conducted by knowledgeable and experienced legal counsel selected and retained by the Contractor at its cost, but reasonably acceptable to the District. Without limiting anything else in any indemnity provisions of the Contract, the Contractor shall also pay the reasonable costs incurred by the District in connection with the monitoring of, and, if necessary, the participation in, the defense of any District Indemnitee by the District's legal counsel. Without jeopardizing or compromising any of its other rights, the District may settle any claim, demand, action or other legal proceeding on terms determined by the District Board to be reasonable and in the best interests of the District and/or any other District Indemnitee. As part of its obligations under this Section and Section 21.1 of these General Provisions, and with respect to any claim, demand, action or other proceeding within the scope of such Sections, within thirty days of receiving an invoice from the District, the Contractor shall reimburse the District for any and all: (i) judgments paid by the District; (ii) amounts paid by the District in settling such claim, demand, action or other proceeding; and (iii) any other legal or other costs and expenses reasonably incurred by the District in connection with such claim, demand, action or other proceeding. If the Contractor fails to pay any such amount(s) within the required time, the unpaid amount(s) shall accrue interest at the legal rate.
- **21.3** Indemnification of District Officers, Employees, and Agents. For purposes of each and every obligation of the Contractor set forth in these General Provisions and other Contract Documents to indemnify, defend and/or hold-harmless the District or the District Indemnitees, the reference to the District or District Indemnitees shall be deemed and construed to be a reference also to the District Board and each member thereof, the District's other officers, employees, and agents, and each of them.
- **21.4 Limitation on Contractor Indemnification Obligations.** Notwithstanding anything to the contrary, the Contractor shall not be responsible or liable under Sections 21.1, 21.2 or 21.3 of these General Provisions, or any other indemnification provisions set forth in the Contract Documents, to the

extent that a claim, demand, action, other proceeding, lien, judgment, damage, loss, cost, expense, or other liability is attributable to the active negligence, sole negligence, or willful misconduct of the District or any other District Indemnitee. Nothing in these General Provisions or the other Contract Documents shall be deemed or construed to require that the Contractor indemnify, defend or hold-harmless the Architect.

- **21.5** Contractor Must Ensure Subcontractors Indemnify District Indemnitees. The Contractor must include provisions in each Subcontract requiring that the Subcontractor also indemnify, defend, and hold-harmless the District and the other District Indemnitees to the same extent required of the Contractor under this Part 21 of these General Provisions.
- **21.6** Insurance Not a Limitation. The Contractor's obligations under this Part 21 shall not be deemed to be: (i) conditioned upon, or in any manner limited by, the any insurance coverage required by the LLB Agreements; or (ii) conditioned upon the receipt by any person or entity of, or limited to the amount of, any insurance proceeds.
- **21.7 Attorney's Fees.** Notwithstanding anything to the contrary, except the right to be indemnified regarding attorney's fees, no person, entity, or party, including but not limited to the District and the Contractor, shall be allowed to recover attorney's fees that are incurred in an action to either enforce or defend the LLB Agreements.

PART 22 CLAIMS AND LEGAL PROCEEDINGS

- 22.1 Requirements and Procedures for Filing Claims Are Mandatory. This Part 22 establishes mandatory requirements and procedures applicable to the filing by the Contractor of any and each demand for: (i) extension of time; (ii) payment of money or damages arising from Work done by or on behalf of the Contractor under the Contract and payment of which is not otherwise expressly provided for or to which the claimant is not otherwise entitled; or (iii) an amount the payment of which is disputed by the District (each of the foregoing a "Claim"). IF THE CONTRACTOR FAILS TO FILE ANY CLAIM IN STRICT ACCORDANCE WITH CERTAIN REQUIREMENTS AND PROCEDURES DESCRIBED IN THIS PART 22, THE CONTRACTOR SHALL BE DEEMED AND CONSTRUED TO HAVE FORFEITED AND WAIVED ANY AND ALL RIGHTS TO ASSERT THE CLAIM ON ANY BASIS OR TO INITIATE AND PURSUE ANY LEGAL ACTION OR OTHER PROCEEDING BASED ON ANY FACTS AND/OR CIRCUMSTANCES FORMING A BASIS FOR THE CLAIM.
- 22.2 Mandatory Time Limits for Filing of Claims. The Contractor must file each Claim within fourteen days of the date the Contractor first becomes aware or reasonably should have first become aware of any basis for the Claim. Additionally, the Contractor must file any and all Claims prior to when the District issues final payment to the Contractor under Section 19.12 of these General Provisions, and the District shall reject any Claim filed thereafter as null and void. IF THE CONTRACTOR FAILS TO FILE A CLAIM WITHIN THE PERMITTED FOURTEEN-DAY PERIOD AND PRIOR TO FINAL PAYMENT, THE CONTRACTOR SHALL BE DEEMED AND CONSTRUED TO HAVE FORFEITED AND WAIVED ANY AND ALL RIGHTS TO ASSERT THE CLAIM ON ANY BASIS OR TO INITIATE AND PURSUE ANY LEGAL ACTION OR OTHER PROCEEDING BASED ON ANY FACTS AND/OR CIRCUMSTANCES FORMING A BASIS FOR THE CLAIM.
- 22.3 Content of Claims and Substantiating Materials. Each Claim must consist of: (i) a cover letter that sets forth a summary of the basis or bases for the Claim (including, without limitation, dates of relevant occurrences, the particular persons involved or that have relevant knowledge, the specific remedy and/or compensation the Contractor is seeking, and, if applicable, the total amount of the Claim and a breakdown of the total amount into general categories of costs incurred by the Contractor); (ii) a reasonably detailed analysis of the contractual bases for the Claim (including, without limitation, identifying all provisions of the Contract Documents relevant to the Claim), the legal bases for the Claim, and any other bases or justifications for the Claim asserted by the Contractor, with cross-references to documents submitted in support of the Claim; and (iii) all documents that support the Contractor's position(s) as described in the Claim, including, by way of example and not as a limitation, any Specifications, Drawings, cost analyses, daily reports, et cetera. A Claim must include all information that the Contractor desires be considered in connection with the review, analysis, and rejection or approval of the Claim.
- 22.4 Mandatory Certification of Claims Subject to Penalty of Perjury. The Contractor must submit each Claim with a written certification by the Contractor stating that: (i) the Contractor has reviewed the Claim and is filing it in a good-faith belief that the Contractor is entitled to the remedy and/or compensation described in the Claim; (ii) each document and item of other supporting information (whether an original or copy) submitted with the Claim is authentic (i.e., not altered or modified in any manner), accurate and complete; (iii) the Claim accurately sets forth, the total amount of the District's monetary and/or other liability for the Claim; and (iv) the Contractor acknowledges that the filing of false and/or fraudulent claims may result in fines and/or imprisonment under Government Code Sections

12650 et seq. and Penal Code Section 72. The duly-authorized representative of the Contractor must sign each such certification under penalty of perjury and the signature must be notarized. IF THE CONTRACTOR FAILS TO SUBMIT THE REQUIRED CERTIFICATION WITH ANY CLAIM, THE CONTRACTOR SHALL BE DEEMED AND CONSTRUED TO HAVE FORFEITED AND WAIVED ANY AND ALL RIGHTS TO ASSERT THE CLAIM ON ANY BASIS OR TO INITIATE AND PURSUE ANY LEGAL ACTION OR OTHER PROCEEDING BASED ON ANY FACTS AND/OR CIRCUMSTANCES FORMING A BASIS FOR THE CLAIM.

- **22.5 Prerequisites for Filing Delay Claims.** NO CLAIM FILED BY THE CONTRACTOR THAT DEMANDS AN EXTENSION OF THE CONTRACT TIME AND/OR AN INCREASE IN THE GMP BASED ON A DELAY IN THE WORK SHALL BE VALID, AND THE DISTRICT SHALL NOT BE REQUIRED TO CONSIDER ANY SUCH CLAIM, UNLESS THE CONTRACTOR DEMONSTRATES PRIOR COMPLIANCE WITH ALL APPLICABLE REQUIREMENTS OF SECTIONS 11.9 AND 11.10 OF THESE GENERAL PROVISIONS. If the District disapproves a Change Order Request submitted in accordance with Section 11.10 of these General Provisions, the period in which the Contractor may file a related Claim shall commence upon receipt by the Contractor of notice of such disapproval.
- **22.6 Method of Filing Claims.** The Contractor must file each complete Claim (i.e., each Claim that satisfies the requirements of Sections 22.1 through 22.5, inclusive, of these General Provisions) by delivering it to the District, with copies to the Architect and Inspector of Record, via personal delivery (signature of receiving person requested) or certified or registered U.S. Mail (postage pre-paid and signature of receiving person requested).
- PCC Claims Procedures. The provisions of Public Contract Code Section 9204, and, to the extent applicable, Public Contract Code Section 20104 et seq. ("PCC Claims Procedures"), as those may be amended from time to time, shall apply to any and each Claim. The PCC Claims Procedures are incorporated herein by this reference. In summary, the PCC Claims Procedures specify requirements and procedures for filing a claim, for responding to a claim, and for disputing the response to a claim. The PCC Claims Procedures require that the claimant provide such documentation as reasonably supports the claim and that, the public agency provide to the claimant a written statement identifying the disputed and undisputed portions of the claim. The public agency must provide such response to the claimant within forty-five days after receiving the claim or, if approval of the response by the governing body of the public agency is required, then not later than three days following the next duly publicly noticed meeting of the governing body after such forty-five-day period. The PCC Claims Procedures specify additional requirements if the public agency does not timely respond, or if the claimant disputes the response. In addition, the PCC Claims Procedures specify requirements for civil actions filed to resolve claims. The PCC Claims Procedures do not apply to tort claims or alter time periods for filing of tort claims in accordance with the Government Code. For additional information, the Contractor should refer to Public Contract Code Section 9204 and 20104 et seq. NOTHING IN THIS SECTION 22.7 OR IN THE PCC CLAIMS PROCEDURES SHALL BE DEEMED OR CONSTRUED TO SUPERSEDE THE REQUIREMENTS OF SECTIONS 22.1 THROUGH 22.6, INCLUSIVE, OF THESE GENERAL PROVISIONS, WHICH ARE APPLICABLE TO EACH AND EVERY CLAIM ARISING FROM THE WORK OR THE CONTRACT.
- **22.8 District Requests for Additional Information.** Within 20 days of receipt of a Claim, the District and/or Architect may request (including for purposes of the PCC Claims Procedures when those are applicable) that the Contractor provide additional information relevant to the Claim that is believed reasonably necessary or convenient for analysis or evaluation of the Claim. The Contractor shall provide such additional information as soon as reasonably possible, but in no event later than five business days after receiving the District's request.

- 22.9 Procedures for Initial Review of Claim by Architect. The Architect, in consultation with the District and the Inspector of Record, will review each Claim and, within twenty days after receipt of the Claim, shall take one or more of the following preliminary actions: (i) request that the Contractor provide additional information pursuant to Section 22.8 of these General Provisions; (ii) if necessary due to the complexity and/or number of issues described in the Claim, provide information to the Contractor as to when the Architect anticipates being able to complete its review pursuant to this Section 22.9; (iii) recommend that the District disapprove the Claim in whole or in part, stating reasons for rejection; (iv) recommend that the District approve the Claim; or (v) suggest a compromise of the Claim. If the Architect anticipates that the complete review of the Claim cannot be completed within forty-five days following receipt by the District of the Claim, the District shall attempt to reach an agreement with the Contractor for an extension of the time to review the Claim. If the Contractor refuses to agree to such an extension, and the District does not otherwise respond to the Claim within forty-five days following receipt of the Claim, the Claim shall be deemed rejected. Regardless of whether a Claim is processed pursuant to the PCC Claims Procedures, the District may, but is not obligated to, notify the Contractor's surety of the nature and amount of the Claim and request the surety's assistance in resolving the Claim.
- **22.10 District Response to Claim Following Initial Architect Review.** The District's response to each Claim shall be subject to approval by the District Board. In each case, following the initial review of a Claim by the Architect, the District will provide to the Contractor within the time required by the PCC Claims Procedures, a written statement identifying the portions of the Claim that are undisputed and that are disputed. If the District determines that some or all of a Claim is undisputed, the District will prepare or obtain and process a Change Order that appropriately documents the resolution of the Claim to the extent that the Claim is undisputed. Payment of any undisputed portion of the Claim shall be made in accordance with the PCC Claims Procedures.
- **22.11** Initiation of Informal Claim-Resolution Efforts. If the Contractor disputes the District's written statement or if the District fails to respond to a Claim within the time required by the PCC Claims Procedures, the Contractor may demand that the District meet and informally confer with the Contractor in an attempt to settle the issues remaining in dispute. Each such demand must be in writing and sent to the District by registered mail or certified mail, return receipt requested. In such event, the District shall schedule a mandatory meeting of the Contractor and the District, Architect, and Inspector of Record, to occur within thirty days following receipt by the District of the Contractor's written demand. The District and the Contractor may bring to such meeting any documents or other materials related to the basis or bases for the Claim and any individual(s) they believe necessary or convenient for purposes of such informal efforts. The individuals present at the meeting shall make good-faith efforts to resolve the Claim. If the District and Prime Contractor are unable to resolve the Claim during the meeting, but agree that further informal efforts would be productive, they may schedule additional meetings or discussions for purposes of continuing the efforts to resolve the Claim.
- **22.12 Documentation of Compromise.** If, following the meet and confer conference, the District and the Contractor agree to a compromise or other resolution of some or all of the disputed portion of a Claim, the Architect will prepare or obtain and process a Change Order, or the District and Contractor shall enter into another appropriate written agreement, that appropriately documents such agreement. The District, if applicable, shall pay any undisputed portion of a Claim in accordance with the PCC Claims Procedures.

22.13 Architect Ruling if Claim Remains Unresolved. If, following the meet and confer

conference, the District and the Contractor do not agree to a compromise or other resolution of the disputed portion of a Claim, the District will issue a written statement to the Contractor identifying the portions of the Claim that remain in dispute and that are undisputed. The District will issue its ruling within ten business days of the conclusion of the informal meet and confer conference. Subject to the provisions of Section 22.14 of these General Provisions, if the Contractor disagrees with the District's statement, as described in writing to the District, the portion of the Claim that remains in dispute shall be submitted to nonbinding mediation in accordance with the PCC Claims Procedures.

- **22.14 Conditions Precedent to Initiating Subsequent Actions.** With respect to each Claim, the compliance by the Contractor with the requirements of Sections 24.1 to 24.11 of these General Provisions shall be a condition precedent to initiation by the Contractor of any legal or equitable action, arbitration or other proceeding relating to or arising from the matters addressed in the Claim.
- 22.15 Contractor Must Continue Work While Claims Pending. Neither the existence of any dispute, nor the filing or other initiation of any Claim or related action, arbitration or other legal proceeding, shall be deemed or construed to constitute a valid basis for the Contractor to stop, delay or change the Work. The Contractor must diligently continue with all Work as required by the Contract Documents (including, without limitation, as set forth in any Architect Field Directives) and in accordance with all milestones set forth in the Master Construction Schedule, regardless of whether: (i) any dispute exists or any Claim, action, arbitration or other legal proceeding has been filed or otherwise initiated; or (ii) the District disapproves any Claim. In the event a Claim is not resolved to the Contractor's satisfaction, the Contractor's sole remedy shall be to initiate an action or other legal proceedings as permitted pursuant to the Contract, but only, as described in Section 22.17 of these General Provisions, after the Project has been completed or the Contract has been terminated.
- 22.16 Resolving Disputes through Binding Arbitration. In the event a dispute remains following compliance with PCC Claims Procedures, the District and the Contractor may agree in writing to resolve any Claim through binding arbitration. In such event, the District and Contractor shall attempt within thirty days thereafter to agree on the arbitrator who will conduct the arbitration. If the parties cannot so agree, they shall request that a judge of the Superior Court for the County designate an arbitrator with experience in public works construction. The District and the Contractor each shall pay one-half of the cost of the arbitration. The arbitrator shall establish procedures and rules to be followed in conducting the arbitration, which, at a minimum, shall specify that the arbitrator must adhere to and apply all substantive statutory, regulatory, administrative and decisional law that is applicable to the dispute. If a party petitions to confirm, correct, or vacate the award as provided by Chapter 4 of Title 9 of the Code of Civil Procedure (commencing with Section 1285), the prevailing party shall not be entitled to reasonable attorneys' fees and expenses incurred in connection with such proceedings. The surety that issued the Performance Bond and/or Payment Bond shall be made a party to any such arbitration and shall be fully bound by any decision of the arbitrator.
- **22.17 Resolving Disputes in Court of Competent Jurisdiction.** If any Claim is not resolved in accordance with the procedures set forth in this Part 22 after compliance by the Contractor with all such procedures as required, then, except if and as limited by Public Contract Code Section 20104.4 when it applies the Contractor may file an action in a court of competent jurisdiction in the County seeking resolution of the Claim through bench trial. No such action may be initiated until after: (i) the entirety of the Work has been fully completed and accepted by the District in accordance with Section 18.9 of these General Provisions; (ii) completion of the Project has occurred as defined in Subsection (c) of Public Contract Code Section 7107; or (iii) the Contract in its entirety has been terminated prior to completion

of the Project. AS A CONDITION PRECEDENT TO THE CONTRACTOR'S RIGHT TO FILE ANY SUCH ACTION, WITHIN FIFTEEN DAYS AFTER COMPLETION OF THE PROCEDURES REQUIRED PURSUANT TO THIS PART 22 WITHOUT RESOLUTION OF THE APPLICABLE CLAIM, THE CONTRACTOR MUST PROVIDE WRITTEN NOTICE TO THE DISTRICT THAT THE CONTRACTOR THEREBY RESERVES ITS RIGHTS TO FILE SUCH ACTION. Any statutory limitation on the period for filing of such an action shall be tolled, from the date the procedures required pursuant to this Part 22 are duly completed without resolution of the applicable Claim, until the date an action may be filed in accordance with the foregoing provisions of this Section. If a Claim for \$375,000 or less remains in dispute following the procedures set forth in this Part 22, the applicable provisions of Public Contract Code Section 20104.4 shall apply.

22.18 Neither Party's Remedies are Limited. Except as expressly stated in these General Provisions or the other Contract Documents, the rights and remedies available in accordance with the Contract are in addition to any rights and remedies available pursuant to applicable law; provided that the exercise of any and all such rights and remedies are subject to procedural requirements made applicable by this Part 22.

22.19 Applicable Law and Venue. The Contract shall be construed in accordance with the laws of the State. Any provision of law purported to be described or specified herein that is incorrectly described or specified shall, nonetheless, be applicable as if correctly described or specified herein. If any action, arbitration, mediation or other proceeding is initiated to enforce or interpret any term of the Contract, such proceeding shall be initiated and conducted only in the County.

PART 23 MISCELLANEOUS PROVISIONS

- **23.1** Entire Understanding and Agreement. The Contract, as defined in these General Provisions (including without limitation, the Contract Documents LLB Agreements) and as amended in accordance with the Contract Documents, constitutes the final, complete and exclusive statement of the terms of the agreement between the District and the Contractor pertaining to performance of the Work. The Contract supersedes all prior and contemporaneous understandings or agreements of the Parties, oral or written, except as those are included in the Contract. Each of the District and the Contractor acknowledges and agrees that neither the other Party, nor its agents or attorneys, has induced the execution of the Contract by making any promise, representation, or warranty whatsoever, express or implied, not set forth in the Contract.
- **23.2** All Legal Requirements Deemed Included. Each and every provision required by any applicable Legal Requirement to be included in the Contract is hereby deemed to be so included, and the Contract shall be construed and enforced as if all such provisions are so included. If, for any reason, any provision required by any Legal Requirement is not expressly included or incorporated into the Contract Documents, or is not correctly included or incorporated into the Contract Documents, then, upon request of either the District or the Contractor, they shall amend the Contract Documents to include or incorporate, or to correctly include or incorporate, such provision.
- **23.3 Execution of Documents in Counterparts.** The District and Contractor may sign the Construction Services Agreement and other Contract Documents in one or more counterparts, which, taken together, shall be deemed and construed to constitute one and the same original instrument. Signature pages may be detached from counterpart originals and combined to physically form one or more copies of the Construction Services Agreement having original signatures of both Parties.
- **23.4 Captions and Headings.** Any and all captions or headings for the Parts, Sections, Subsections and other provisions of these General Provisions and the other Contract Documents are provided solely for convenience of the reader and shall not be deemed or construed to establish, define, or limit the content or meaning of such provisions.
- 23.5 No Third-Party Beneficiaries of Contract. Except to the extent provided by any applicable Legal Requirement (e.g., requirements for payment of prevailing wages to workers on the Project), no party other than the District or the Contractor may claim or assert any right or benefit arising from the Contract. Each provision of the Contract Documents shall be deemed and construed to benefit only the District and/or the Contractor unless and only to the extent the provision is included in the Contract specifically as a result of any Legal Requirement intended to benefit any third party. Provisions of the Contract Documents that relate to or permissibly expand on any Legal Requirement intended to benefit any third party, but that are not necessary for compliance with that Legal Requirement, shall be deemed and construed as being included in the Contract for the convenience of the Parties and shall not be deemed or construed to benefit any third party or as providing a basis for any claim, demand, action or other proceeding by a third party relating to the Contract.
- **23.6** Circumscribed Right to Assign Contract. The Contractor may not, without the prior written consent of the District: (i) assign, transfer, pledge, or hypothecate the Contract or any interest therein; or (ii) sublet or lend the use of the Project Site or any portion thereof. Any consent by the District to any of the foregoing prohibited acts shall be deemed and construed to apply only in the particular circumstances

for which consent is given and shall not be deemed or construed as consent for any subsequent similar or other circumstances. Except as expressly permitted by the Contract Documents, the Contractor shall not assign the Contract or any obligations under the Contract without the prior written consent of the District, which consent the District may grant or deny in its sole discretion. Any attempted or purported assignment by the Contractor of the Contract or any obligations under the Contract, without such prior written consent, shall be null and void, and the Contractor shall remain responsible for all of its obligations under the Contract. Subject to the foregoing, the Contract shall inure to the benefit of, and be binding on, the authorized successors and assigns of the Contractor.

- any time(s) to require performance of any requirement of the Contract shall in no manner affect the right at a later time to enforce the same or any other provision of the Contract. Except as expressly provided in the Contract Documents, the forbearance or indulgence by either the District or the Contractor in any regard whatsoever shall not constitute a waiver of the requirement at issue. Until complete performance by a Party of the requirement at issue, the District or Contractor, as applicable, shall be entitled to invoke any remedy available to it in accordance with the Contract Documents, regardless of any such forbearance or indulgence. No requirement of the Contract to be performed by or on behalf of either the District or the Contractor can be waived except by the written consent of the other. Unless expressly provided in any such written waiver, no waiver by either the District or the Contractor of a breach or the performance of any requirement of the Contract shall be deemed to be construed as a continuing waiver of future breaches or future performance of the same or any other requirement.
- **23.8** Requirements of Contract are Severable. If a court of competent jurisdiction determines, for any reason, that any provision or requirement of the Contract (including, without limitation, any provision of any of the LLB Agreements) is invalid or unenforceable, such determination shall not invalidate or render unenforceable any other provision or requirement of the Contract. In such event, the remaining provisions and requirements shall be interpreted, to the extent permitted by law, in a manner that is consistent with the intent and purpose of the invalid or unenforceable provision or requirement.
- **23.9 Assignment of Anti-Trust Claims.** The Contractor hereby assigns to the District all rights, title and interest in and to all causes of action the Contractor may have under Section 4 of the Clayton Act (15 USC Sec. 15) or under the Cartwright Act (Chapter 2 (commencing with Section 16700) of Part 2 of Division 7 of the Business and Professions Code), arising from purchase of goods, services, or materials under the Contract. This assignment shall become effective at the time the District tenders the final Construction Progress Payment to the Contractor under Section 19.12 of these General Provisions, without further acknowledgment by either the District or the Contractor.

23.10 Service of Demands and Other Notices.

23.10.1 General Requirements. Unless the Contract Documents expressly provide otherwise with respect to any particular circumstance, demands and other notices required or permitted to be given under the Contract Documents (each a "Notice") must be in writing and must be given or served in accordance with this Section 23.10.

23.10.2 Methods of Delivery. Each Notice must be sent via: (i) personal delivery (with signature of recipient and recipient's name legibly written on delivery receipt); (ii) facsimile transmission (with confirmation of transmission printed from the sender's machine and retained in the sender's records, a copy of such confirmation provided to recipient upon request, and the original of the Notice deposited into the United States mail, first-class postage prepaid, within twelve hours after transmission);

(iii) registered or certified United States mail (postage pre-paid and return receipt requested); (iv) FedEx, U.P.S. or other reliable, private delivery service (with signature of recipient obtained on electronic or other delivery receipt); or (v) electronic mail (email) transmission (with confirmation of transmission printed from the sender's machine and retained in sender's records, and a copy of such confirmation provided to recipient upon request).

23.10.3 Persons to Whom Notices Must be Sent. Notices sent to the District must be addressed and delivered to each of the Authorized District Representatives identified in Exhibit G to the Construction Services Agreement, at the District's main administrative offices, and copies of each such Notice must be sent to the Architect. Notices sent to the Contractor must be addressed and delivered to either the Job Superintendent or any of the Authorized Contractor Representatives. A Party may change its address, facsimile transmission number, or person to whom attention should be directed, by giving Notice in accordance with this Section 23.10.

23.10.4 Giving or Service of Notice. A Notice shall be deemed given or served only upon actual receipt by the addressee. In the case of email, "actual receipt" shall mean delivery to the recipient's email in-box. If a Notice, including, without limitation, any Notice sent by email, is delivered after 4:00 p.m. on any day (whether or not it is a business day), the Notice shall be deemed to have been given or served as of 9:00 a.m. on the next subsequent business day. For purposes of the LLB Agreements, a "business day" is any day that is not any of the following: (i) a Saturday or Sunday; (ii) a federal or State holiday; or (iii) a State- or District-mandated furlough day with respect to the District's administrative personnel. As an additional condition to sending a Notice by email, the reference line must indicate that it is a "Notice Under Project LLB Agreements." Because email addresses are subject to change more frequently than physical addresses, if a Notice is to be sent by email, unless the sender has personal knowledge of the correct, then-current email address of each intended recipient, the sender must call and verify the then-current email address of each intended recipient prior to sending the Notice.

23.10.5 Applicability of Section. The requirements of this Section 23.10 shall not be deemed or construed to apply to: (i) communications of the District and/or the Contractor necessary for day-to-day administration of the Contract or the Work; or (ii) service of process in accordance with any applicable law or rule of court.

23.11 Public Inquiries and Complaints. If the Contractor or any Subcontractor or other person or entity on, at or in the vicinity of the Project Site on account of the Work receives any inquiry, complaint, or other communication regarding the nature, status, et cetera, of the Work or the Project, from any homeowner, business owner, member of the press, or other member of the public, the inquiry, complaint or other communication must be referred to the District for response. The Contractor shall as necessary provide to the District any information in the Contractor's possession or control that is reasonably required for the District to respond to any such inquiry, complaint, or other communication.

23.12 District Notice of Third-Party Claims. In accordance with Public Contract Code Section 9201, the District shall timely notify the Contractor if the District receives any third-party claim relating to the Work or the Contract. The District shall be entitled to recover from the Contractor the District's reasonable costs incurred in providing such notification.

(End of General Provisions.)

EXHIBIT E ITEMIZATION OF GMP

(Exhibit to begin on the next page.)

EXHIBIT F REQUIREMENTS FOR CONTRACTOR INSURANCE

- 1. Insurance Requirements. Before the commencement of the Work, the Contractor shall purchase from and maintain in a company or companies lawfully authorized to do business in California as admitted carriers with a financial rating of at least an A status as rated in the most recent edition of Best's Insurance Reports or as otherwise amended in these Contract Documents, such insurance as will protect the District from claims set forth below, which may arise out of or result from the Contractor's operations under the Contract and for which the Contractor may be legally liable, whether such operations are by the Contractor, by a Subcontractor, by anyone directly or indirectly employed by any of them, or by anyone for whose acts any of them may be liable:
 - a. Claims for damages because of bodily injury, sickness, disease, or death of any person District would require indemnification and coverage for employee claim;
 - b. Claims for damages insured by usual personal injury liability coverage, which are sustained by a person as a result of an offense directly or indirectly related to employment of such person by the Contractor or by another person;
 - Claims for damages because of injury or destruction of tangible property, including loss of use resulting therefrom, arising from operations under the Contract Documents;
 - d. Claims for damages because of bodily injury, death of a person, or property damage arising out of the ownership, maintenance, or use of a motor vehicle, all mobile equipment, and vehicles moving under their own power and engaged in the Work;
 - e. Claims involving contractual liability applicable to the Contractor's obligations under the Contract Documents, including liability assumed by and the indemnity and defense obligations of the Contractor and the Subcontractors; and
 - f. Claims involving completed operations, independent contractors' coverage, and broad form property damage, without any exclusions for collapse, explosion, demolition, underground coverage, and excavating ("XCU").
 - g. Claims involving sudden or accidental discharge of contaminants or pollutants.
- 2. Subcontractor Insurance Requirements. The Contractor shall require its Subcontractors to take out and maintain similar public liability insurance and property damage insurance required hereunder in like amounts. A "claims made" or modified "occurrence" policy shall not satisfy the requirements of this Article without prior written approval of the District.

- 3. Additional Insured Endorsement Requirements. The Contractor shall name, on any policy of insurance required hereunder, the District and its officers, employees, agents and independent contractors as additional insureds. Subcontractors shall name the Contractor, the District, and their officers, employees, agents and independent contractors as additional insureds. The Additional Insured Endorsement included on all such insurance policies shall be an ISO CG 20 33 (04/813), or an ISO CG 20 38 (04/13) and ISO CG 20 37 (04/13) or their equivalent as determined by the District in its sole discretion, and must state that coverage is afforded the additional insured with respect to claims arising out of operations performed by or on behalf of the insured. If the additional insureds have other insurance which is applicable to the loss, such other insurance shall be on an excess or contingent basis. The insurance provided by the Contractor hereunder must be designated in the policy as primary to any insurance obtained by the District. The amount of the insurer's liability shall not be reduced by the existence of such other insurance.
- **4. Specific Insurance Requirements.** Contractor shall take out and maintain and shall require all Subcontractors, if any, whether primary or secondary, to take out and maintain:
 - a. Comprehensive General Liability Insurance with a combined single limit per occurrence of not less than \$2,000,000.00 or Commercial General Liability Insurance (including automobile insurance) which provides limits of not less than:

	_	/ 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	44 000 000 00
1	Per occurrence	(combined single limit)	\$1,000,000.00
	i ci occurrence	(combined single minic)	71,000,000.00

ii. Project Specific Aggregate (for this Project only) \$2,000,000.00

iii. Products and Completed Operations \$1,000,000.00

iv. Personal and Advertising Injury Limit \$1,000,000.00

- b. Insurance Covering Special Hazards. The following special hazards shall be covered by riders or riders to above mentioned public liability insurance or property damage insurance policy or policies of insurance, in amounts as follows:
 - i. Automotive and truck where operated in amounts \$1,000,000.00
 - ii. Material Hoist where used in amounts \$1,000,000.00
 - iii. Explosion, Collapse and Underground (XCU coverage) \$1,000,000.00
 - iv. In addition, provide Excess Liability Insurance coverage in the amount of Two Million Dollars (\$2,000,000.00).
- **5. Workers' Compensation Insurance.** At all times during the Work, the Contractor shall provide workers' compensation insurance for all of the Contractor's employees engaged in Work under this Contract on or at the Project Site and, in case any of the Contractor's

Work is subcontracted, the Contractor shall require the Subcontractor to provide workers' compensation insurance for all the Subcontractor's employees engaged in Work under the subcontract. Such workers' compensation insurance shall be in an amount of at least \$3,000,000. Any class of employee or employees not covered by a Subcontractor's insurance shall be covered by the Contractor's insurance. In case any class of employees engaged in Work under this Contract on or at the Project Site is not protected under the Workers' Compensation laws, the Contractor shall provide or cause a Subcontractor to provide adequate insurance coverage for the protection of those employees not otherwise protected. The Contractor shall file with the District certificates of insurance and in compliance with Labor Code section 3700.

6. Builder's Risk/All Risk

a. Course-of-Construction Insurance Requirements. The Contractor, during the progress of the Work and until final acceptance of the Work by District upon completion of the entire Contract, shall maintain Builder's Risk, Course of Construction, or similar first party property coverage issued on a replacement value basis consistent with the total replacement cost of the structures where work is being performed inclusive of all Work for the Project included within the Contract Documents. Coverage is to insure against all risks of accidental direct physical loss, and must include, by the basic grant of coverage or by endorsement, the perils of vandalism, malicious mischief (both without any limitation regarding vacancy or occupancy), fire, sprinkler leakage, civil authority, sonic boom, flood, collapse, wind, lightning, smoke, and riot. The coverage must include debris removal, demolition, increased costs due to enforcement of building ordinance and law in the repair and replacement of damage and undamaged portions of the property, and reasonable costs for the Architect's and engineering services and expenses required as a result of any insured loss upon the Work and Project which is the subject of the Contract Documents, including the underlying structure where Work is being performed, completed Work and Work in progress, to the full insurable value thereof. Such insurance shall include the District as additional named insureds, and any other person with an insurable interest as designated by the District.

The Contractor shall submit to the District for its approval all items deemed to be uninsurable. The risk of the damage to the Work due to the perils covered by the "Builder's Risk/All Risk" Insurance, as well as any other hazard which might result in damage to the Work, is that of the Contractor and the surety, and no claims for such loss or damage shall be recognized by the District nor will such loss or damage excuse the complete and satisfactory performance of the Contract by the Contractor.

7. Fire Insurance. Before the commencement of the Work, the Contractor shall procure, maintain, and cause to be maintained at the Contractor's expense, fire insurance on all

- Work subject to loss or damage by fire. The amount of fire insurance shall be sufficient to protect the Project against loss or damage in full until the Work is accepted by the District.
- **8. Other Insurance.** The Contractor shall provide all other insurance required to be maintained under applicable laws, ordinances, rules, and regulations.
- 9. Proof of Insurance. The Contractor shall not commence Work nor shall it allow any Subcontractor to commence Work under the LLB Agreements until all required insurance and certificates have been obtained and delivered in duplicate to the District for approval subject to the following requirements:
 - a. Certificates and insurance policies shall include the following clause:
 - i. "This policy shall not be non-renewed, canceled, or reduced in required limits of liability or amounts of insurance until notice has been mailed to the District. Date of cancellation or reduction may not be less than thirty (30) days after the date of mailing notice."
 - ii. Certificates of insurance shall state in particular those insured, the extent of insurance, location and operation to which the insurance applies, the expiration date, and cancellation and reduction notices.
 - iii. Certificates of insurance shall clearly state that the District is named as an additional insured under the policy described and that such insurance policy shall be primary to any insurance or self-insurance maintained by District.
 - iv. The Contractor and its Subcontractors shall produce a certified copy of any insurance policy required under this Article upon written request of the District.
- 10. Compliance. In the event of the failure of Contractor to furnish and maintain any insurance required hereunder, the Contractor shall be in default under the LLB Agreements. Compliance by Contractor with the requirement to carry insurance and furnish certificates or policies evidencing the same shall not relieve the Contractor from liability assumed under any provision of the Contract Documents, including, without limitation, the obligation to defend and indemnify the District and District Indemnitees.
- 11. No Waiver Created through Payments. The making of any payments under the LLB Agreements shall not be construed as creating an insurable risk interest by or for the District or be construed as relieving the Contractor or any Subcontractors of responsibility for loss from any direct physical loss, damage, or destruction occurring prior to completion of the Work.

12. Waiver of Subrogation. Contractor waives (to the extent permitted by law) any right to recover against the District for damages to the Work, any part thereof, or any and all claims arising by reason of any of the foregoing, but only to the extent that such damages and/or claims are covered by property insurance and only to the extent of such coverage (which shall exclude deductible amounts) by insurance actually carried by the District.

The provisions of this section are intended to restrict each party to recovery against insurance carriers only to the extent of such coverage and waive fully and for the benefit of each, any rights and/or claims which might give rise to a right of subrogation in any insurance carrier. The District and the Contractor shall each obtain in all policies of insurance carried by either of them, a waiver by the insurance companies thereunder of all rights of recovery by way of subrogation for any damages or claims covered by the insurance.

EXHIBIT G REPRESENTATIVES AND CONTACT INFORMATION

		strict Representatives ; (ii)			; and (iii)
respect to a Contractor telephone r	administration of Representatives in number	the LLB Agreements must contact facsim	and the perforr nile number is	mance of the Work to obtain his app	The Authorized ropriate personalNotices and
addressed a		ny of the Authorized	District Repres	entatives should, a	as applicable, be
	ion School Distric	t	Wit	th copies (same add	lress) to:
	mino Real, Ste. 10 California 92130	00,			
	_	BLE] Project Consult		•	
Authorized personal te Notices and	Contractor Repre lephone number. other communic	h respect to the Prosentative must contain ations to	ctemail addre	to obtainess is to e addressed as follo	n his appropriate
(i) Superintend be the prim	dentary contact for th	; and (ii) who i e Contractor with resp e Work. The Aut	pever may, from is the Authorized pect to the admin horized District	m time to time, so dime, so di	erve as the Job entative who will Agreements and must contact
number to	the Job Superint the Authorized D d other commun	endent is different, the district Representative cations to any of the	is <i>[insert fac</i> ne Job Superinte ns upon establish	csimile number], and ndent shall provide ning a job trailer at	d, if the facsimile his/her facsimile the Project Site.

Attn: [Insert name from clause (i) above] [Insert Street Address] [Insert City, State, Zip]

3. Architect Representative. The Architect for the Project is [insert name of architect firm]. [Insert name of AOR] is the Architect of Record for the Project and is the Architect representative who will be the primary Architect contact person with respect to the Project. The Authorized District Representatives and Authorized Contractor Representative must contact [insert Mr. or Ms. and name of AOR] to obtain [insert his or her] appropriate personal telephone number. The facsimile number for [insert Mr. or Ms. and name of AOR] is [insert facsimile number]. Notices and other communications to the Architect should be addressed as follows:

[Insert Architect Firm Name]
Attn: [Insert AOR Name]
[Insert Street Address]
[Insert City, State, Zip]

ATTACHMENT 4 - NON-COLLUSION/CONFLICT OF INTEREST FORM

(Attachment to begin on the next page)

NON-COLLUSION DECLARATION

State of California)	
County of))	
Ι,	declare as follows:	
person, partnership, company, collusive or sham; that the bidd put in a false or sham bid, and with any bidder or anyone else bidder has not in any mann conference with anyone to fix the or cost element of the bid price public body awarding the contained in the bid are true; a or her bid price or any breakdow thereto, or paid, and will not proceed that the bid are true; and the bid price or any breakdow thereto, or paid, and will not proceed the bid are true; and the bid price or any breakdow thereto, or paid, and will not proceed the bid price or any breakdow thereto, or paid, and will not proceed the bid price or any breakdow thereto.	association, organization, or conder has not directly or indirectly on as not directly or indirectly on to put in a sham bid, or that any er, directly or indirectly, soughe bid price of the bidder or any organization of that of any other bidder, ract of anyone interested in the ind, further, that the bidder has wenthereof, or the contents thereofoxy, any fee to any corporation	the party rest of, or on behalf of, any undisclosed reporation; that the bid is genuine and not induced or solicited any other bidder to olluded, conspired, connived, or agreed rone shall refrain from bidding; that the ght by agreement, communication, or other bidder, or to fix any overhead, profit, or to secure any advantage against the exproposed contract; that all statements not, directly or indirectly, submitted his of, or divulged information or data relative on, partnership, company, association, of to effectuate a collusive or sham bid.
I declare under penalty of percorrect.	jury under the laws of the Stat	te of California that the foregoing is true and
	-	Proper Name of Bidder / Company Name
	-	Signature of Authorized Agent/Officer
	-	Date



CERTIFICATION REGARDING EXECUTIVE ORDER N-6-22 BY GOVERNOR OF THE STATE OF CALIFORNIA REGARDING CONTRACTS AND SANCTIONS ON THE RUSSIAN FEDERATION

On March 4, 2022, Governor Gavin Newsom issued Executive Order N-6-22 ("EO") regarding sanctions in response to the military conflict between the Russian Federation and the Ukraine. The EO is located at:					
https://www.gov.ca.gov/wp-content/uploads/2022/03/3.4.22-Russia-Ukraine-Executive-Order.pdf.					
The EO requires certain State agencies to notify contractors and grantees of their obligations to comply with existing economic sanctions imposed by the United States government in response to Russia's actions in Ukraine, as well as any sanctions imposed by the State of California.					
It is presently unclear which contracts or projects may be subject to the EO. Nevertheless, out of an abundance of caution, the District is respectfully requesting written confirmation that you, or the entity you represent, are in compliance with the EO and sanctions.					
Further information may be found at: https://ofac.treasury.gov/sanctions-programs-and-country-nformation/ukraine-russia-related-sanctions.					
The undersigned hereby certifies to the District, under penalty of perjury under the laws of the State of California, that the following is true and correct:					
 I am a duly authorized representative of the Contractor and, in that capacity, I have executed this certification on behalf of the Contractor. 					
ii) The appropriate box is checked immediately below (check only one box), and the statement relating to the Contractor's status regarding the EO following such box is true and correct.					
☐ The Contractor <u>is</u> in compliance with the EO and any sanctions discussed therein.					
☐ The Contractor is <u>not</u> in compliance with the EO and any sanctions discussed therein					
Representative Name:					
Representative Title:					
Representative Signature:					
Date Signed:					

Project: ______
Contractor: _____



CALIFORNIA AIR RESOURCES BOARD COMPLIANCE FORM

Project:	
Bid No.	

This form must be submitted with the bid proposal at due date and time established in bidding documents or the bid may be rejected as nonresponsive.

The California Air Resources Board ("CARB") has implemented regulations for the use of certain vehicles on public works projects. For any project awarded after January 1, 2024, and under the regulations (see Cal. Code Regs., tit. 13, § 2449), the Anaheim Elementary School District ("District") must obtain valid Certificates of Reported Compliance ("CRC") from all contractors and listed subcontractors before awarding a project. The District will be required to retain CRCs for three years after the project is complete, and the CRCs are subject to review by the CARB upon five calendar days' notice.

In addition to this Form, the foregoing CRCs are also <u>required</u> at time of bid submittal and must be included with bid proposal by the stated bid due date & time. Proof of bidder and subcontractor certification must be attached to this page at the time of bid submittal.

The ARB Diesel Off-road Online Reporting System (DOORS) reporting system is where bidders may obtain Certificates of Reported Compliance: DOORS (ca.gov). There is also an additional website that provides bidders with more information on the regulation along with FAQs and user guides. The website if the Off-Road Zone: https://ww2.arb.ca.gov/our-work/programs/truckstop-resources/road-zone.

Pleas also see the following:

Fact Sheet: Contracting Requirements | California Air Resources Board

https://ww2.arb.ca.gov/resources/fact-sheets/fact-sheet-contracting-requirements

Fact Sheet: Renewable Diesel Fuel Requirements | California Air Resources Board

https://ww2.arb.ca.gov/resources/fact-sheets/fact-sheet-renewable-diesel-fuel-requirements

Fact Sheet: Added Vehicle Restrictions and Tier Phase-Out Requirements | California Air Resources Board

https://ww2.arb.ca.gov/resources/fact-sheets/fact-sheet-added-vehicle-restrictions-and-tier-phase-out-requirements

Project Name
Bidder Acknowledges CARB Compliance requirements (sign)
Name & Title of Bidder:
Date: